

UNIVERSIDADE TECNICA DE LISBOA INSTITUTO SUPERIOR TECNICO

# MULTI-BAND and WIDEBAND LOW NOISE AMPLIFIERS

MIGUEL ANDRADE MARTINS (Licenciado)

Dissertação para a obtenção do Grau de Doutor em Engenharia Electrotécnica e de Computadores

Orientador Doutor Jorge Manuel dos Santos Ribeiro Fernandes

Juri

Presidente: Reitor da Universidade Técnica de Lisboa Vogais: Doutor Dinis Gomes de Magalhães dos Santos Doutora Maria João Ramos Marques Coelho Carrilho do Rosário Doutor João Carlos Palma Goes Doutor Jorge Manuel dos Santos Ribeiro Fernandes Doutor Manuel Medeiros Silva

Tese realizada sob a orientação de Jorge Manuel dos Santos Ribeiro Fernandes Professor Auxiliar do Departamento de Engenharia Electrotécnica e de Computadores do Instituto Superior Técnico

### Sumário

A maioria das arquitecturas de receptores de rádio utiliza somente uma banda de frequência, relativamente estreita em relação à frequência central. Actualmente há uma procura grande de equipamentos capazes de comunicar com diferentes normas, utilizando diferentes bandas de frequência. Isto requer que os blocos do receptor sejam multi-banda ou tenham banda de passagem larga de modo a acomodarem as diferentes bandas em que se pretende efectuar a comunicação.

Nesta tese estudam-se diferentes topologias de amplificadores de baixo ruído, quer multi-banda, quer de banda larga.

Tipicamente, um amplificador de baixo ruído recebe sinais de alta frequência de uma antena, através de uma linha de transmissão. A distância da antena ao amplificador é geralmente da ordem de grandeza do comprimento de onda e por isso a impedância de entrada do amplificador deve estar adaptada à impedância característica da linha que vem da antena, para maximizar a potência de sinal fornecida ao amplificador. O amplificador de baixo ruído deve também amplificar o sinal recebido com adição de ruído mínima. A montagem que serve de base a muitos amplificadores de baixo ruído é a montagem cascode com degeneração indutiva.

Nesta tese são propostos dois circuitos de amplificadores de baixo ruído operando em duas (ou mais) bandas, ambos baseados na montagem cascode com degeneração indutiva. Ambos os circuitos têm uma entrada com adaptação de impedância abrangendo duas ou mais bandas simultaneamente. O primeiro circuito encaminha as diferentes bandas por caminhos distintos, utilizando os transistores cascode para activar uma ou mais saídas. O segundo circuito tem uma única saída onde ocorrem duas bandas simultaneamente. este circuito permite variar o ganho relativo nas duas bandas e, no limite, cancelar uma das bandas, através da variação de uma tensão de polarização.

Nesta tese também se mostra que é possivel realizar amplificadores de baixo ruído com banda larga utilizando duas malhas de realimentação. Em primeiro lugar, são determinadas quais as topologias de realimentação dupla adequadas à realização de amplificadores de baixo ruído. Das montagens possíveis, há uma que é analizada com mais detalhe. Uma primeira análise é feita considerando um modelo detalhado da malha de realimentação e um bloco de amplificação ideal. Posteriormente, considera-se um bloco de amplificação não ideal e um modelo simples de malha de realimentação e estabelecem-se as condições para que o bloco de amplificação se possa considerar ideal.

É projectado um protótipo do amplificador de baixo ruído com duas malhas de realimentação estudado em mais detalhe. O seu dimensionamento é descrito em detalhe e os resultados das medidas efectuadas sobre um circuito de teste são apresentados e analizados.

Palavras-Chave:

Micro-electrónica, Rádio Receptores sem Fios, Amplificador de Baixo Ruído, Amplificador de Baixo Ruído Multi-Banda, Amplificador de Baixo Ruído de Banda Larga, Realimentação com Duas Malhas, Transformador Integrado.

#### Abstract

Most wireless receiver architectures use a single frequency band, with a bandwidth much smaller than the central frequency. Nowadays, there is a great demand for equipments able to operate with different communication standards, using different frequency bands. The receiver building blocks must be either multi-band or wideband, to accommodate the different frequency bands.

In this thesis, different multi-band and wideband low noise amplifiers (LNAs) are studied.

Typically, an LNA receives high frequency signals from an antenna, through a transmission line. The distance between the antenna and the LNA is usually of the order of the signal wavelength; thus, the LNA input impedance must be matched to the transmission line in order to maximize the power transfer. The LNA must amplify the received signal with reduced noise addition. The cascode stage with inductive degeneration is the most widely used topology in LNA design.

In this thesis, two LNA circuits operating in two (or more) bands are proposed. Both circuits are based on the cascode LNA with inductive degeneration. Both circuits have input impedance matching for two or more frequency bands simultaneously. The first circuit splits the different bands through different outputs, using the cascode transistors to activate one or more outputs. The second circuit has a single output where both frequency bands are present simultaneously. This circuit is able to vary the relative gain at both bands (and may even cancel one of them) by changing the cascode bias voltage.

In this thesis, it is also shown that it is possible to design wideband LNAs using amplifiers with two feedback loops; these are referred to as double loop feedback (DLF) LNAs. The double loop feedback topologies suitable for LNA design are determined. From the possible topologies, one is studied in more detail. A performance analysis is done using a detailed model for the feedback network and an ideal amplifying block. Another analysis, using a detailed amplifier model and a simple feedback model, is performed to establish the conditions for the amplifying block to be considered ideal.

A prototype of the DLF LNA, which was studied in more detail, was produced. An important element of this LNA is an integrated transformer, which is described in detail. Measurement results of the transformer alone and of the complete LNA circuit are presented and discussed.

#### Key-Words:

Microelectronics, Wireless Receivers, Low Noise Amplifier, Multi-band LNA, Wideband LNA, Double Loop Feedback, Integrated Transformer.

#### Acknowledgements

I would like to thank the people and institutions who have supported me through the PhD work.

First and foremost, I acknowledge my supervisor and friend Prof. Jorge Fernandes, whose expertise, guidance, support, and patience have made this work possible.

An important contribution, mainly in the writing of this thesis, was given by Prof. Manuel M. Silva.

Part of this work was realized at TUDelft, where Koen van Hartingsveldt and Prof. Chris Verhoeven were a generous help.

I also want to thank personally:

- the other members of the group of *Circuitos Analógicos e Mistos* do INESC-ID: Edgar Albuquerque and Luís Oliveira;
- José Faustino for his support on the prototype development;
- Filipa, for her patient support;
- my family, and particularly my parents, Maria da Conceição and Albino;
- all my friends, who helped me to keep my mental sanity.

Different institutions were important for this work:

- INESC-ID (Instituto de Engenharia de Sistemas e Computadores Investigação e Desenvolvimento) and IST (Instituto Superior Técnico) provided the necessary conditions to realize this work;
- FCT (Fundação para a Ciência e Tecnologia) supported this work through the project POCTI/38533/ESE/2001 and scholarship BD 12155/2003;
- TUDelft (Technische Universiteit Delft) hosted the part of the work developed abroad;
- Chameleon RF project supported the LNA prototyping;
- AMS (Austria Mikro Systeme International AG) provided the transformer characterization (within the framework of the Chameleon RF project).

We know accurately only when we know little, with knowledge doubt increases. Johann Wolfgang von Goethe

Wichtig ist, dass man nicht aufhört fragen. <br/>1\$ Albert Einstein

<sup>&</sup>lt;sup>1</sup>It is important that men never stops questioning.

### Contents

Sι	ımár	io				v
A	bstra	$\mathbf{ct}$				vii
A	cknov	wledge	ements			ix
C	onter	$\mathbf{nts}$				$\mathbf{x}\mathbf{v}$
$\mathbf{Li}$	ist of	Figur	es			xix
Li	ist of	Table	s		X	xvii
Li	ist of	$\mathbf{Symb}$	ols and Acronyms		х	xix
1	Intr	oducti	ion			1
	1.1	Backg	round and Motivation			3
	1.2	Organ	ization of the Thesis			5
	1.3	Origin	al Contributions	•		6
<b>2</b>	Тор	ics on	Radio Receivers			7
	2.1	Introd	luction	•		9
	2.2	Wirele	ess Receiver Architectures			11
		2.2.1	Common Wireless Receiver Architectures			12
		2.2.2	Receiver Building Blocks			13
	2.3	Basic	Concepts of RF Design			15
		2.3.1	Impedance Matching and Reflection Coefficient			15
		2.3.2	Scattering Parameters			17
		2.3.3	Power Gain			19
		2.3.4	Signal Distortion and Linearity			20
		2.3.5	Bandwidth			21
	2.4	Noise				22

		2.4.1	Noise and the Wiener-Khintchine Theorem	22
		2.4.2	Noise Figure	23
		2.4.3	Cascaded Systems	25
		2.4.4	Noise Sources	26
		2.4.5	Noise in Transistors	28
	2.5	Integr	ated Inductors and Transformers	31
		2.5.1	Integrated Inductors	31
		2.5.2	Integrated Transformers	35
	2.6	Conclu	usions	40
3	Ove	erview	of Low Noise Amplifiers	41
	3.1	Introd	luction	43
	3.2	LNA ]	Basic Concepts	44
	3.3	LNA '	Topologies	45
		3.3.1	LNA using Input Resistor	45
		3.3.2	Common-Gate LNA	46
		3.3.3	Common-Source LNA with Inductive Degeneration $\ldots \ldots \ldots$	47
		3.3.4	LNA with Current Reuse	48
		3.3.5	LNA with Resistive Feedback	49
		3.3.6	Discussion	50
	3.4	Casco	de LNA with Inductive Degeneration	51
		3.4.1	Effect of $C_{gd}$	51
		3.4.2	Input Impedance and Noise Figure	53
	3.5	Doubl	e Loop Feedback LNAs	56
		3.5.1	Four Feedback Topologies	56
		3.5.2	Feedback Low Noise Amplifiers	58
		3.5.3	Double Loop Feedback (DLF) $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	59
		3.5.4	DLF Amplifiers suitable for LNA Design	63
		3.5.5	DLF LNA Topologies	63
	3.6	Concl	usions	71
4	Mu	lti-ban	d LNA using Cascode Transistors for Band Selection	73
	4.1	Introd	luction	75
	4.2	A Nev	v Multi-Band Low Noise Amplifier	76
		4.2.1	Topics on Multi-band Low Noise Amplifiers	76
		4.2.2	A New Multi-Band LNA	77
		4.2.3	Noise Analysis	79
	4.3	Simula	ation Results	82

	4.4	Conclusi	ons				86
<b>5</b>	Cor	ncurrent	Dual-Band	LNA	using	Magnetically	Coupled
	Ind	uctors					87
	5.1	Introduc	tion				89
	5.2	Concurre	ent Multi-Band	LNAs .			90
	5.3	Concurre	ent Dual-Band (	CMOS L	NA using	Magnetically Co	upled In-
		ductors .					92
		5.3.1 V	Vorking Principl	e			92
		5.3.2 F	requency Response	nse			93
	5.4	Effect of	the Inductors F	arasitic	Resistanc	e	96
	5.5	Simulati	on Results				
		5.5.1 C	Circuit Dimension	ning			
		5.5.2 V	Voltage Gain Var	iation .			101
	5.6	Effect of	Reversing the C	Drientatio	on of the	Magnetic Couplin	ıg
	5.7	Conclusi	ons				107
6	Doι	ıble Looj	p Feedback LN	JA			109
	6.1	Introduc	tion				111
	6.2	DLF LN	A with Ideal Blo	ocks			112
	6.3	Feedback	k Network Analy	vsis			114
		6.3.1 E	Effect of Inducta	nces $L_{wi}$	and Resis	stances $R_{wi}$	115
		6.3.2 E	Effect of Inter-W	inding C	apacitanc	$\cos C_{wi} \ldots \ldots$	118
	6.4	Amplifyi	ing Block Analys	sis			122
		6.4.1 In	nput Impedance	and Vol	tage Gain	n Evaluation usin	g a Non-
		Ie	deal Amplifying	Block .			122
		6.4.2 C	Conditions for an	Ideal A	mplifying	Block	126
	6.5	Simulati	on Results				129
		6.5.1 F	eedback Networ	k Design	••••		129
		6.5.2 A	Amplifying Block	Design			130
	6.6	Noise an	alysis				136
	6.7	Conclusi	ons				141
7	LN	A Protot	ype Design ar	ıd Expe	rimenta	l Results	143
	7.1	Introduc	tion				145
	7.2	DLF LN	A Design				146
		7.2.1 Т	The Feedback Ne	twork D	esign		147
		7.2.2 A	Amplifying Block	Design			148

		7.2.3 Buffer Design	. 150
		7.2.4 Simulation Results	. 151
	7.3	DLF LNA Layout	. 155
		7.3.1 Transformer layout	. 155
		7.3.2 General Layout Considerations	. 158
		7.3.3 Post-Layout Simulations	. 160
	7.4	Test Board Design and Wire Bonding	. 163
	7.5	Experimental Results	. 166
		7.5.1 Transformer $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	. 166
		7.5.2 LNA	. 168
		7.5.3 Discussion and improvements for a second prototype $\ldots$ .	. 172
	7.6	Conclusions	. 177
8	Con	clusions and Future Work	179
	8.1	Conclusions	. 181
	8.2	Future Work	. 182
$\mathbf{A}$	2-P	ort Matrices	185
Б	וח		•
В	Pole	es and Zeros of the DLF LINA considering Transformer Parasit	107
	Cap	acitances and Resistances	
	D 1		100
	B.1	Input Impedance Analysis	. 188
	B.1 B.2	Input Impedance Analysis	. 187 . 188 . 190
С	B.1 B.2 <b>Sou</b>	Input Impedance Analysis	. 187 . 188 . 190 <b>193</b>
C D	B.1 B.2 Sou	Input Impedance Analysis	<ul> <li>. 187</li> <li>. 188</li> <li>. 190</li> <li><b>193</b></li> <li><b>195</b></li> </ul>
C D	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DL1</li> <li>D.1</li> </ul>	Input Impedance Analysis	<ul> <li>. 187</li> <li>. 188</li> <li>. 190</li> <li>. 193</li> <li>. 195</li> </ul>
C D	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> </ul>	Input Impedance Analysis	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>195</li> <li>195</li> <li>198</li> </ul>
C D E	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DL1</li> <li>D.1</li> <li>D.2</li> <li>Tec.</li> </ul>	Input Impedance Analysis	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>195</li> <li>195</li> <li>198</li> <li>201</li> </ul>
C D E F	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> <li>Tec:</li> <li>Imp</li> </ul>	Input Impedance Analysis       .         Voltage Gain Analysis       .         rce Transformations       .         F LNA Input Noise Sources       .         Demonstration of (6.12)       .         Demonstration of (6.44)       .         hnology File for ASITIC Program         Dedance Evaluation of Integrated Circular Spiral Inductors	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>193</li> <li>195</li> <li>195</li> <li>195</li> <li>198</li> <li>201</li> <li>205</li> </ul>
C D E F	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> <li>Tec:</li> <li>Imp</li> <li>F.1</li> </ul>	Input Impedance Analysis       .         Voltage Gain Analysis       .         rce Transformations       .         F LNA Input Noise Sources       .         Demonstration of (6.12)       .         Demonstration of (6.44)       .         hnology File for ASITIC Program         edance Evaluation of Integrated Circular Spiral Inductors         Introduction       .	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>193</li> <li>195</li> <li>195</li> <li>198</li> <li>201</li> <li>205</li> <li>205</li> </ul>
C D F	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> <li>Tec:</li> <li>Imp</li> <li>F.1</li> <li>F.2</li> </ul>	Input Impedance Analysis	<ul> <li>187</li> <li>. 188</li> <li>. 190</li> <li>193</li> <li>195</li> <li>. 195</li> <li>. 195</li> <li>. 198</li> <li>201</li> <li>205</li> <li>. 205</li> <li>. 207</li> </ul>
C D F	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> <li>Tec:</li> <li>Imp</li> <li>F.1</li> <li>F.2</li> <li>F.3</li> </ul>	Input Impedance Analysis       .         Voltage Gain Analysis       .         rce Transformations       .         F LNA Input Noise Sources       .         Demonstration of (6.12)       .         Demonstration of (6.44)       .         hnology File for ASITIC Program         Dedance Evaluation of Integrated Circular Spiral Inductors         Introduction       .         Description of the problem       .         Results       .	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>195</li> <li>195</li> <li>195</li> <li>201</li> <li>205</li> <li>207</li> <li>211</li> </ul>
C D F R	<ul> <li>B.1</li> <li>B.2</li> <li>Sou</li> <li>DLI</li> <li>D.1</li> <li>D.2</li> <li>Tec:</li> <li>Imp</li> <li>F.1</li> <li>F.2</li> <li>F.3</li> <li>efere</li> </ul>	Input Impedance Analysis	<ul> <li>187</li> <li>188</li> <li>190</li> <li>193</li> <li>195</li> <li>195</li> <li>195</li> <li>201</li> <li>205</li> <li>207</li> <li>211</li> <li>215</li> </ul>

# List of Figures

1.1	Input of a general wireless receiver	3
2.1	Basic configuration of a communication system.	9
2.2	Possible block diagram of a wireless receiver	11
2.3	Narrowband wireless receiver.	12
2.4	2-port block	15
2.5	Transmission line with characteristic impedance $Z_0$ terminated with	
	impedance $Z_L$ and excited by voltage generator $V_s$	16
2.6	2-port block with incident and reflected waves	17
2.7	System modeling $f(x(t))$ having input $x(t)$ and output $y(t)$	20
2.8	Definition of $IIP_3$	21
2.9	2-port with transfer function $H(f)$ excited by noise source with power	
	spectral density $N_x(f)$	23
2.10	Noisy 2-port with available power gain $G_A(f)$ excited by signal and	
	noise source	24
2.11	2-port circuit block with equivalent noise sources	25
2.12	Cascade of 2-ports	25
2.13	MOS transistor noise sources for RF design	28
2.14	Equivalent input noise sources in a MOS transistor	29
2.15	BJT noise sources for RF design.	29
2.16	BJT with equivalent input referred noise sources. $\ldots$ $\ldots$ $\ldots$ $\ldots$	30
2.17	2-D view of an integrated inductor	32
2.18	3-D view of an integrated inductor.	32
2.19	$\pi$ circuit model of an integrated spiral inductor	33
2.20	2-D view of a patterned ground shield	35
2.21	2-D view of a tapped transformer	36
2.22	2-D view of an interleaved transformer	37
2.23	3-D view of a stacked transformer	37

2.24	2-D view of a stacked transformer in which the windings are shifted	
	to control the magnetic coupling coefficient.	38
2.25	Symbol of an ideal transformer.	38
2.26	Transformer 2-port symbol considering winding inductances $L_{mm}$	39
3.1	MOS incremental $\pi$ model	45
3.2	LNA with resistive input power matching.	46
3.3	Incremental model of the common-gate LNA (biasing not represented).	46
3.4	Common-Source LNA using inductive degeneration (biasing of $M_1$	
	not represented). $\ldots$	47
3.5	Incremental model of the common-source LNA using inductive dege-	
	neration	48
3.6	LNA using current reuse (biasing of $M_1$ and $M_2$ not represented)	49
3.7	LNA with resistive feedback (biasing of $M_1$ not represented)	49
3.8	Incremental model of the LNA with resistive feedback.	50
3.9	Transformer feedback LNA (biasing of $M_1$ not represented)	51
3.10	LNA using tuned inductor (biasing of $M_1$ not represented)	52
3.11	Cascode LNA (biasing of $M_1$ not represented)	52
3.12	Cascode configuration incremental model.	53
3.13	Incremental model of the degenerated cascode LNA with noise sources.	54
3.14	Incremental model of the degenerated cascode LNA with an input	
	equivalent voltage noise source.	55
3.15	Block diagram of a feedback system.	56
3.16	Series-series feedback	57
3.17	Series-shunt feedback	57
3.18		
9 10	Shunt-series feedback	57
3.19	Shunt-series feedback.    Shunt-shunt feedback.	57 57
3.19 3.20	Shunt-series feedback.       Shunt-shunt feedback.         Series-series feedback with two-terminal element.       Series-series	57 57 58
<ul><li>3.19</li><li>3.20</li><li>3.21</li></ul>	Shunt-series feedback.       Shunt-shunt feedback.         Shunt-shunt feedback.       Shunt-shunt feedback with two-terminal element.         Shunt-shunt feedback with two-terminal element.       Shunt-shunt feedback with two-terminal element.	57 57 58 58
<ol> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> </ol>	Shunt-series feedback.       Shunt-shunt feedback.         Shunt-shunt feedback.       Shunt-shunt feedback with two-terminal element.         Shunt-shunt feedback with two-terminal element.       Shunt-shunt feedback with two-terminal element.         Series-shunt feedback with a transformer.       Shunt-shunt feedback with a transformer.	57 57 58 58 59
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> </ul>	Shunt-series feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Shunt-shunt feedback with two-terminal element.Series-shunt feedback with a transformer.Shunt-series feedback with a transformer.	57 57 58 58 59 59
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> <li>3.24</li> </ul>	Shunt-series feedback.       .         Shunt-shunt feedback.       .         Series-series feedback with two-terminal element.       .         Shunt-shunt feedback with two-terminal element.       .         Shunt-shunt feedback with two-terminal element.       .         Shunt-shunt feedback with a transformer.       .         Shunt-shunt feedback with a transformer.       .         Amplifying block having finite input impedance.       .	57 58 58 59 59 59
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> <li>3.24</li> <li>3.25</li> </ul>	Shunt-series feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Shunt-shunt feedback with two-terminal element.Shunt-shunt feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback having finite input impedance.Shunt-series feedback having finite input impedance.States feedback having finite input impedance.States feedback having finite input impedance.States feedback having finite input impedance. </td <td>57 58 58 59 59 59 60</td>	57 58 58 59 59 59 60
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> <li>3.24</li> <li>3.25</li> <li>3.26</li> </ul>	Shunt-series feedback.Shunt-shunt feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Shunt-shunt feedback with two-terminal element.Series-shunt feedback with a transformer.Series-shunt feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback and transformer.Series-shunt feedback with a transformer.Shunt-series feedback and transformer.Series-shunt feedback and transformer.Shunt-series feedback amplifier sampling different variables and com-	57 57 58 59 59 59 60
3.19 3.20 3.21 3.22 3.23 3.24 3.25 3.26	Shunt-series feedback.Shunt-shunt feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Series-series feedback with two-terminal element.Series-shunt feedback with two-terminal element.Shunt-shunt feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback having finite input impedance.Series-shunt feedback having finite input impedance.Two feedback loops, both comparing (a) current; or (b) voltage.Series-shunt feedback having finite sampling different variables and comparing different variables.	<ul> <li>57</li> <li>57</li> <li>58</li> <li>59</li> <li>59</li> <li>60</li> <li>60</li> </ul>
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> <li>3.24</li> <li>3.25</li> <li>3.26</li> <li>3.27</li> </ul>	Shunt-series feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Shunt-shunt feedback with two-terminal element.Shunt-shunt feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback with a transformer.Shunt-series feedback with a transformer.Two feedback loops, both comparing (a) current; or (b) voltage.Double loop feedback amplifier sampling different variables and comparing different variables.Double loop feedback amplifier sampling the same variable and com-	<ul> <li>57</li> <li>57</li> <li>58</li> <li>58</li> <li>59</li> <li>59</li> <li>60</li> <li>60</li> </ul>
<ul> <li>3.19</li> <li>3.20</li> <li>3.21</li> <li>3.22</li> <li>3.23</li> <li>3.24</li> <li>3.25</li> <li>3.26</li> <li>3.27</li> </ul>	Shunt-series feedback.Shunt-shunt feedback.Series-series feedback with two-terminal element.Series-series feedback with two-terminal element.Shunt-shunt feedback with two-terminal element.Series-shunt feedback with a transformer.Shunt-series feedback with a transformer.Series-shunt feedback with a transformer.Shunt-series feedback applifier input impedance.Series-shunt feedback applifier input impedance.Shunt-series feedback having finite input impedance.Series-shunt feedback applifier sampling different variables and comparing different variables.Double loop feedback amplifier sampling the same variable and comparing different variables.Series-shunt feedback applifier sampling the same variable and comparing different variables.	<ul> <li>57</li> <li>57</li> <li>58</li> <li>59</li> <li>59</li> <li>60</li> <li>60</li> <li>62</li> </ul>

3.29	Amplifier having series-shunt feedback inside shunt-shunt feedback.	•	63
3.30	Amplifier having shunt-series feedback inside series-series feedback.	•	63
3.31	Amplifier having shunt-shunt feedback inside series-shunt feedback.	•	63
3.32	DLF LNA type 1	•	64
3.33	DLF LNA type 1 and the $R_f$ noise source		65
3.34	Noise transformations to determine the equivalent LNA input noise		
	source		66
3.35	DLF LNA type 1 with equivalent input noise source.		66
3.36	DLF LNA type 2		67
3.37	DLF LNA type 3		68
3.38	DLF LNA type 4	•	69
4.1	Two multi-band receiver architectures.	•	76
4.2	Cascode LNA with inductive degeneration (biasing of $M_1$ not repre-		
	sented).	•	77
4.3	Dual-band LNA based on cascode switching ( $M_1$ bias not included).	•	78
4.4	Incremental model of the input part of the multi-band LNA with		
	noise sources.	•	79
4.5	Incremental model of the multi-band LNA showing noise source trans-		
	formations.	•	80
4.6	Voltage gain at "active" outputs 1 and 2 when one cascode transistor		
	is in saturation and the other is cutoff. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	•	83
4.7	Comparison of voltage gain at active and inactive outputs	•	83
4.8	Voltage gain at outputs 1 and 2, with both outputs active and with		
	only one output active	•	84
4.9	Voltage gain at output 2 when $V_{BIAS2}$ varies and $V_{BIAS1} = V_{DD}$	•	84
4.10	$S_{11}$ plots in three different situations, showing the low influence of		
	the different operating modes	•	85
4.11	Simulated noise figure of the complete dual-band LNA circuit	•	85
5.1	Two multi-band receiver architectures.		90
5.2	a) Variable capacitor. b) Variable inductor	•	91
5.3	Dual-band LNAs using multi-resonant circuits. (a) LNA based on		
	cascode topology with inductive degeneration. (b) LNA based on		
	the common-gate stage ( $C_A$ and $C_B$ realize a series-shunt feedback).		
	Biasing of $M_1$ not presented	•	91
5.4	Concurrent dual-band LNA based on inductor magnetic coupling. $% \mathcal{A}$ .		92
5.5	Double tuned circuit with coupled inductors	•	93

5.6	Possible plot of the absolute value of the load impedance $Z_{load}$
5.7	Possible plot of the load impedance $Z_{load}$ , considering the inductor
	series resistances
5.8	Voltage gain
5.9	$S_{11}$ simulation of the concurrent dual-band LNA
5.10	Noise figure: theoretical values compared with the simulation results. 100
5.11	Zero frequency versus $\alpha$
5.12	Voltage gain with equal cascode transistors: $\alpha$ varies from 0 to 1
	$(V_{BIAS1} = V_{DD} \text{ and } 0 \le V_{BIAS2} \le V_{DD}).$
5.13	Voltage gain with unequal cascode transistors : $\alpha$ varies from 0 to 6
	$(V_{BIAS1} = V_{DD} \text{ and } 0 \le V_{BIAS2} \le V_{DD}).$
5.14	Concurrent dual-band LNA topology with opposite magnetic coupling. 104 $$
5.15	Zero position versus $\alpha.$ Curve 1: $k=0.35.$ Curve 2: $k=0.50.$ 105
5.16	Voltage gain frequency response versus $\alpha$ . $\alpha$ in the range 0 to 1
	$(V_{BIAS2} \in [1, 1.9] \text{ V}) \text{ and } k = 0.35 106$
5.17	Voltage gain frequency response versus $\alpha$ . $\alpha$ in the range 0 to 1
	$(V_{BIAS2} \in [1, 1.2] \text{ V}) \text{ and } k = 0.50 106$
6.1	DLF LNA type 1, sampling the output voltage
6.2	Transformer model
6.3	Transformer model including winding resistances $R_{wi}$ and inter-
	winding capacitances $C_{wi}$
6.4	DLF LNA with transformer model including inductances $L_{wi}$ and
	resistances $R_{wi}$
6.5	Asymptotic Bode plot of the input impedance $ Z_{in}(f) $ of the DLF
	LNA considering the transformer inductances $L_{wi}$ and resistances $r_{wi}$ . 116
6.6	Asymptotic Bode plot of the DLF LNA voltage gain $ A_v $ considering
	the transformer inductances $L_{wi}$ and resistances $r_{wi}$
6.7	DLF LNA noise sources
6.8	DLF LNA with transformer model including inductances $L_{wi}$ , resis-
	tances $R_{wi}$ and, inter-winding capacitances $C_{wi}$
6.9	Possible plot of the DLF LNA input impedance considering the com-
	plete transformer model including inductances, resistances and ca-
	pacitances
6.10	Possible plot of the DLF LNA voltage gain considering the complete
	transformer model including inductances, resistances and capacitances. 120 $$
6.11	General amplifier model
6.12	Simplified amplifier model for circuit analysis

6.13	Transformer used in the DLF LNA of figure 6.1
6.14	Feedback block formed by $R_f$ used in the DLF LNA of figure 6.1. $$ 125
6.15	Conditions 1 to 3 along frequency and representation of the minimum
	$G_m$ determination
6.16	Cascode stage
6.17	Simplified incremental model of the cascode stage
6.18	Amplifying block transconductance and conditions 1 to 3. Transcon-
	ductance determined for a current of 5 mA and condition curves de-
	termined with values of table 6.3
6.19	Real part of $Z_{in}$
6.20	Imaginary part of $Z_{in}$
6.21	$S_{11}$
6.22	voltage gain
6.23	DLF LNA noise sources
6.24	BJT noise sources for RF design
6.25	Equivalent noise sources at the transistor input
6.26	Equivalent noise sources at the LNA input
6.27	Noise figure
7.1	DLF LNA to be designed
7.2	Block diagram of the prototype (A is the amplifying block and $\beta$ is
	the feedback network)
7.3	DLF LNA (biasing of $Q_1$ not shown)
7.4	Amplifying block biasing
7.5	Buffer with biasing circuit
7.6	Complete DLF LNA circuit
7.7	$S_{11}$ with DC current values: 2.5, 5.0, 7.5, and 10.0 mA
7.8	Voltage gain with DC current values: 2.5, 5.0, 7.5, and 10.0 mA 153
7.9	Noise figure with DC current values: 2.5, 5.0, 7.5, and 10.0 mA. $\ .$ 153
7.10	$S_{11}$ considering resistances and inter-winding capacitances, with DC
	current values: 2.5, 5.0, 7.5, and 10.0 mA. $\dots \dots \dots$
7.11	Voltage gain considering resistances and inter-winding capacitances,
	with DC current values: 2.5, 5.0, 7.5, and 10.0 mA. $\dots \dots \dots$
7.12	Noise figure considering resistances and inter-winding capacitances,
	with DC current values: 2.5, 5.0, 7.5, and 10.0 mA
7.13	Patterned Ground Shield
7.14	Transformer layout
7.15	ASITIC results for the transformer

7.16	Complete DLF LNA and buffer layout
7.17	$S_{11}$ of the complete circuit (LNA and buffer): pre and post-layout
	simulations
7.18	$S_{12}$ of the complete circuit (LNA and buffer): pre and post-layout
	simulations
7.19	$S_{21}$ of the complete circuit (LNA and buffer): pre and post-layout
	simulations
7.20	$S_{22}$ of the complete circuit (DLF LNA and buffer): pre and post-
	layout simulations
7.21	Noise figure of the complete circuit (DLF LNA and buffer): pre and
	post-layout simulations
7.22	Board 1 layout - natural size
7.23	Test circuit with pads. $\dots \dots \dots$
7.24	Test setup
7.25	Test board with die. $\ldots \ldots \ldots$
7.26	$S_{11}$ of the transformer - experimental and simulated results 166
7.27	$S_{12}$ of the transformer - experimental and simulated results 167
7.28	$S_{21}$ of the transformer - experimental and simulated results 167
7.29	$S_{22}$ of the transformer - experimental and simulated results 168
7.30	Measured $S_{11}$ and simulation considering bonding inductances 170
7.31	Measured $S_{12}$ and simulation considering bonding inductances 170
7.32	Measured $S_{21}$ and simulation considering bonding inductances 171
7.33	Measured $S_{22}$ and simulation considering bonding inductances 171
7.34	Measured noise figure and simulation considering bonding inductances.172
7.35	Layout of the second circuit prototype
7.36	$S_{11}$ considering $L_{GND} = L_{VDD} = 1$ nH (solid line) and $L_{GND} =$
	$L_{VDD} = 0.2 \text{ nH} \text{ (dashed line)}. \dots \dots$
7.37	$S_{12}$ considering $L_{GND} = L_{VDD} = 1$ nH (solid line) and $L_{GND} =$
	$L_{VDD} = 0.2 \text{ nH} \text{ (dashed line)}. \dots \dots$
7.38	$S_{21}$ considering $L_{GND} = L_{VDD} = 1$ nH (solid line) and $L_{GND} =$
	$L_{VDD} = 0.2 \text{ nH (dashed line)}. \dots \dots$
7.39	$S_{22}$ considering $L_{GND} = L_{VDD} = 1$ nH (solid line) and $L_{GND} =$
	$L_{VDD} = 0.2 \text{ nH} \text{ (dashed line)}. \dots \dots$
7.40	Noise figure considering $L_{GND} = L_{VDD} = 1$ nH (solid line) and
	$L_{GND} = L_{VDD} = 0.2 \text{ nH} \text{ (dashed line)}. \dots \dots$
8.1	Proposed dual-band LNA - Mixer

A.1	2-port block
B.1	DLF LNA with transformer model including winding inductances $L_{wi}$ ,
	series resistances $K_{wi}$ and inter-winding capacitances $C_{wi}$
C.1	Voltage source shifting (Blakesley transformation)
C.2	Splitting of a current source
C.3	Norton-Thevenin sources equivalence
C.4	2-port transformation of noise sources
D.1	DLF LNA noise sources
D.2	Transformation of the noise voltage source $v_{n,Rw2}$
D.3	Transformation of the noise voltage source $v_{n,Rf}$
D.4	Transformation of the noise voltage source $v_{n,Rw1}$
D.5	DLF LNA noise sources
D.6	Transformation of the noise voltage source $v_{n,Rw1}$
E.1	Figure containing the different layers of a technology process 203
F.1	Spiral square inductor
F.2	Vertical plane projection of the concentric inductor
F.3	Concentric ring inductor
F.4	Inductance A vs frequency
F.5	Quality factor of inductor A
F.6	Inductance B vs frequency
F.7	Quality factor of inductor B
F.8	Inductance C vs frequency
F.9	Quality factor of inductor C

### List of Tables

3.1	Four feedback topologies
3.2	Input impedance for the DLF amplifier combinations of figure 3.26 $$ 61
3.3	Input impedance for the DLF amplifier combinations of figure 3.27 $$ 62
4.1	Element values used in simulations and theoretical curves 82
5.1	Element values of the concurrent dual-band CMOS LNA using cou-
	pled inductors
6.1	Transformer parameters
6.2	Frequencies of the poles and zeros determined in (6.14) and (6.15) 130
6.3	Bipolar cascode incremental parameters
7.1	LNA guidelines
7.2	Transformer parameters
7.3	Main transformer parameters
7.4	List of laboratory equipment
7.5	Estimate of the bond wire inductances
7.6	DC voltages and currents
A.1	2-port matrix descriptions
F.1	Geometrical parameters (N - number of turns, $D$ - diameter, $w$ -
	width, $s$ - section), inductance at maximum quality factor, $L$ , and
	maximum quality factor, $Q$ , of three inductors in medium-resistivity
	factor

### List of Symbols and Acronyms

$\delta$	Skin effect parameter
$\Delta f$	Bandwidth
$\varepsilon_{\rm ox}$	Oxide permittivity
$\gamma$	Bias dependent factor
Γ	Reflection coefficient
$\lambda$	Wavelength
$\mu$	Air magnetic field
$\Psi$	Magnetic flux
$\sigma$	Metal conductivity
ADC	Analog-to-Digital Converter
AMS	Austria Mikro Systeme
F	Noise Factor
$\mathrm{DLF}$	Double Loop Feedback
$\operatorname{IF}$	Intermediate Frequency
$i_n$	noise current
G	Power gain
$G_A$	Available gain
$g_m$	Transistor transconductance
$G_m$	Amplifying block transconductance
$G_T$	Transducer power gain
H	Magnetic field
$IIP_3$	Third-order intercept point
j	Imaginary unit
k	Magnetic coupling coefficient
$k_B$	Boltzmann constant
$K_1, K_2, \dots$	Taylor coefficients
LNA	Low Noise Amplifier
LO	Local Oscillator
n	Transformer voltage ratio
	Noise power
NF	Noise Figure
$P_{-1dB}$	1 dB compression point
PSD	Power Spectral Density
q	Electron charge
	Radio Frequency
$n_S$	Source Resistance
S C;	Signal power
SiCo	Silicon Cormonium
SIGe	Silicon Ovido
$SIO_2$	Sectoring parameter
$\mathcal{D}_{xy}$ SNP	Signal to Noise Batio
T	Absolute temperature
1 ?	Noise voltage
$Z_{n}$	Characteristic impodance
$\omega_0$	Unaracteristic impedance

## Chapter 1

### Introduction

1.1	Background and Motivation	3
1.2	Organization of the Thesis	5
1.3	Original Contributions	6

#### 1.1 Background and Motivation

A few years ago, wireless receivers worked typically at a single frequency band using a single communication standard; however, the interest in wireless receivers working with different communication standards at different frequencies is increasing [1].

In figure 1.1 the input of a general wireless receiver is represented. It comprises several blocks that can be found in most wireless receiver architectures. These blocks are an antenna, a low noise amplifier (LNA), a mixer, and a local oscillator (LO).



Figure 1.1: Input of a general wireless receiver.

Typically, the antenna receives a high frequency carrier modulated by a signal that is amplified by the LNA and is then multiplied at the mixer by a periodic signal generated at the LO [2–5]. The simplest form of obtaining multi-standard receivers consist of duplicating the receiver and tuning the replicas to different frequencies; however, this is not the most economical solution [1]. An alternative consists on using multi-band blocks [6].

This thesis is dedicated to the study of LNAs suitable for multi-standard receivers.

The LNA is the first block of a receiver, and is responsible for the amplification of the radio signal received by the antenna. The received radio frequency (RF) signal has typically a high frequency, which leads to propagation effects (if the signal travels distances of the same order of the signal wavelength); these should be minimized by proper impedance matching. This is the case of the connection between the antenna and the LNA input, which is typically performed by a standard 50  $\Omega$  transmission line that should be terminated with 50  $\Omega$  at both ends. Adapting the impedances at both ends of the transmission line prevents the signal power to be reflected towards

the antenna, when it is received by the LNA. This is specially important when the signal-to-noise ratio (SNR) is low. The LNA should amplify the signal power, with minimum addition of noise, to minimize the SNR degradation. Three major LNA parameters are [2–5, 7–9]:

- gain;
- input impedance;
- noise factor.

The LNA frequency response is also important [4, 10–21]. LNAs can be:

- narrowband;
- multi-band;
- wideband.

Narrowband LNAs work in a narrow band around a high frequency (the bandwidth is several orders of magnitude lower than the center frequency). A multi-band LNA is able to operate at different frequency bands, and a wideband LNA has a large bandwidth (the bandwidth can be of the same order of magnitude of the center frequency). It is desirable that multi-band and wideband LNAs are not obtained simply by replicating several narrowband LNAs, to save power and die-area [6]. In this thesis, techniques to design multi-band and wideband LNAs are studied.

The study of multiband LNAs starts by an overview of their most common design techniques. Some drawbacks are pointed out, like parasitic resistances in the signal path due to switches, or a large die area. To overcome these drawbacks, a new topology based on using inductors with magnetic coupling is proposed. The magnetic coupling leads to the dual-band capability of this LNA, and using current steering it is possible to control the relative gain in the two bands.

Wideband LNAs are important, to explore forms of communication, other than the typical carrier based ones [10, 22]. This thesis explores the applicability of double loop feedback (DLF) to design wideband LNAs [23, 24]. A DLF amplifier is studied in detail and a prototype is designed and measured to confirm the feasibility of this type of wideband LNAs.

#### 1.2 Organization of the Thesis

This thesis is divided into eight chapters and six appendixes. The first three chapters, including this one, show the interest of multi-band and wideband LNAs, describe the basic concepts of radio receivers, and present an overview of the design of low noise amplifiers. Chapters four to six describe the new work developed about multi-band and wideband LNAs. Chapter seven describes the prototyping of a wideband LNA and finally chapter eight draw some conclusions and perspectives future work. Chapters are described ahead in more detail.

Chapter two describes the most important topics on RF design and some major LNA parameters. The scattering parameters are introduced, due to their importance in RF design, mainly to evaluate input impedance matching and gain. The most important noise sources and the circuits noise performance are discussed. Finally, a section is dedicated to the design of inductors and transformers, due to their importance on LNA implementation.

Chapter three gives an overview of LNA implementation. It describes LNA basic concepts, and presents several LNA topologies. Due to its importance on LNA design, the cascode LNA with inductive degeneration is described in more detail. A little known type of LNAs is obtained by using double loop feedback (DLF). In this chapter, DLF topologies suitable for LNA implementation are investigated considering ideal blocks.

Chapters four and five are devoted to multi-band LNAs. In chapter four a multiband LNA having multiple outputs is studied. This topology receives a multi-band signal that is amplified and split through different narrowband paths. It has the advantage that there are no parasitic resistances in the signal path. In chapter five an LNA with magnetically coupled inductors is proposed. Two different bands can be tuned in the same output. It is shown that the gain at each band can be controlled easily through the variation of a single bias voltage.

In chapter six, one of the DLF topologies suitable for LNA realization that were identified in chapter three is analyzed in more detail. The analysis is divided into two steps. First the LNA performance limits are evaluated assuming an ideal amplifying block. Then, the amplifying block is analyzed and dimensioned in order to be approximately ideal. In chapter seven, a prototype of the DLF LNA studied in chapter six is designed and experimental results are presented.

Finally in chapter eight, some conclusions are given and suggestions for future research are proposed.

#### **1.3** Original Contributions

The original contributions presented in this thesis are:

- 1. study of a multi-band LNA, based on the cascode LNA with inductive degeneration, that has multiple outputs, each one controlled by a cascode transistor;
- proposal of a new dual-band LNA based on current balancing and inductor magnetic coupling, analysis of this new LNA in terms of noise, gain and input impedance [25, 26];
- 3. study of DLF topologies suitable for LNA realization;
- 4. study of one DLF LNA, concerning the input impedance, voltage gain and noise performance, considering an ideal amplifying block [27, 28];
- 5. study of the conditions for the amplifying block in the DLF LNA to be considered ideal [29].

The author has collaborated in a study of the impedance evaluation of integrated circular spiral inductors. Since this matter is somewhat outside the scope of this thesis, it is presented in appendix F [30].
# Chapter 2

# **Topics on Radio Receivers**

2.1	Introduction	9
2.2	Wireless Receiver Architectures	11
2.3	Basic Concepts of RF Design	15
2.4	Noise	22
2.5	Integrated Inductors and Transformers	31
2.6	Conclusions	40

## 2.1 Introduction

In figure 2.1 a basic configuration of a communication system is represented [5]. The transceiver (*trans*mitter and receiver) is responsible for processing a signal transmitted through a channel and to process it into an information recognizable by another part of the system where the transceiver is inserted. The received/sent signals are characterized by a frequency spectrum, that theoretically can occupy frequencies from zero to infinity; however, lower frequencies carry few information and require long antennas, and higher frequencies are highly attenuated [31].



Figure 2.1: Basic configuration of a communication system.

The transceiver performance is determined by the capability of transmitting information. Shannon's equation (2.1) expresses the transceiver capacity C (in bits/second) in terms of the channel bandwidth B [Hz], the signal power S [W], and the noise power N [W] [2, 10].

$$C = B \log_2\left(1 + \frac{S}{N}\right) \tag{2.1}$$

The ratio S/N is the signal-to-noise ratio (SNR), and measures the signal integrity, since it is proportional to the probability of recovering the information after its transmission [3].

In this thesis, only the receiver is considered, and inside it, a great attention is given to one building block: the low noise amplifier (LNA). The LNA is usually the first receiver block and is responsible for the amplification of the typically weak signals received at the antenna. It should be designed to amplify signal power Swithout degrading significantly the SNR - Shannon's equation (2.1).

This chapter is divided into six sections. Section 2.1 presents a short introduction to the LNAs working context. In section 2.2, basic wireless receiver architectures are presented and some building blocks surrounding the LNA are briefly referred. Section 2.3 presents the most important LNA parameters: input impedance matching, gain and bandwidth. The scattering-parameters are also introduced due to their importance in the LNA design. Noise performance is another fundamental LNA parameter, and due to its importance it has section 2.4 dedicated exclusively to it. Finally, section 2.5 describes the design of inductors and transformers, due to their importance as building elements of modern LNA circuits. In section 2.6 some conclusions are drawn.

## 2.2 Wireless Receiver Architectures

In figure 2.2 a division of a modern wireless receiver into an RF part and a digital part is represented. The RF part of the wireless receiver is responsible for the analog signal processing since this is received by the antenna till it is converted into a binary code at the analog-to-digital converter (ADC), being the remaining signal processing in the digital domain at the digital part.



Figure 2.2: Possible block diagram of a wireless receiver.

Theoretically, wireless receivers can work at any frequency; however, low frequencies lead to very long and non-portable antennas, and high frequencies suffer from high attenuation when the electromagnetic wave is propagating through open space. The frequency range of portable wireless receivers is bounded by these considerations and is typically between hundreds of MHz and tens of GHz [2].

With respect to signal bandwidth, wireless receivers can be divided into:

- narrowband receivers;
- multi-band receivers;
- wideband receivers.

In a narrowband receiver, the signal bandwidth is much lower than the carrier frequency [5]. A multi-band receiver can work at several narrowbands. It can work at all narrowbands simultaneously or one at a time. In a wideband receiver, it is possible not to use carrier and the signal bandwidth can be of the same order of magnitude of the central frequency [32].

A quantitative criterion to distinguish a wideband from a narrowband receiver is that the fractional bandwidth is higher than 0.2 for a wideband receiver:

$$2\frac{f_H - f_L}{f_H + f_L} > 0.2\tag{2.2}$$

where  $f_H$  and  $f_L$  are, respectively, the upper and the lower frequencies of the signal band [33, 34]. If a signal is wideband and its bandwidth is higher than 500 MHz, the signal is considered an ultra-wideband signal [34, 35].

## 2.2.1 Common Wireless Receiver Architectures

Most of the wireless receivers used nowadays are carrier-based and narrowband. This means that a high frequency carrier that is modulated by a signal, to be recovered should be down-converted at the receiver to a lower frequency before further processing. Narrowband receivers can be of two basic types [2–5, 7, 36–38]:

- homodyne or Zero-IF (direct conversion) receiver;
- heterodyne or IF (Intermediate Frequency) receiver.

The block diagram of figure 2.3 represents the different blocks of a narrowband receiver and is applicable both to homodyne and heterodyne receivers. The main building blocks are: antenna, low noise amplifier (LNA), local oscillator (LO), mixer, and analog-to-digital converter (ADC).



Figure 2.3: Narrowband wireless receiver.

The LNA is responsible for the amplification of the RF signal received by the antenna. The amplified RF signal is then multiplied by a sinusoid (generated by the LO) at the mixer. In an homodyne receiver, the LO sinusoid and the carrier frequency are equal, resulting in a baseband signal at the mixer output. The spurious components resulting from the mixing can then be filtered out (the dashed-line blocks in figure 2.3 are not used). In an heterodyne receiver, the LO frequency is different from the carrier frequency, resulting in an intermediate frequency at the mixer. To relax the filter specifications, more than one mixing/filtering stages can be used, each having a different LO frequency.

receiver, often used, is the low-IF receiver.

These architectures are the most commonly used; however, others can be found in literature [10, 18, 33, 35, 39, 40]. In the following, the wireless receiver building blocks are shortly described and a special emphasis is given to the LNA.

## 2.2.2 Receiver Building Blocks

### Low Noise Amplifier (LNA):

The LNA is the first block of a wireless receiver and its input is typically connected to an antenna. The distance between the two is commonly of the order of the wavelength, which means that the propagation effects require that connections are made using standard impedance transmission lines [7, 31]. To maximize the power transfer, the LNA and the antenna should be impedance matched<sup>1</sup>.

As the LNA is at the beginning of the receiver chain, the noise at its input and the noise generated at the LNA are amplified by the gain of the subsequent blocks. Since the input SNR is typically low, the LNA noise contribution should be as low as possible to prevent the degradation of the SNR, when the input signal is amplified. The LNA linearity is not a major concern because the overall wireless receiver linearity is dominated by the last blocks of the receiver chain, mainly the down-converting mixers [4, 38, 41]. LNAs should also have a good feed-forward isolation to prevent instability, and signal leakage from the subsequent blocks that might be reradiated by the antenna [4]. The LNA output impedance does not have to be impedance matched because the following receiver block, usually a mixer, is also integrated and the interconnection can be made short. Finally, LNAs are typically used in portable receivers and should be very efficient in terms of power use.

Chapter 3 is dedicated to the LNA implementation and characterization. There, an overview is made of some of the most common LNA topologies used nowadays.

<sup>&</sup>lt;sup>1</sup>The most typical value of characteristic impedance is 50  $\Omega$ , and this value corresponds to the characteristic impedance of coaxial cables, at which the power transfer capability and the attenuation loss have a better compromise. The exact number is merely conventional and is due to simplicity reasons [2].

## Local Oscillator (LO)

An oscillator generates a periodic signal. A voltage controlled oscillator (VCO) is an oscillator, the frequency of which is controlled by a voltage. The periodic signal generated by an LO is used by the mixer to down-convert the high frequency RF signal. One important specification of the LO is low phase noise to prevent down-conversion of the signals of adjacent channels [3, 5].

### Mixer

After the LNA, the signal is usually down-converted to a lower frequency by a mixer, which multiplies the LNA output signal by a constant frequency signal. The mixer is a three port block with two inputs and one output. One input receives the LNA output signal and the other input is the LO sinusoid; the output is the product of both input signals, which corresponds to an intermediate frequency IF in an heterodyne conversion and to the baseband signal in a direct conversion receiver [3, 5, 7, 38].

## Analog-to-Digital Converter (ADC)

The analog-to-digital converter converts the analog baseband signal into a digital signal. The main ADC specifications are the sampling frequency and the resolution; i.e., the number of bits of the output code [42].

## 2.3 Basic Concepts of RF Design

Linear circuits having multiple ports can be represented by linear combinations of the voltages and currents at these ports. The matrix description of linear circuits with two ports (2-port) is particularly useful to represent circuits having only one input and one output [43]. Using the general 2-port block, with its port voltages  $v_i$  and currents  $i_i$ , represented in figure 2.4, different description matrices can be obtained, their choice being dependent on the circuit where the 2-port block is inserted. In appendix A are listed the different matrices.



Figure 2.4: 2-port block.

When the signal wavelength is of the same order of the circuit dimensions or interconnects length, the circuit analysis should account for propagation effects [2, 3, 7, 8].

In this thesis, the LNA input is considered to be at a distance of the order of magnitude of the input signal wavelength; and thus, its input should be impedance matched to the input transmission line in order to maximize the power transfer from the signal source and simultaneously overcome undesired signal reflections. The expected LNA circuit dimension is much lower than the signal wavelength, so it can be treated as a lumped element circuit. The importance of impedance matching is discussed in the following.

#### 2.3.1 Impedance Matching and Reflection Coefficient

Figure 2.5 represents a transmission line with characteristic impedance  $Z_0$  excited by a high frequency signal generator  $V_s$  and terminated by a load impedance  $Z_L$ . The voltage and current have a spacial variation along the line [9]:

$$\begin{cases} V(x) = V^{+}e^{-j\beta x} + V^{-}e^{j\beta x} \\ I(x) = \frac{1}{Z_{0}}(V^{+}e^{-j\beta x} - V^{-}e^{j\beta x}) \end{cases}$$
(2.3)

where  $V^+$  and  $V^-$  are respectively the amplitudes of the incident and reflected voltage waves,  $\beta = 2\pi/\lambda$ ,  $\lambda$  is the signal wavelength and x is the position. The different signs of the exponents account for the opposite directions in which the incident and reflected waves travel.



Figure 2.5: Transmission line with characteristic impedance  $Z_0$  terminated with impedance  $Z_L$  and excited by voltage generator  $V_s$ .

The presence of impedance  $Z_L$  at position x = 0 makes the signal wave to be partially reflected towards the generator, and as a result of this reflection, next boundary conditions are obtained:

$$\begin{cases}
V(0) = V^{+} + V^{-} \\
I(0) = \frac{1}{Z_{0}}(V^{+} - V^{-}) \\
Z_{L} = \frac{V(0)}{I(0)}
\end{cases}$$
(2.4)

The ratio of  $V^-$  and  $V^+$  is the reflection coefficient  $\Gamma_L$  from (2.4) obtaining:

$$\Gamma_L = \frac{V^-}{V^+} = \frac{Z_L - Z_0}{Z_L + Z_0} \tag{2.5}$$

 $\Gamma_L$  corresponds to the reflection coefficient when observing from the transmission line to the load impedance  $Z_L$ .

The situation represented in figure 2.5 can be found at the LNA input (here represented by impedance  $Z_L$ ), where the presence of a reflected voltage  $V^-$  means that a voltage wave is propagated towards the antenna and reradiated.

This reradiation is undesired and should be minimized, by reducing the reflected wave  $V^-$  by proper impedance sizing. Making the load impedance equal to the transmission line characteristic impedance ensures that the reflection coefficient is zero and consequently the reflected voltage wave  $V^-$  is canceled. This situation corresponds to the maximum power transfer from the transmission line to the LNA input because the incident voltage amplitude is maximum.

The theory described here is used to define a form of representing 2-port considering propagation effects.

#### 2.3.2 Scattering Parameters

The scattering parameters or S-parameters allow a 2-port description in terms of incident and reflected power waves. Consider the 2-port represented in figure 2.6, in which ports 1 and 2 have respectively impedances  $Z_1$  and  $Z_2$ , and are connected to transmission lines of characteristic impedance  $Z_0$ . The 2-port is supplied by a signal source  $V_s$ , having an output impedance  $Z_S = Z_0$  connected directly to the transmission line. The 2-port has a load impedance  $Z_L = Z_0$  connected to the 2-port output via a transmission line. Due to reflections, each port has an incident and a reflected power at their ports represented respectively by  $P_m^+$  and  $P_m^-$ , where m is the port number [9].



Figure 2.6: 2-port block with incident and reflected waves.

The average power absorbed by each port is equal to the difference between both power components and can be determined by using the incident and reflected wave amplitudes in (2.4). The average power is defined as

$$P_{\rm av} = \frac{1}{2} \operatorname{Re}\{VI^*\}$$
(2.6)

Using equations of (2.4), the reflection coefficient (2.5) and the average power

definition (2.6), the average power absorbed by each port is:

$$P_1 = P_1^+ - P_1^- = \frac{1}{2} \frac{|V_1^+|^2}{Z_0} (1 - |\Gamma_i|^2)$$
(2.7)

$$P_2 = P_2^+ - P_2^- = \frac{1}{2} \frac{|V_2^+|^2}{Z_0} (1 - |\Gamma_o|^2)$$
(2.8)

where  $\Gamma_i$  and  $\Gamma_o$  are:

$$\Gamma_i = \frac{Z_1 - Z_0}{Z_1 + Z_0} \quad \text{and} \quad \Gamma_o = \frac{Z_2 - Z_0}{Z_2 + Z_0}$$
(2.9)

With (2.4) and (2.6), the incident and reflected power amplitudes can also be determined as a function of the voltages and currents at the respective port:

$$\begin{cases}
P_m^+ = \frac{1}{2} \operatorname{Re}\{V_m^+ I_m^{+*}\} = \frac{|V_m(0) + Z_0 I_m(0)|^2}{4Z_0} \\
P_m^- = \frac{1}{2} \operatorname{Re}\{V_m^- I_m^{-*}\} = \frac{|V_m(0) - Z_0 I_m(0)|^2}{4Z_0}
\end{cases}, m \in \{1, 2\} \quad (2.10)$$

Again,  $Z_0$  must be real to consider (2.10) valid. Consider now the square root of the incident and reflected power waves, also known as normalized incident and normalized reflected power waves, and represented by  $a_m$  and  $b_m$  respectively. Using (2.10) they are:

$$\begin{cases} a_m = \sqrt{P_m^+} = \frac{V_m + Z_0 I_m}{2\sqrt{Z_0}} \\ b_m = \sqrt{P_m^-} = \frac{V_m - Z_0 I_m}{2\sqrt{Z_0}} \end{cases}$$
(2.11)

The S-parameters relate  $a_m$  and  $b_m$  of each port by means of a matrix defined as:

$$\begin{bmatrix} b_1 \\ b_2 \end{bmatrix} = \begin{pmatrix} S_{11} & S_{12} \\ S_{21} & S_{22} \end{pmatrix} \begin{bmatrix} a_1 \\ a_2 \end{bmatrix}$$
(2.12)

Determining  $S_{11}$  with (2.4), (2.11) and (2.12)

$$S_{11} = \frac{b_1}{a_1} \bigg|_{a_2=0} = \frac{V_1 + Z_0 I_1}{V_1 - Z_0 I_1} = \Gamma_i$$
(2.13)

It is concluded that  $S_{11}$  is equal to the reflection coefficient  $\Gamma_1$ . It can be shown that  $S_{22} = \Gamma_o$ . This means that  $S_{11}$  and  $S_{22}$  measure the ratio between the reflected and incident power waves at port 1 and 2 (when no incident wave is present on the other port), being measures of the input and output 2-port impedance matchings.

 $S_{21}$  is a voltage gain between the incident voltage at port 1 and the reflected voltage at port 2. Using (2.4), (2.11) and (2.12),  $S_{21}$  is

$$S_{21} = \left. \frac{b_2}{a_1} \right|_{a_2=0} = \frac{V_2 - Z_0 I_2}{V_1 + Z_0 I_1} = \frac{V_2^-}{V_1^+}$$
(2.14)

 $S_{12}$  is known as the reverse voltage gain as it measures the ratio between the incident voltage at port 2 and the reflected voltage at port 1. As determined for  $S_{21}$ ,  $S_{12}$  is

$$S_{12} = \left. \frac{b_1}{a_2} \right|_{a_1=0} = \frac{V_1 - Z_0 I_1}{V_2 + Z_0 I_2} = \frac{V_1^-}{V_2^+} \tag{2.15}$$

### 2.3.3 Power Gain

Due to the power reflections at the 2-port input and output, different power gain values can be determined. The most used are [7, 9]:

• Transducer Power Gain:

$$G_T = \frac{\text{power delivered to the load}}{\text{power available from the source}} = \frac{P_2^- - P_2^+}{P_s}$$
(2.16)

• Power Gain (Operating Power Gain):

$$G = \frac{\text{power delivered to the load}}{\text{power supplied to the 2-port}} = \frac{P_2^- - P_2^+}{P_1^+ - P_1^-}$$
(2.17)

• Available Power Gain,  $G_A$ , is the ratio between the power at the output of the 2-port and the power delivered by the power source:

$$G_A = \frac{\text{power available at the 2-port output}}{\text{power available from the source}} = \frac{P_2^-}{P_s}$$
(2.18)

Since these ratios can vary by several orders of magnitude, it is convenient to express them in dB.

$$G_{\rm dB} = 10 \log_{10} G \tag{2.19}$$

### 2.3.4 Signal Distortion and Linearity

Distortion results from circuit non-idealities. Consider the system represented in figure 2.7 and two input signals:  $x_1(t)$  and  $x_2(t)$ .

$$x(t) \longrightarrow f(x(t)) \longrightarrow y(t)$$

Figure 2.7: System modeling f(x(t)) having input x(t) and output y(t).

Consider now the system excited individually by two different inputs  $x_1(t)$  and  $x_2(t)$ , originating respectively  $y_1$  and  $y_2$ . The system is linear if,

$$ax_1(t) + bx_2(t) \to ay_1(t) + by_2(t)$$
 (2.20)

where a and b are constant [3]. When the system is non-linear, time invariant, and memoryless, and function f(x(t)) has all derivatives till m order around a point of operation, it is possible to represent the system by a Taylor series till m degree [44] [45]:

$$y(t) = K_1 x(t) + K_2 x(t)^2 + K_3 x(t)^3 + \dots + K_m x(t)^m$$
(2.21)

where  $K_m$  are the Taylor coefficients [44]:

$$K_m = \frac{1}{m!} \frac{\partial^m f}{\partial x^m} \tag{2.22}$$

Usually, only the first three terms of the Taylor series are considered. These three terms represent a compromise between analysis complexity and approximation error [46].  $K_1$  is the system gain, and  $K_2$  and  $K_3$  are respectively the second-order and the third-order nonlinearity coefficients, respectively [45]. A practical measure of linearity is the third-order intercept point (IIP<sub>3</sub>) and in this case a double tone input signal is considered:

$$x(t) = A\cos\omega_1 t + A\cos\omega_2 t \tag{2.23}$$

where  $\omega_1$  and  $\omega_2$  are two frequencies inside the bandwidth of operation. Substituting (2.23) in (2.21) and considering only the first three terms, y(t) becomes:

$$y(t) = (K_1 + \frac{9}{4}K_3A^2)A\cos\omega_1 t + (K_1 + \frac{9}{4}K_3A^2)A\cos\omega_2 t +$$

$$+ \frac{3}{4}K_3A^3\cos(2\omega_1 - \omega_2)t + \frac{3}{4}K_3A^3\cos(2\omega_2 - \omega_1)t$$
(2.24)

Assuming weak distortion  $(K_1 \gg \frac{9}{4}K_3A^2)$ , the amplitude A for which the output components at  $\omega_1$  (and  $\omega_2$ ) equal the components at  $2\omega_1 - \omega_2$  (and  $2\omega_2 - \omega_1$ ) is the third-order intercept point IIP<sub>3</sub> [3]:

$$IIP_{3} = \sqrt{\frac{4}{3} \frac{K_{1}}{K_{3}}}$$
(2.25)

A graphical definition of IIP<sub>3</sub> is represented in figure 2.8. The  $P_{-1dB}$  is the 1 dB compression point, which corresponds to the input power for which the output power is 1 dB below the expected, and the dynamic range is the power range between the noise floor and the  $P_{-1dB}$ .



Figure 2.8: Definition of  $IIP_3$ .

## 2.3.5 Bandwidth

Frequency bandwidth of a system is the range of signal frequencies for which the gain has a specified maximum loss (typically 3 dB) with respect to the maximum value. The bandwidth of a signal is the frequency range where almost all signal power (usually 98 or 99%) is located [3]. The signal bandwidth is usually represented by B or by  $\Delta f$ .

## 2.4 Noise

In electronics, noise is a random fluctuation of a voltage or a current within a circuit [47]. Noise has different sources, having different physical explanations. Most of these sources can be characterized statistically.

In wireless receivers, noise can be divided into external and fundamental noise [48]. External noise is generated outside the receiver, for instance electromagnetic interference in the transmission channel. This kind of noise can be minimized by good electromagnetic shielding and adequate system architecture. Fundamental noise results from physical phenomena in the electronic devices. Several types of fundamental noise are identified: thermal noise, flicker noise and others.

## 2.4.1 Noise and the Wiener-Khintchine Theorem

The frequency spectrum of a deterministic signal is determined by its Fourier transform, defined as:

$$X(f) = \int_{-\infty}^{+\infty} x(t)e^{-j2\pi ft}dt$$
(2.26)

In the case of noise, it is possible to determine the noise power distribution in the frequency domain - power spectral density (PSD), N(f). Its definition is [3, 47]:

$$N(f) = \lim_{T \to \infty} \frac{\overline{|X_T(f)|^2}}{T}$$
(2.27)

where  $X_T(f)$  is:

$$X_T(f) = \int_0^T x_n(t) e^{-j2\pi f t} dt$$
(2.28)

Consider now the 2-port represented in figure 2.9, which has a network function H(f) and is excited by a noise source with PSD  $N_x(f)$ . The output power spectral density  $N_y(f)$  is given by:

$$N_y(f) = |H(f)|^2 N_x(f)$$
(2.29)

The relation (2.29) is known as the Wiener-Khintchine theorem and its demonstration can be found in [3, 7, 41, 47, 49].



Figure 2.9: 2-port with transfer function H(f) excited by noise source with power spectral density  $N_x(f)$ .

#### 2.4.2 Noise Figure

The most common noise performance measure used in RF for 2-ports is the noise factor, F, or noise figure, NF, when expressed in dB. It is defined as the ratio between the 2-port total output noise power and the 2-port output noise power due only to the 2-port input noise:

$$F = \frac{\text{Total Output Noise Power}}{\text{Output Noise Power due to the Source}}$$
(2.30)

The noise factor as defined above can also be determined by the ratio between the input and output signal-to-noise ratio,  $SNR_i$  and  $SNR_o$  respectively:

$$F = \frac{\mathrm{SNR}_i}{\mathrm{SNR}_o} \tag{2.31}$$

The SNR is defined as the ratio of the average signal power S and the average noise power N and is independent of frequency. Thus, the noise factor as defined above does not give information about the circuit noise performance along the frequency; however, (2.31) is a good measure of the SNR degradation that a (noisy) signal suffers when processed by a 2-port. Therefore, the noise factor in (2.31) is also known as excess noise factor [2, 47, 50–52].

To express the noise performance as a function of frequency, other noise factor expressions have been proposed, of which the most used is the spot noise factor. To determine the spot noise factor, consider the noisy 2-port of figure 2.10, excited by source characterized by a signal power  $S_i(f)$  and a noise power  $N_i(f)$ .

The source has an available signal power, represented hereafter by  $S_i(f)$ , which is frequency dependent. The source also has available noise power, frequency dependent, represented by  $N_i(f)$ . The available power is the (signal or noise) power that



Figure 2.10: Noisy 2-port with available power gain  $G_A(f)$  excited by signal and noise source.

would be supplied by the source to a matched load impedance [50]. The source is loaded by a 2-port which has an available power gain  $G_A(f)$  and that generates a noise power that reported to the output has a value of  $N_{p,o}$ . The spot noise factor F is defined as [52]:

$$F(f) = \frac{\frac{S_i(f)}{N_i(f)}}{\frac{G_A(f)S_i(f)}{G_A(f)N_i(f) + N_{p,o}(f)}}$$
(2.32)

The numerator of (2.32) is the ratio of the spectral densities of the signal and noise powers at the input, while the denominator corresponds to the same ratio but at the output. Note that the noise power at the output has two terms, one due to the input  $G_A(f)N_i(f)$  and one due to the 2-port itself,  $N_{p,o}$ . If this last term were zero, i.e., if the 2-port were noiseless, the noise factor would be 1. (2.32) can be simplified to:

$$F(f) = 1 + \frac{N_{p,o}(f)}{G_A(f)N_i(f)} = 1 + \frac{N_{p,i}(f)}{N_i(f)}$$
(2.33)

 $N_{p,i}$  is the noise power generated by the 2-port and reported to its input. The noise generated inside a 2-port can be represented by the equivalent input noise voltage  $v_{n,i}$  and the equivalent input noise current  $i_{n,i}$ , as shown in figure 2.11. The source has a resistance  $R_S$  that generates a noise voltage  $v_{n,S}$ .

The noise current source can be transformed into a voltage by multiplying it by the source impedance. The equivalent input noise voltage, considering the source transformations of appendix C, becomes:

$$v'_{n,i} = v_{n,i} + R_S i_{n,i} \tag{2.34}$$



Figure 2.11: 2-port circuit block with equivalent noise sources.

The input noise power spectral density due to the 2-port is determined by the Wiener-Khintchine theorem (2.29):

$$N_i(f) = N_{v_{n,i}}(f) + R_S^2 N_{i_{n,i}}(f)$$
(2.35)

 $N_{v_{n,i}}$  and  $N_{i_{n,i}}$  are the power spectral density due to the 2-port input noise voltage and current respectively. The noise factor is finally:

$$F(f) = 1 + \frac{N_i(f)}{N_S(f)}$$
(2.36)

where  $N_S(f)$  is the power spectral density of the noise generated by the source resistance  $R_S$ .

#### 2.4.3 Cascaded Systems

A wireless system can be analyzed as a cascade of 2-ports, as represented in figure 2.12. The equations of the noise factor F and the IIP<sub>3</sub> of the cascade, give an idea of the relative importance of the different 2-ports on the overall performance.



Figure 2.12: Cascade of 2-ports.

The noise factor referred to the input of the first block is [3, 36, 50]:

$$F = F_1 + \frac{F_2 - 1}{G_{A1}} + \frac{F_3 - 1}{G_{A2}G_{A1}} + \dots + \frac{F_n - 1}{\prod_{i=1}^n G_{Ai}}$$
(2.37)

This is known as Friis Law, and shows that the gain of the first blocks reduces the influence of the noise from last blocks, meaning that the first blocks have a greater contribution to the overall cascade noise factor.

The IIP<sub>3</sub> equation for the cascade of figure 2.12 is [3]:

$$\frac{1}{\text{IIP}_3} = \frac{1}{\text{IIP}_{3,1}} + \frac{G_{A1}}{\text{IIP}_{3,2}} + \frac{G_{A1}G_{A2}}{\text{IIP}_{3,3}} + \dots + \frac{\prod_{i=1}^n G_{Ai}}{\text{IIP}_{3,n}}$$
(2.38)

The conclusion for linearity is the opposite of that for noise; i.e., the last cascade blocks have more influence on IIP<sub>3</sub> than the first ones. Thus, from (2.37) and (2.38) it is concluded that the 2-ports located at the beginning of the cascade have more influence on the noise performance, while the last ones are more relevant for the linearity performance. As the LNA is the first 2-port of the cascade, its'noise factor is a more important specification than linearity.

#### 2.4.4 Noise Sources

Several noise sources are identified in electronic components. The most important for RF design are thermal noise, flicker noise and shot noise.

#### Thermal Noise

Thermal noise occurs because thermally excited electrons, when traveling between two points of a conducting material, have a brownian movement due to its collisions within the fixed atoms [53, 54]. This phenomenon produces voltage fluctuations at the device terminals. The noise power in a conductor is independent of the material and frequency, and depends only on the temperature and resistance. At thermal equilibrium, the noise power spectral density is [47]:

$$N(f) = 4k_B T R \tag{2.39}$$

 $k_B$  is the Boltzmann constant<sup>2</sup> ( $k_B = 1.380 \times 10^{-23} \text{ JK}^{-1}$ ), R is the resistance and T the absolute temperature in Kelvin [K].

#### Flicker Noise or 1/f Noise

Two theories are found in literature to explain flicker noise [47, 48, 53, 55]:

- The carrier number fluctuation theory explains the flicker noise as the random trapping and release of charges in the oxide near the Si-SiO<sub>2</sub> interface, which causes surface potential variations. This affects the channel carrier density and, consequently, the current flowing in the transistor channel.
- The **mobility fluctuation theory** explains flicker noise as a bulk mobility fluctuation.

Flicker noise is proportional to the inverse of frequency, and that is why this noise is also known as 1/f noise. This type of noise is more relevant for mixers that perform down-conversions. As the LNA works at high frequencies, flicker noise is negligible in comparison with other noise sources [53].

#### Shot Noise

Electrical current is the result of a flow of discrete electrical charges - electrons. The electrical current, at a certain section, has a discrete fluctuation which is a multiple of a minimum value - the electron charge q ( $q = 1.602176 \times 10^{-19}$  C) [54]. There are two conditions that should be verified for the occurrence of shot noise:

- flow of current through a device;
- existence of a potential barrier through which the electrons jump.

The second occurs in p-n junctions and the shot noise results from the randomness of the time of arrival of one unit charge to the boundary of the barrier. The noise power spectral density of shot noise is [2, 55]:

$$N(f) = 2qI_{\rm DC} \tag{2.40}$$

<sup>&</sup>lt;sup>2</sup>The Boltzmann constant is represented in this thesis with an indice B to distinguish it from the transformer magnetic coupling coefficient represented by k.

where q is the electron charge and  $I_{\rm DC}$  is the DC current that flows through the device.

## 2.4.5 Noise in Transistors

#### Noise in MOS Transistors:

For LNA design, two noise sources are considered in MOS transistors: the thermal noise generated in the gate resistance  $R_g$  and represented by a voltage source  $v_{n,R_g}$ , and the thermal noise generated by the channel admittance  $g_{d0}$  and represented by a current source  $i_{n,d}$ ; both sources are represented in figure 2.13.



Figure 2.13: MOS transistor noise sources for RF design.

The noise power spectral densities of the two noise sources are respectively [53]:

$$\begin{cases} N_{R_g}(f) = 4k_B T R_g \\ N_{i_d}(f) = 4k_B T \gamma g_{d0} \end{cases}$$
(2.41)

where  $g_{d0}$  is the zero-bias drain conductance and is related with the transistor transconductance  $g_m$  by a constant  $\alpha$  ( $g_m = \alpha g_{d0}$ ). For long channel transistors  $\alpha = 1$  [47, 56].  $\gamma$  is a (dimensionless) bias dependent factor ( $\gamma$  is 2/3 for long channel transistors and higher for short channel transistors) [56]. The equivalent input noise sources  $v_n$  and  $i_n$ , in figure 2.14, are [41]:

$$\begin{cases} v_n = v_{n,Rg} + \frac{1}{g_m} \left( 1 + \frac{R_g}{Z_{gs}} \right) i_{n,d} \\ i_n = i_{n,d} \end{cases}$$
(2.42)

where  $Z_{gs}$  is the impedance due to the gate-source capacitance.



Figure 2.14: Equivalent input noise sources in a MOS transistor.

#### Noise in Bipolar Transistors:

In a BJT, there are the following noise sources: the thermal noise voltage  $v_{n,R_b}$  produced by the base resistance  $R_b$ , the base shot noise current  $i_{n,b}$  and the collector shot noise current  $i_{n,c}$ . These noise sources are represented in figure 2.15:



Figure 2.15: BJT noise sources for RF design.

The represented noise sources have the following noise power spectral noise densities [57]:

$$\begin{cases}
N_{R_b}(f) = 4k_B T R_b \\
N_{I_B}(f) = 2q I_B \\
N_{I_C}(f) = 2q I_C
\end{cases}$$
(2.43)

where  $R_b$  is the base resistance of the transistor and  $I_B$  and  $I_C$  are, respectively, the DC base current and the DC collector current. It is possible to determine the equivalent transistor input current and voltage noise sources in accordance with figure 6.25:

$$\begin{cases} v_n = v_{n,R_b} + \frac{1}{g_m} \left( 1 + \frac{R_b}{Z_{be}} \right) i_{n,c} \\ i_n = i_{n,b} + \frac{1}{g_m Z_{be}} i_{n,c} \end{cases}$$
(2.44)



Figure 2.16: BJT with equivalent input referred noise sources.

where  $Z_{be}$  is the base-emitter capacitance.

## 2.5 Integrated Inductors and Transformers

This section gives an overview of integrated inductors and transformers. It starts with a discussion about inductors realized in integrated technologies. After, a generic inductor model, using lumped element models is presented and its most relevant limitations are discussed. Finally, the realization of integrated transformers using different topologies is shortly discussed.

## 2.5.1 Integrated Inductors

An inductor is usually realized as a coil of conducting material, which generates a magnetic field H when current i passes through it. The inductance L is defined by the magnetic flux  $\Psi$  generated by the current i, where  $\Psi$  is [58]:

$$\Psi = Li \tag{2.45}$$

L depends only on the circuit geometry and magnetic field permittivity.

In an integrated circuit, an inductor can be as a lumped element when implemented with one or several turns of metal layers [59]. Another implementation consists on using transmission lines; however, integrated circuits are not long enough for the required lengths [60].

As the number of metal layers is limited, the preferred geometry is the spiral coil, which minimizes the number of metal layers to two: one for the coil itself and one for the connection of the inner spiral point. The spirals can be circular, square or polygonal. The square geometry is not the most area efficient; however it is the simplest to implement and one of the simplest to model [2, 61].

In figure 2.17 the 2-D view of a spiral square inductor in a silicon process is represented, and in figure 2.18 the 3-D view is shown. The following design parameters are indicated in figures 2.17 and 2.18 [62]:

- external diameter  $d_{out}$ ;
- edge-to-edge spacing between adjacent turns s;
- turn metal width w;
- number of turns  $n_{turn}$ ;

 $\bullet\,$  distance between the metal layer and the substrate a.



Figure 2.17: 2-D view of an integrated inductor.



Figure 2.18: 3-D view of an integrated inductor.

The metal thickness e is a technology parameter that cannot be dimensioned by the designer.

In figure 2.19 the equivalent single- $\pi$  model of an integrated inductor is represented. It includes the following elements [2, 63, 64]:

- series inductance L;
- coil series resistance  $R_s$ ;
- inductor-substrate capacitance  $C_{ox}$ ;
- substrate capacitance  $C_{si}$ ;

• substrate spreading resistance  $R_{si}$ .



Figure 2.19:  $\pi$  circuit model of an integrated spiral inductor.

The effect of the inter-turn fringing capacitance is usually small because the adjacent turns are almost equipotential - the wave length is much larger than the inductor dimensions [65].

Much work was done in order to determine the inductor model elements. The first reported work (widely referenced) about the inductance determination of a square spiral inductor is given in [61]; however this method does not provides simple equations. One simple equation to determine an approximation of the inductance value can be found in [2]:

$$L \approx \mu_0 r n^2 \tag{2.46}$$

where  $\mu_0$  is the air magnetic constant ( $\mu_0 = 4\pi \times 10^{-7} \text{ NA}^{-2}$ ), *n* is the number of turns and  $r = d_{\text{out}}/2$  is the radius. More complex and more approximate equations can be found. One, only applicable to square spirals, is [66]:

$$L \approx \frac{2\mu_0 n^2 d_{\text{avg}}}{\pi} \left( \ln \frac{2.067}{\rho} + 0.178\rho + 0.125\rho^2 \right)$$
(2.47)

where  $d_{\text{avg}}$  is the arithmetic mean of the inner  $d_{\text{in}}$  and outer diameters  $d_{\text{out}}$  and  $\rho$  is

$$\rho = \frac{d_{\text{out}} - d_{\text{in}}}{d_{\text{out}} + d_{\text{in}}} \tag{2.48}$$

Nowadays inductances are determined for any geometry, using electromagnetic simulators that solve Maxwell's equations by computational methods [67].

Equations for other model parameters can also be found in literature. The series resistance is given by [2]:

$$R_s = \frac{l}{w\sigma\delta(1 - e^{-t/\delta})} \tag{2.49}$$

where  $\sigma$  is the metal conductivity, l, w and t are, respectively, the total length, the width and the thickness of the winding, and  $\delta$  is the skin effect parameter

$$\delta = \sqrt{\frac{2}{\omega\mu_0\sigma}} \tag{2.50}$$

The inductor-substrate capacitance  $C_{\text{ox}}$  is [2]:

$$C_{\rm ox} = w l \frac{\varepsilon_{\rm ox}}{t_{\rm ox}} \tag{2.51}$$

The product wl is the winding metal area,  $\varepsilon_{\text{ox}}$  is the oxide permittivity and  $t_{\text{ox}}$  is the oxide thickness measured from the substrate to the winding.

Integrated silicon technologies are not optimized for inductor design. They have typically a limited number of metal layers and the thickness between them is not controllable, limiting the inductors optimization. Besides, the low substrate resistance degrades inductances and the silicon cost proportional to area also demands a reduction in the number of inductors, due to their low inductance per square unit in comparison to the inductances required for design. Thus, the inductor design results from the compromise between these constrains. Some practical design guidelines of an integrated inductor are given in the following [2, 60].

#### Practical Considerations on Integrated Inductor Design:

- The use of the top metal layer decreases simultaneously the inductor series resistance  $R_s$  and the oxide capacitance  $C_{\text{ox}}$ .  $R_s$  is lowered because the top metal layer is in some technologies, thicker than the others and  $C_{\text{ox}}$  is minimized because the top layer is further away from the substrate.
- In a multilevel metal process it is possible to use different metal layers to increase the inductance value and simultaneously optimize the inductor area.

- It is possible to connect two or more metal layers, by using vias, to lower the series resistance. The penalty is an increase in oxide capacitance, due to the use of lower metal layers.
- If the inductor is placed over a doped substrate, the magnetic field penetrates into the substrate, inducing loop currents that degrades the self inductance of the spiral inductor [65]. To prevent the electric field from penetrating the substrate a patterned ground shield can be used. The shield has to be patterned to avoid the magnetically induced eddy currents to form in the substrate and the patterned slots should be sufficiently narrow to prevent the vertical electric field to penetrate the substrate. One trade-off is an increase in the inductor substrate capacitance because of the reduction of the effective oxide thickness [62]. In figure 2.20 a possible patterned ground shield is presented.



Figure 2.20: 2-D view of a patterned ground shield.

If two inductors are close to each other, there is magnetic coupling between them that should be minimized. However, this magnetic coupling can also be used to design integrated transformers, as described next.

## 2.5.2 Integrated Transformers

Integrated transformers are realized using two spiral inductors magnetically coupled. Different transformer configurations, can be obtained according to the inductors design and relative position between the two inductors. The most common configurations are [62]:

- tapped transformer,
- interleaved transformer, and
- stacked transformer.

They are described next.

## 2.5.2.1 Common Integrated Transformer Configurations

### Tapped Transformer:

The tapped transformer, represented in figure 2.21, has both windings in same metal layer, which is suitable for technologies with few metal layers. This configuration also minimizes the capacitance between windings. The magnetic coupling coefficient is low [68].



-

Figure 2.21: 2-D view of a tapped transformer.

## Interleaved Transformer:

The interleaved transformer is shown in figure 2.22. Both windings are realized in the same metal layer. The capacitance between windings is higher than in the tapped transformer, due to the interleaving. It is also characterized by a weak magnetic coupling and less flexibility in setting the number of turns and relative size of the two windings.





Figure 2.22: 2-D view of an interleaved transformer.

### Stacked Transformer:

Another configuration for an integrated transformer consists of stacking two spiral inductors as represented in the 3-D view in figure 2.23. This configuration requires more metal layers than the two configurations presented before; however, it achieves higher magnetic coupling. The magnetic coupling coefficient can be controlled by displacing one of the windings relatively to the other, as represented in figure 2.24. This reduces the capacitance between windings.



Figure 2.23: 3-D view of a stacked transformer.



Figure 2.24: 2-D view of a stacked transformer in which the windings are shifted to control the magnetic coupling coefficient.

### 2.5.2.2 Transformer Equations

An ideal transformer is described by the following equations:

$$\begin{cases} v_1 = nv_2 \\ i_2 = ni_1 \\ v_1i_1 = v_2i_2 \end{cases}$$
(2.52)

 $v_m$  and  $i_m$  are respectively the voltage and current at port m. The ideal transformer symbol of figure 2.25 is used from now on.



Figure 2.25: Symbol of an ideal transformer.

Practical integrated transformers are implemented with two windings magnetically coupled. The magnetic coupling coefficient k varies between zero, for independent inductors, and one, for perfect magnetic coupling. A transformer considering winding inductances is represented in figure 2.26.  $L_{11}$  and  $L_{22}$  are, respectively, the self-inductances of windings 1 and 2.



Figure 2.26: Transformer 2-port symbol considering winding inductances  $L_{mm}$ .

The equations that describe the transformer of figure 2.26 are:

$$\begin{cases}
V_1(s) = sL_{11}I_1(s) + sL_MI_2(s) \\
V_2(s) = sL_MI_1(s) + sL_{22}I_2(s)
\end{cases}$$
(2.53)

where  $L_M = k\sqrt{L_{11}L_{22}}$  is the mutual inductance, which has an upper limit  $\sqrt{L_{11}L_{22}}$ , when k = 1 [69]. Using (2.53), and matrix **H** definition of appendix A, it can be shown that the 2-port **H** matrix of this transformer is

$$\mathbf{H}_{T} = \begin{pmatrix} sL_{11}(1-k^{2}) & \frac{1}{n} \\ & & \\ -\frac{1}{n} & \frac{1}{sL_{22}} \end{pmatrix}$$
(2.54)

where n is

$$n = \frac{1}{k} \sqrt{\frac{L_{22}}{L_{11}}} \tag{2.55}$$

This matrix representation is used in feedback systems when analyzed using matrices.

## 2.6 Conclusions

This chapter reviews the most important theoretical aspects of an LNA design. LNAs are part of modern wireless receivers and are responsible for the amplification of signals received from the antenna. The amplified signal is usually downconverted by a mixer connected ahead of the LNA. The importance of the LNA input impedance matching on the maximization of the power transfer from the antenna is emphasized. The LNA should have enough gain to amplify the weak RF signals received by the antenna and reduce the noise influence of other receiver blocks - Friis Law. The LNA should also generate low noise to avoid a degradation of the SNR of the input signal. Finally, some aspects of the technological implementation of the LNA are discussed, mainly concerning integrated inductors and transformers.

# Chapter 3

# **Overview of Low Noise Amplifiers**

3.1	Introduction	43
3.2	LNA Basic Concepts	44
3.3	LNA Topologies	45
3.4	Cascode LNA with Inductive Degeneration	51
3.5	Double Loop Feedback LNAs	56
3.6	Conclusions	71
# 3.1 Introduction

This chapter is an overview of LNA design. LNAs are characterized by three main parameters: input impedance, gain, and noise figure. Typically, the input impedance should have a precise value, while the gain should be maximized with the minimum noise addition. Several circuits achieve a precise input impedance; however, not all are suitable for LNA design.

LNA circuits have typically an amplifying block and a resonant network to achieve the required input impedance. These two blocks may form sometimes a single loop feedback network. Single loop feedback circuits are studied in the literature and an overview is presented in this chapter. Double loop feedback (DLF) circuits are studied in this chapter to evaluate their suitability to LNA design.

The major LNA parameters are discussed in section 3.2 and the typical LNA topologies are presented in section 3.3. The common-source LNA using inductive degeneration is the most widely used, and it is described in more detail in section 3.4. Double loop feedback topologies are described in section 3.5. In section 3.6 some final remarks are discussed.

# 3.2 LNA Basic Concepts

In the LNA design there is a trade-off between its main specifications [9]:

- input impedance matching;
- high gain in the bandwidth of interest;
- low noise factor;
- high frequency operation;
- bandwidth.

The ideal LNA has a finite input impedance matched to the output impedance of the preceding block, in order to maximize the power transfer of the received high frequency signal. This signal should be amplified without noise addition. The LNA has a bandwidth specification and it is often included in a portable system which limits the power consumption.

Usually, the output impedance of the block preceding the LNA is purely resistive and equal to 50  $\Omega$  [2]; thus, the LNA input impedance should also be 50  $\Omega$ , to prevent undesired reflections, and, consequently, to maximize the signal power transfer. The LNA gain is important to amplify the typically low received signals. A low noise factor is another important characteristic because the LNA is usually at the beginning of the RF receiver chain, so all the noise that it produces is amplified by the whole chain. Linearity has a lower priority in comparison with other characteristics: it is more important for the blocks situated at the end of the RF system chain, because they deal with larger signals. Concerning bandwidth, LNAs can be narrowband, multi-band or wideband.

The most important LNA specifications are the input impedance, the noise factor and the (voltage or current) gain, and few circuits can satisfy these specifications simultaneously at the required frequency band. In the following sections, an overview on the most used LNA topologies is presented.

## 3.3 LNA Topologies

Of the three most important LNA characteristics (input impedance, noise factor and gain), only the input impedance should have a precise value; i.e., it should equal the output impedance of the preceding block. The noise factor should be below a specific value and the gain should be above a specific value. Thus, LNAs are designed to have a precise input impedance with high gain and low noise figure.

LNAs can be implemented in several technologies, and in this thesis two technologies are considered: MOS and bipolar transistors. MOS circuits will be considered here, and the simple transistor incremental model in figure 3.1 will be adopted. It includes only the transconductance  $g_m$  and the gate-source capacitance  $C_{gs}$ . Other elements, like the gate resistance  $R_g$  and the gate-drain capacitance  $C_{gd}$  produce second-order effects that are only relevant for more detailed analysis, and are considered later in this chapter.



Figure 3.1: MOS incremental  $\pi$  model.

Several forms of obtaining CMOS LNAs are presented next.

### 3.3.1 LNA using Input Resistor

The simplest form of obtaining real input impedance is to place a resistor in parallel with the amplifying block, as shown in figure 3.2 [56].

The input impedance  $Z_{in}$  is  $R_{S1}/Z_1$ . If  $Z_1 \gg R_{S1}$ , the input impedance  $Z_{in}$  is approximately equal to  $R_{S1}$ , and there is input impedance matching, if  $R_{S1} = R_S$ .

Assuming that the 2-port is noiseless, the only noise contribution is the thermal



Figure 3.2: LNA with resistive input power matching.

noise generated by  $R_{S1}$ . The noise factor is [2, 56]:

$$F = \frac{N_S + N_{S1}}{N_S} = \frac{R_S + R_{S1}}{R_S} = 2$$
(3.1)

where  $N_S$  and  $N_{S1}$  are the noise power spectral densities due to the source resistance  $R_S$  and the input matching resistance  $R_{S1}$ , respectively. (3.1) shows that having a resistor in parallel with the LNA input impedance, of a value equal to the source impedance, leads to a noise figure penalty of 3 dB. A 3 dB noise figure is used as a boundary, below which, the LNA has a low NF.

#### 3.3.2 Common-Gate LNA

The common-gate LNA, shown in figure 3.3, uses the transistor transconductance to realize the LNA input impedance matching [56, 70–73]. Replacing the common-gate transistor by the incremental model of figure 3.1, leads to:

$$Z_{in} \approx \frac{1}{g_m} \tag{3.2}$$



Figure 3.3: Incremental model of the common-gate LNA (biasing not represented). Concerning the noise analysis and considering the CMOS transistor noise sources (2.41), the noise power spectral density due to the common-gate transistor referred at its input can be determined as:

$$N_i(f) = 4k_B T (R_g + R_S^2 \gamma g_m) \tag{3.3}$$

 $R_g$  is the gate resistance of  $M_1$  and  $\gamma$  is a bias dependent factor ( $\gamma$  is 2/3 for long channel transistors and higher for short channel transistors [56]). Using (3.3) and (2.41), the noise factor is:

$$F = \frac{N_S + N_i}{N_S} = 1 + \frac{R_g}{R_S} + R_S \gamma g_m$$
(3.4)

Assuming that  $R_g = 0$  and  $\gamma = 2/3$ , this LNA topology has a minimum noise factor of 5/3; i.e.,  $NF \approx 2.2$  dB.

One limitation of this topology is that the voltage gain cannot be optimized because  $g_m$  is fixed to ensure the input impedance matching.

### 3.3.3 Common-Source LNA with Inductive Degeneration

Another LNA topology, first proposed in [56], is represented in figure 3.4: it is a common-source stage with inductive source degeneration.



Figure 3.4: Common-Source LNA using inductive degeneration (biasing of  $M_1$  not represented).

From the incremental model in figure 3.5, the input impedance  $Z_{in}$  is:

$$Z_{in} = \frac{g_m L_S}{C_{gs}} + \frac{1}{sC_{gs}} + sL_S$$
(3.5)



Figure 3.5: Incremental model of the common-source LNA using inductive degeneration.

The input impedance is real at the resonant frequency:

$$Z_{in} = \frac{g_m L_S}{C_{gs}} \qquad \text{at} \qquad \omega_0 = \frac{1}{\sqrt{L_S C_{gs}}} \tag{3.6}$$

The real part of the input impedance depends on three different parameters  $(g_m, C_{gs} \text{ and } L_S)$  which are theoretically noiseless: this means that it is possible to achieve a real input impedance without addition of noise using this topology. The LNA gain is proportional to  $g_m$ , which can be maximized, while the values of  $C_{gs}$  and  $L_S$  are used to obtain the required input impedance. This is not possible in the common-gate LNA, where the maximization of  $g_m$  affects the input impedance.

#### 3.3.4 LNA with Current Reuse

The LNA using current reuse is a variant of the common-source LNA using inductive degeneration. This topology, represented in figure 3.6, uses both PMOS and NMOS transistors in a common-source configuration with inductive degeneration. This topology achieves the same noise factor and the same transconductance of the single NMOS transistor common-source LNA using half the current consumption. However, this topology requires an higher voltage supply to feed the two stacked transistors. [74–76].



Figure 3.6: LNA using current reuse (biasing of  $M_1$  and  $M_2$  not represented).

#### 3.3.5 LNA with Resistive Feedback

The LNA in figure 3.7 uses one resistor connected in a shunt-shunt feedback topology [77, 78]. The LNA input impedance, from the incremental model of figure 3.8, is:

$$Z_{in} = \frac{R_F + Z_L}{1 + g_m Z_L + (R_F + Z_L) s C_{gs}}$$
(3.7)



Figure 3.7: LNA with resistive feedback (biasing of  $M_1$  not represented). If the load impedance  $Z_L$  is high enough,  $Z_{in}$  simplifies to:

$$Z_{in} = \frac{1}{g_m + sC_{gs}} \tag{3.8}$$

For frequencies at which  $sC_{gs}$  is negligible in comparison with  $g_m$ , the input impedance is almost real and equal to  $1/g_m$ . Neglecting  $Z_L$  and  $C_{gs}$ , it can also be shown that the voltage gain is

$$A_{v} = \frac{v_{o}}{v_{s}} = \frac{1 - R_{F}g_{m}}{1 + R_{S}g_{m}}$$
(3.9)



Figure 3.8: Incremental model of the LNA with resistive feedback.

If  $g_m$  is high enough, the voltage gain becomes

$$A_v \approx -\frac{R_F}{R_S} \tag{3.10}$$

The noise figure also benefits from the transistor  $g_m$  increase, but, as in the common-gate LNA, the maximization of  $g_m$  in order to improve the LNA voltage gain (3.9) affects the input impedance (3.8).

## 3.3.6 Discussion

Another LNA is the distributed LNA, based on transmission lines, which is out of the scope of this work, that is restricted to lumped element circuits. Typically distributed LNAs have higher power consumption than other circuits [79–82].

In the common-gate LNA and in the LNA using resistive feedback, the precise input impedance value relies on a precise value of  $g_m$ . This is a disadvantage because the gain maximization is not possible. For the remaining topologies (common source LNA with inductive degeneration, LNA with current reuse), the input impedance depends on three different parameters ( $g_m$ ,  $C_{gs}$  and  $L_S$ ), allowing gain optimization simultaneously with input impedance matching. The topology with the best noise performance is the common-source LNA topology with inductive degeneration (the noise performance of this topology is developed ahead in this chapter). This makes this topology the most widely chosen for LNA design . This is analyzed with more detail in the following section.

## 3.4 Cascode LNA with Inductive Degeneration

The common-source LNA using inductive degeneration is further studied in this section, due to its importance in the design of the multi-band LNA proposed in this thesis. This topology is presented in figure 3.4 and is claimed to be the LNA topology with better noise performance [56]. This topology was analyzed in last section considering a simple model, neglecting the gate-drain capacitance  $C_{gd}$  and the gate resistance  $R_g$ . This will be considered in the following analysis.

## **3.4.1** Effect of $C_{gd}$

Throughout the overview presented in section 3.3, the transistor model does not include the gate-drain capacitance  $C_{gd}$ . This capacitance creates a signal path from the output to the input, that degrades the LNA reverse isolation. Several techniques have been proposed to overcome this problem, and three of them are discussed next.

### Transformer Feedback

The LNA proposed in [83] and represented in figure 3.9 uses one transformer to feed back part of the output signal in order to cancel the signal feed back through  $C_{gd}$ .



Figure 3.9: Transformer feedback LNA (biasing of  $M_1$  not represented).

The use of the source inductor  $L_S$  coupled to the output inductor  $L_O$  leads to an input impedance different from that of the common-source LNA studied before [84]. Another trade-off is the transformer design and all concern on its parasitics.

#### **Tuned Inductor Feedback**

The LNA of figure 3.10, also proposed in [83], uses inductor  $L_f$  to resonate with  $C_{gd}$ . This technique requires an extra inductor and a large capacitor that leads to a die area increase. The use of a resonant inductor also affects the LNA resonant frequency.



Figure 3.10: LNA using tuned inductor (biasing of  $M_1$  not represented).

#### Cascode LNA

Another technique consists of using the cascode transistor  $M_2$  of figure 3.11.



Figure 3.11: Cascode LNA (biasing of  $M_1$  not represented).

Besides the advantage of reducing significantly the Miller effect due to  $C_{gd}$  of transistor  $M_1$ , the cascode circuit presents other advantages [85–89]:

- the output-input isolation is improved without significant power consumption increase;
- the LNA output impedance is increased;

• the noise interference of systems ahead of the LNA is reduced.

This LNA circuit requires a higher supply voltage than the last two; however, it has a much lower area. The bias voltage  $V_{BIAS}$  is usually equal to the supply voltage  $V_{DD}$  to prevent the use of another DC voltage.

#### 3.4.2 Input Impedance and Noise Figure

The cascode LNA with inductive degeneration is represented in figure 3.11. The capacitor  $C_x$  is connected in parallel with  $C_{gs}$  of transistor  $M_1$ , and is used to add an extra variable to the circuit design. The incremental model of the cascode configuration formed by  $M_1$  and  $M_2$  is represented in figure 3.12. The input impedance is:

$$Z_{in} = \frac{g_{m1}L_S}{C_t} + \frac{1}{sC_t} + sL_S$$
(3.11)



Figure 3.12: Cascode configuration incremental model.

where  $C_t = C_{gs1} + C_x$ . This amplifier has a real input impedance at its resonant frequency; i.e.,

$$Z_{in} = \frac{g_{m1}L_S}{C_t} \tag{3.12}$$

when

$$\omega_0 = \frac{1}{\sqrt{L_S C_t}} \tag{3.13}$$

Figure 3.13 represents the amplifier incremental model suitable for noise analysis, which includes only the transistor  $M_1$  noise sources. The non-resistive passive elements are considered noiseless and the noise due to  $M_2$  is considered negligible.



Figure 3.13: Incremental model of the degenerated cascode LNA with noise sources.

In the incremental model of figure 3.13, the transistor  $M_1$  output resistance  $r_{o1}$  is considered high enough to be negligible. One way of determining the noise factor consists of transferring all noise sources to the LNA input. Considering the MOS equivalent input noise sources, described in section 2.4, the equivalent noise sources referred to the input, and in accordance with figure 3.14, are:

$$v_{n,i} \approx v_{n,R_g} + \left(A_v Z_S - \frac{1}{G_m}\right) i_{n,d} \tag{3.14}$$

where  $G_m = i_y/v_x$  is the transconductance of the incremental model and  $A_v = v_x/v_y$ is the voltage ratio between points x and y:

$$G_m(s) = \frac{i_y}{v_x} = \left(R_S + R_g + \frac{1}{Y_t} + Z_S + \frac{g_m L_S}{C_t}\right)^{-1} \frac{g_m}{Y_t}$$
(3.15)

$$A_v(s) = \frac{v_x}{v_y} = \frac{(R_S + R_g)Y_t + 1 + (g_{m1} + Y_t)Z_S}{(g_{m1} + Y_t)Z_S}$$
(3.16)

 $Y_t$  is the admittance due to  $C_t$  and  $Z_s$  is the admittance due to  $L_s$ .

The input noise power spectral density, using the Wiener-Khintchine theorem, is:

$$N_i(f) = N_{R_g} + \left| A_v Z_S - \frac{1}{G_m} \right|^2 N_{i_d}$$
(3.17)

and the noise factor, determined at the resonance frequency  $\omega_0$ , assuming a matched input impedance  $R_S = g_m L_S/C_t$ , and  $|G_m|^{-1} \gg A_v Z_S$ , is:

$$F(\omega_0) = 1 + \frac{N_i(\omega_0)}{N_S(\omega_0)} \approx 1 + \frac{R_g}{R_S} + \frac{\gamma g_{d0}(R_S + R_g)^2 \omega_0^2 C_t^2}{R_S g_m^2}$$
(3.18)



Figure 3.14: Incremental model of the degenerated cascode LNA with an input equivalent voltage noise source.

If  $R_S \gg R_g$ ,  $F \approx 1 + \gamma g_{d0} \frac{\omega_0^2 C_t^2 R_S}{g_m^2}$  (3.19)

From (3.19), it is concluded that the noise factor improves (decreases) for technologies with higher transition frequencies  $f_t$ . The gate resistance  $R_g$  can be minimized by proper design, for example by multi-fingering of the transistor gate [2]. This topology has been widely studied [2, 3, 7, 8, 45, 56, 85, 86, 90–99].

## 3.5 Double Loop Feedback LNAs

### 3.5.1 Four Feedback Topologies

In section 3.3, some known LNA topologies are discussed and most of them use feedback in an implicit way. Feedback has several advantages like: better noise performance, gain insensitivity against parameter changes, precise control of input and output impedances, increased bandwidth and reduced non-linear distortion [43, 89, 100–102].

A feedback amplifier has an amplifying block A, which provides gain, and a feedback network  $\beta$ , which samples the output signal and feeds the result to the amplifying block input as shown in figure 3.15 [102–104]. If the feedback results in an output signal increase, it is called positive feedback (used in oscillator design), otherwise it is negative feedback (used in amplifier design) [43].



Figure 3.15: Block diagram of a feedback system.

The gain is

$$A_F = \frac{x_o}{x_i} = \frac{A}{1 + A\beta} \tag{3.20}$$

where  $A\beta$  is the loop gain and  $\beta$  is the feedback factor. If  $A\beta \gg 1$ ,  $A_F$  depends only on the feedback network:

$$\lim_{A \to \infty} A_F = \frac{1}{\beta} \tag{3.21}$$

In this situation,  $A_F$  is insensitive to the amplifying block variations, due to temperature, age or bias conditions. This leads to a common design methodology, that consists of designing the feedback network assuming that the amplifying block is designed to have a high gain. An ideal amplifying block can be represented by a nullor, which is a 2-port with infinite gain (voltage gain, current gain, transconductance

and transimpedance [41, 101]). The nullor chain matrix is:

$$\begin{bmatrix} v_1 \\ i_1 \end{bmatrix} = \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} \begin{bmatrix} v_2 \\ -i_2 \end{bmatrix}$$
(3.22)

Four different feedback circuit configurations can be considered: series-series (figure 3.16), series-shunt (figure 3.17), shunt-series (figure 3.18) and shunt-shunt (figure 3.19).



 $i_{1}$ +
2-port
+ A  $v_{1}$ - $\beta$ 

Figure 3.16: Series-series feedback.



Figure 3.18: Shunt-series feedback.

Figure 3.17: Series-shunt feedback.



Figure 3.19: Shunt-shunt feedback.

For each feedback configuration, there is a matrix description that is the most suitable for describing the complete feedback system, and that consists on simply adding both 2-port matrices, as indicated in table 3.1 [43, 105]. The different matrices are defined in appendix A.

-			
Feedback	Input	Output	Most
configuration	comparison	sampling	suitable
	quantity	quantity	2-port matrix
series-series	voltage	current	Z
series-shunt	voltage	voltage	hybrid $\mathbf{H}$
shunt-series	current	current	hybrid <b>H</b> '
shunt-shunt	current	voltage	Y

Table 3.1: Four feedback topologies.

## 3.5.2 Feedback Low Noise Amplifiers

During the LNA design, it is convenient to start by replacing the amplifying block by a nullor. Thus, it is possible to evaluate the performance limits of the feedback amplifier due only to the feedback network non-idealities. This allows hierarchization of the design process, by first designing the feedback network and then designing an amplifying block that approximates the nullor performance.

The passive elements used in the feedback block can be two terminal (resistors, capacitors, and inductors) or four terminal (transformers). Using one two-terminal element, two feedback configurations are possible: series-series and shunt-shunt feedback. Using transformers, the other two feedback configurations are also possible: series-shunt and shunt-series feedback. It is not recommended to use transformers in the series-series and shunt-shunt feedback configurations, because the transformer current-voltage ratio is not linear. In figures 3.20 to 3.23 the four different feedback configurations using a single element are represented.



Figure 3.20: Series-series feedback with two-terminal element.



Figure 3.21: Shunt-shunt feedback with two-terminal element.



Figure 3.22: Series-shunt feedback with a transformer.

Figure 3.23: Shunt-series feedback with a transformer.

 $v_2$ 

Consider now the amplifying block with finite input impedance  $Z_i$ , represented in figure 3.24.



Figure 3.24: Amplifying block having finite input impedance.

The feedback amplifier input impedance is  $Z_i(1+A\beta)$  for the input series connection (figures 3.20 and 3.22) and is  $Z_i/(1+A\beta)$  for the input parallel connection (figures 3.21 and 3.23) [89]. If the amplifying block gain is infinite, it is not possible to obtain finite input impedances using a single loop feedback [23]. To obtain a finite and precise input impedance using a single loop, the amplifying block gain must be finite. This is what happens in the common-source LNA using either inductive degeneration or resistive feedback (discussed in section 3.3).

Since it is not possible to design an LNA using a single loop feedback if the amplifying block gain is infinite, double loop feedback is evaluated in the following.

## 3.5.3 Double Loop Feedback (DLF)

Consider the case where both feedback loops compare the same variable. In figure 3.25 a) current is compared and in figure 3.25 b) voltage is compared. If the amplifying block A is a nullor, in the first case  $v_i = 0$ , so the input impedance  $(v_i/i_i)$  is zero. In the second case  $i_i = 0$ ; so the input impedance is infinite. Thus, it is not possible to obtain a finite input impedance using two feedback loops comparing the

same variable at the input.



Figure 3.25: Two feedback loops, both comparing (a) current; or (b) voltage.

Consider now the comparison of different variables. First consider only DLF amplifiers having different variables sampled at the output. The different possibilities are represented in figure 3.26.



Figure 3.26: Double loop feedback amplifier sampling different variables and comparing different variables.

Topology in figure 3.26	Feedback loops	Input impedance
(1)	series-series inside shunt-shunt	$\frac{\beta_1}{\beta_2} \frac{i_o}{v_o}$
(2)	shunt-series inside series-shunt	$\frac{\beta_2}{\beta_1} \frac{i_o}{v_o}$
(3)	series-shunt inside shunt-series	$\frac{\beta_1}{\beta_2} \frac{v_o}{i_o}$
(4)	shunt-shunt inside series-series	$\frac{\beta_2}{\beta_1} \frac{v_o}{i_o}$

Table 3.2: Input impedance for the DLF amplifier combinations of figure 3.26.

The input impedance of the different DLF amplifiers obtained is listed in table 3.2. In all cases the input impedance is not independent of the output load  $(v_o/i_o)$ .

Consider now DLF amplifiers sampling the same variable and comparing different variables. In figure 3.27 the different amplifier combinations are presented.

The input impedance of the DLF amplifier combinations of figure 3.27 is listed in table 3.3.

It is concluded that it is now possible to have a finite input impedance, independent of the output load. Thus, the feedback amplifier must have the same quantity sampled at the output and different quantities at the input [23, 106]. The only DLF amplifiers that are suitable for LNA design are those having the following feedback loops:

- shunt-series and series-series;
- shunt-shunt and series-shunt.

These DLF amplifiers will be studied considering an ideal amplifying block, but replacing the feedback loops by circuit elements.



Figure 3.27: Double loop feedback amplifier sampling the same variable and comparing different variables.

Topology of figure 3.27	Feedback loops	Input impedance
(1)	series-shunt inside shunt-shunt	$\frac{\beta_1}{\beta_2}$
(2)	shunt-series inside shunt-shunt	$\frac{\beta_2}{\beta_1}$
(3)	series-series inside shunt-series	$\frac{\beta_1}{\beta_2}$
(4)	shunt-series inside series-series	$\frac{\beta_2}{\beta_1}$

Table 3.3: Input impedance for the DLF amplifier combinations of figure 3.27.

=

## 3.5.4 DLF Amplifiers suitable for LNA Design

Using the feedback types of figures 3.20 to 3.23, the four different DLF amplifiers suitable to obtain finite input impedances independent on the output load are represented in figures 3.28 to 3.31 using lumped elements.



Figure 3.28: Amplifier having series-series Figure 3.29: Amplifier having series-shunt feedback inside shunt-series feedback. feedback inside shunt-shunt feedback.



Figure 3.30: Amplifier having shunt-series Figure 3.31: Amplifier having shunt-shunt feedback inside series-series feedback. feedback inside series-shunt feedback.

These DLF LNAs using lumped elements are now analyzed in terms of input impedance, gain and noise factor. For convenience, the order by which these topologies are analyzed does not respect the order of figures 3.28 to 3.31.

## 3.5.5 DLF LNA Topologies

The first two DLF LNA topologies that will be analyzed sample the output voltage with both feedback blocks. In this analysis, the transformer is ideal, and the amplifying block is an unilateral and ideal voltage amplifier with a voltage gain of A, an infinite input impedance and a zero output impedance. All LNAs are

analyzed in terms of input impedance, gain and noise factor and only the resistive element of the feedback network is noisy.

# DLF LNA type 1 - DLF LNA having shunt-shunt feedback inside series-shunt feedback

The DLF LNA type 1 is represented in figure 3.32.



Figure 3.32: DLF LNA type 1.

The input impedance of the DLF LNA of figure 3.32 is determined considering the following equations:

$$\begin{cases} v_i = -\frac{v_n}{n} - \frac{v_o}{A} \\ i_i = (v_i - v_o) \frac{1}{Z_f} \end{cases}$$
(3.23)

where A is the voltage gain of the amplifying block. The input impedance  $Z_{in}$  and the voltage gain  $v_o/v_i$  are:

$$Z_{in} = \frac{v_i}{i_i} = \frac{A+n}{A(n+1)+n} Z_f$$
(3.24)

$$A_v = \frac{v_o}{v_i} = -n\frac{A}{A-n} \tag{3.25}$$

If the amplifying block is ideal (infinite voltage gain A), the input impedance and the voltage gain become:

$$\lim_{A \to \infty} Z_{in} = \frac{Z_f}{n+1} \tag{3.26}$$

$$\lim_{A \to \infty} A_v = -n \tag{3.27}$$

Observing the input impedance (3.26) and knowing that the transformer ratio n is real and positive, it is concluded that the only way to obtain a positive and finite real input impedance using an ideal amplifying block is by having a real feedback impedance  $Z_f$ . From now on,  $Z_f$  will be replaced by  $R_f$  in this topology.

Concerning the noise performance, it is assumed that  $R_f$  is the only noise source of the circuit, with noise power spectral density:

$$N_{R_f}(f) = 4k_B T R_f \tag{3.28}$$

In figure 3.33 is represented the DLF LNA with the voltage noise source due to the thermal noise generated in  $R_f$ .



Figure 3.33: DLF LNA type 1 and the  $R_f$  noise source.

Using the source transformations of appendix C, the noise voltage source is first converted into a current source  $i_{n,Rf} = v_{n,Rf}/R_f$  - step (1). By its turn, this current source is split into two sources, one connected to the LNA input and one connected to the LNA output - step (2). The current source connected to the output is in parallel with the (zero) output impedance of the amplifying block and thus its effect is canceled. These transformations are represented in figure 3.34.

The equivalent input noise source due to the feedback resistance corresponds to a current source  $i_{n,i}$  in accordance with figure 3.35.



Figure 3.34: Noise transformations to determine the equivalent LNA input noise source.



Figure 3.35: DLF LNA type 1 with equivalent input noise source.

 $i_{n,i}$  is

$$i_{n,i} = \frac{v_{n,R_f}}{R_f} \tag{3.29}$$

This equivalent noise current can be converted to a noise voltage by multiplying it by the input resistance  $R_s$ :

$$v_{n,i} = i_{n,i}R_S = v_{n,R_f}\frac{R_S}{R_f}$$
(3.30)

If a matched condition is considered,  $R_f/(n+1) = R_S$ , the noise power spectral density due to the feedback resistor, and reported to the DLF LNA input, becomes:

$$N_i = 4k_B T R_f \left| \frac{R_S}{R_f} \right|^2 = \frac{4k_B T R_S}{n+1}$$
(3.31)

and the minimum achievable noise factor becomes:

$$F_{\min} = \frac{N_{R_S} + N_i}{N_{R_S}} = 1 + \frac{1}{n+1}$$
(3.32)

If n is higher than 1, F is much less than 2, which is the value that would be obtained if the LNA used a single resistor in parallel with its input to realize the impedance matching. If n is very high, F tends to its absolute minimum of 1, meaning that the LNA could be considered (approximately) noiseless.

# DLF LNA type 2 - DLF LNA having shunt-shunt feedback inside series-shunt feedback

The DLF LNA type 2 is represented in figure 3.36.



Figure 3.36: DLF LNA type 2.

Concerning the DLF LNA of figure 3.36, its equations are,

$$\begin{cases} v_i = -\frac{v_o}{n} - \frac{v_o}{A} \\ i_i = \left(v_i + \frac{v_o}{n} - v_o\right) \frac{1}{Z_f} \end{cases}$$
(3.33)

The LNA input impedance is

$$Z_{\rm in} = \frac{v_i}{i_i} = Z_f \frac{A+n}{An+n} \tag{3.34}$$

Considering an ideal amplifying block with infinite gain, the input impedance becomes:

$$Z_{in} = \frac{Z_f}{n} \tag{3.35}$$

Again in this topology  $Z_f$  has to be real to obtain a real input impedance. The

voltage gain is the same as for the previous circuit.

It can be shown that the noise factor is [24]:

$$F_{\min} = 1 + \frac{1}{n}$$
 (3.36)

Comparing the two DLF LNAs analyzed before, one sees that the only difference between them is the input impedance denominator, which is n + 1 in the first DLF LNA and is just n in the second. This is a minor difference if  $n \gg 1$ .

The following two DLF LNA topologies that will be analyzed sample the output current with both feedback blocks. The transformer is again ideal and the amplifying block is, in this case, an unilateral and ideal current amplifier with infinite current gain, input and output impedances. The input impedance, the current gain and the noise factor are determined and only the resistive element of the feedback network is considered noisy.

# DLF LNA type 3 - DLF LNA having series-series feedback inside shunt-series feedback

The DLF LNA type 3 is represented in figure 3.37.



Figure 3.37: DLF LNA type 3.

In the DLF LNA with current output the parameter that should be maximized is the transconductance instead of the voltage gain. Assuming an ideal amplifying block,

$$\begin{cases} v_i = Z_f i_o \\ i_i = \frac{i_o}{n} \end{cases}$$
(3.37)

The input impedance and transconductance are:

$$Z_{in} = nZ_f \tag{3.38}$$

$$\frac{i_o}{v_i} = \frac{1}{Z_f} \tag{3.39}$$

Again in this case, the feedback impedance  $Z_f$  has to be real. Since  $R_f$  is multiplied by the transformer ratio n,  $R_f$  is lower than 50  $\Omega$ , assuming that n > 1.

Concerning the noise factor, it can be shown that [24]:

$$F_{\min} = 1 + \frac{1}{n}$$
 (3.40)

# DLF LNA type 4 - DLF LNA having shunt-series feedback inside series-series feedback

The DLF LNA type 4 is represented in figure 3.38.



Figure 3.38: DLF LNA type 4.

Assuming an ideal amplifying block,

$$\begin{cases} v_i = Z_f(i_i + i_o) \\ i_i = \frac{i_o}{n} \end{cases}$$
(3.41)

The input impedance and transconductance are

$$Z_{in} = (n+1)Z_f (3.42)$$

$$\frac{i_o}{v_i} = \frac{n}{n+1} \frac{1}{Z_f}$$
 (3.43)

As an image of the other DLF LNA topologies, the feedback impedance  $Z_f$  must be real to obtain a real input impedance. The input impedance and the current gain are close to that determined for last topology. It can be shown that the noise factor is

$$F_{\min} = 1 + \frac{1}{n+1} \tag{3.44}$$

Assuming that n is equal in the two last topologies, this topology achieves a slightly lower noise factor than the other one.

#### Discussion:

Using two feedback loops, it is possible to obtain four different LNA topologies, two of them sampling the output voltage and the others sampling the output current. These topologies are suitable for LNA design because it is possible to have gain with a real input impedance and noise performance is acceptable.

In chapter 6 the DLF LNA of figure 3.32, which samples the output voltage is analyzed in more detail, considering an improved feedback network model. The implementation of the amplifying block will also be developed.

# 3.6 Conclusions

The most important LNA parameters are input impedance, gain, and noise factor and the LNA design results from the tradeoff between these parameters. There are several LNA topologies that have these three parameters simultaneously, and among them the one that achieves the best tradeoff is the cascode LNA using inductive degeneration. Theoretically, this topology achieves a real input impedance and a high gain with the lowest noise figure.

If the LNA is realized as a feedback amplifier having a single feedback loop, it is shown that a finite input impedance is achieved only if the amplifying block has a finite gain.

Another form of obtaining a finite input impedance, with gain and low noise figure consists of using double loop feedback. Only the DLF amplifiers sampling the same variable at the output and comparing different variables at the input have a finite input impedance independent from the load impedance. It is shown that they have additionally gain and low noise factor.

With lumped elements, four different DLF LNA topologies can be designed: two sampling current and two sampling voltage.

# Chapter 4

# Multi-band LNA using Cascode Transistors for Band Selection

4.1	Introduction	75
4.2	A New Multi-Band Low Noise Amplifier	76
4.3	Simulation Results	82
4.4	Conclusions	86

# 4.1 Introduction

Conventional receiver architectures use narrowband low noise amplifiers (LNAs), with impedance matching to the antenna for maximum power transfer, and which isolate the antenna from the rest of the receiver and amplify the input signal with minimum addition of noise. However, narrowband LNAs are limited to a single frequency band. The demand for more functionality, like the reception with different wireless standards, leads to the research of new multi-band LNA topologies. This chapter is dedicated to the study of multi-band LNAs.

A multi-band CMOS LNA, which receives two or more frequency bands simultaneously, and separates them into different outputs using replicated cascode transistors, is presented in this chapter. This LNA does not have switches in the signal path. Either only one band or two or more bands simultaneously can be selected. This LNA is suitable for receivers using multi-band antennas, and where each band has to be processed by a dedicated circuit; i.e., the circuits ahead of the multi-band LNA are narrowband.

In section 4.2, the multi-band LNA having independent outputs activated by cascode transistors, is presented and analyzed. In section 4.3, simulation results are presented to demonstrate the circuit performance. Finally, in section 4.4 some conclusions are drawn.

# 4.2 A New Multi-Band Low Noise Amplifier

In this section, some topics on multi-band LNAs are discussed, and a multi-band LNA having independent outputs activated by cascode transistors is presented and analyzed.

## 4.2.1 Topics on Multi-band Low Noise Amplifiers

Figure 4.1 presents two forms of obtaining multi-band receivers (from the LNA point of view). In figure 4.1 a) there is a complete duplication of the narrowband receiving circuits [107], while in figure 4.1 b) a single multi-band LNA with two different narrow frequency bands splits the received dual-band signal through different paths [108]. The last solution has the advantage of using a single antenna and may also lead to die area and power savings. Other forms of designing multi-band receivers and multi-band LNAs are developed in the next chapter.



Figure 4.1: Two multi-band receiver architectures.

In this thesis, the multi-band LNAs of interest can operate at different frequencies, without a large increase in power or area, compared with a narrowband LNA [6]. A multi-band LNA is not a collection of several independent narrowband LNAs, so the solution of figure 4.1 a) is ruled out, and the approach of figure 4.1 b) will be the object of study. Most multi-band CMOS LNAs are based on the cascode topology with inductive degeneration (figure 4.2), which was studied in section 3.4.

The inductive degeneration provides a real input impedance and signal gain at a specific frequency band, without significant noise addition. Cascode transistor  $M_2$  is used in this circuit to reduce the Miller effect due to the gate-drain capacitance of  $M_1$  ( $C_{gd1}$ ). When the effect of  $C_{gd1}$  is neglected, the input impedance can be written as:

$$Z_{in} = \frac{g_{m1}L_S}{C_t} + \frac{1}{sC_t} + sL_S$$
(4.1)



Figure 4.2: Cascode LNA with inductive degeneration (biasing of  $M_1$  not represented).

where  $C_t = C_{gs1} + C_x$ . When  $C_t$  and  $L_S$  resonate, the input impedance is real. Other advantages of using  $M_2$  are improved reverse isolation between input and output, high output resistance, and better noise performance [86].  $M_1$  converts the input voltage into a current, that is converted back to an amplified voltage by the output network, formed by inductor  $L_O$  and capacitor  $C_O$ . This output network is tuned to a frequency equal to:

$$\omega_0 = \frac{1}{L_O C_O} \tag{4.2}$$

The input and output networks should be tuned to the same frequency.

This LNA topology is widely studied in literature, and several studies on how to optimize different LNA parameters like noise performance, linearity or voltage supply are found in [83, 90, 92, 99, 109, 110].

### 4.2.2 A New Multi-Band LNA

The LNA in figure 4.3 is based on the cascode LNA topology with inductive degeneration. It has wide-band input matching obtained with a resonant filter ( $L_1$  and  $C_1$ ) added to the LNA of figure 4.2. At the output, this LNA has two independent resonant circuits tuned to distinct frequencies and connected to  $M_1$  through different cascode transistors. Each cascode transistor can be used to switch off the corresponding output branch by lowing the gate voltage (which is usually constant and typically equal to the supply voltage in the conventional circuit of figure 4.2).



Figure 4.3: Dual-band LNA based on cascode switching ( $M_1$  bias not included).

The circuit of figure 4.3 is represented with two output networks, but it can have more output branches, each one tuned to a different frequency and selected by the corresponding cascode transistor.

When several input signal bands are present simultaneously, each output network is only sensitive to the frequency band for which it is tuned. When only one cascode transistor is on, the LNA works as a normal single-band LNA and, apart from parasitic coupling effects, the inactive outputs will not affect the active one.

The input impedance matching network is built with inductor  $L_1$  and capacitor  $C_1$  providing wideband impedance matching that embraces the multiple tuning frequencies [108, 111]. It has lower and upper 3 dB cutoff frequencies,  $\omega_{lower}$  and  $\omega_{upper}$ , determined by:

$$\begin{cases} \omega_{lower} = \frac{R_{in}}{L_1} \\ \omega_{upper} = \frac{1}{R_{in}C_1} \end{cases}$$
(4.3)

where  $R_{in}$  is given by

$$R_{in} = \frac{g_{m1}L_S}{C_t} \tag{4.4}$$
The LNA of figure 4.3, having only one input, can be connected to a multi-band antenna. This LNA does not have switches in the signal path, because each output is activated by the corresponding cascode transistor.

There is also the possibility of controlling the gain of one output simply by varying the cascode transistor gate voltage (and consequently varying its quiescent current).

#### 4.2.3 Noise Analysis

In chapter 3, the noise factor of the cascode LNA using inductive degeneration was obtained considering a single-band matching network. In this chapter the input matching network is wideband and is determined by  $L_1$  and  $C_1$ . Figure 4.4 represents the incremental model of the input part of the multi-band LNA of figure 4.3. The noise sources considered are the thermal noise voltage due to the gate resistance  $v_{n,Rg}$  and the noise current source due to the channel resistance  $i_{n,d1}$ .



Figure 4.4: Incremental model of the input part of the multi-band LNA with noise sources.

In figure 4.5, noise source transformations (appendix D) are used to determine the noise voltage  $v_{n,i}$ , referred to the LNA input.

In step 1 the current source  $i_{n,d1}$  is split into two current sources. In step 2, one of the current sources is converted into a voltage source  $Z_S i_{n,d1}$ , where  $Z_S$  is the impedance due to  $L_S$ .  $Z_S i_{n,d1}$  is referred to the input, by dividing it by the voltage gain  $A_v = v_z/v_x$ , in step 3. In step 4, the other current source  $i_{n,d1}$  is referred to the input by using the LNA transconductance  $G_m = i_y/v_x$ . Finally, in step 5, the voltage source  $v_{n,Rg}$  is translated to the input by dividing it by the voltage gain  $A'_v = v_w/v_x$ .

# CHAPTER 4. MULTI-BAND LNA USING CASCODE TRANSISTORS FOR BAND SELECTION



Figure 4.5: Incremental model of the multi-band LNA showing noise source transformations.

 $v_{n,i}$  is:

$$v_{n,i} = \left(A_v^{\prime-1}\right)v_{n,R_g} + \left(A_v^{-1}Z_S - \frac{1}{G_m}\right)i_{n,d1}$$
(4.5)

where  $A'_v$ ,  $A_v$  and  $G_m$  are:

$$A'_{v} = \frac{v_{w}}{v_{x}} = \frac{Z_{a} \| Z_{in}}{R_{S} + Z_{a} \| Z_{in}}$$
(4.6)

$$A_{v} = \frac{v_{z}}{v_{x}} = \frac{v_{w}}{v_{x}}\frac{v_{z}}{v_{w}} = A_{v}'\left(\frac{(g_{m1} + Y_{t})Z_{S}}{R_{g}Y_{t} + 1 + (g_{m1} + Y_{t})Z_{S}}\right)$$
(4.7)

$$G_m = \frac{i_y}{v_x} = \frac{v_w}{v_x}\frac{i_y}{v_w} = A'_v \frac{g_{m1}}{1 + R_g Y_t + g_{m1} Z_S + Y_t Z_S}$$
(4.8)

where  $Y_t = Y_{gs} + Y_x$  is the admittance due to  $C_{gs}$  and  $C_x$ .  $Z_{in}$  is

$$Z_{in} = R_g + \frac{1}{Y_t} + Z_S + \frac{g_{m1}L_S}{C_t}$$
(4.9)

and  $Z_a$  is

$$Z_a = \frac{sL_1}{s^2 L_1 C_1 + 1} \tag{4.10}$$

The noise power spectral density is determined using (4.5) and the Wiener-Khintchine theorem:

$$N_i(f) = \left| A_v^{\prime - 1} \right|^2 N_{R_g}(f) + \left| A_v^{-1} Z_S - \frac{1}{G_m} \right|^2 N_d(f)$$
(4.11)

The noise factor is finally:

$$F(f) = 1 + \frac{N_i(f)}{N_S}$$
(4.12)

Analyzing (4.11) and (4.12), it is visible that the noise performance improves if the LNA transconductance is high. The degeneration increases the noise factor; and thus,  $L_S$  should be low. These two specifications are in accordance with what is found in literature concerning the cascode LNA with inductive degeneration [56]. However, it is expected a different frequency response to that verified in the narrowband cascode LNA due to the presence of  $Z_a$ . The presence of several cascode transistors (one for each output network) could mean an increase in the noise figure; however, the current that bias the gain transistor ( $M_1$  of figure 4.3) is divided by each cascode transistors reducing their transconductance, and consequently the thermal noise generated. Thus, it is expected a small difference between the different operating modes.

## 4.3 Simulation Results

To evaluate the feasibility of the dual-band LNA of figure 4.3, a circuit was designed using AMS 0.35  $\mu$ m CMOS technology with 3 V supply. The frequencies of 900 MHz and 1.8 GHz are chosen because they are two widely used frequencies. The cascode transistors have the same dimensions, so their bias currents are the same when both outputs are active. All transistors are designed with the minimum length to improve the transition frequency. The output inductor series resistance affects the voltage gain characteristic. The output network inductors  $L_{11}$  and  $L_{22}$  and their series resistances correspond to integrated spirals provided by the AMS technology. Inductors  $L_S$  and  $L_1$  of the input network are assumed to be ideal. The values used in the simulations are listed in table 4.1.

Parameter	Value	Parameter	Value	
$W_{M1}$	$800 \ \mu m$	$L_S$	1.2 nH	
$W_{M_{c1}}$	$36~\mu{ m m}$	$L_{11}$	13.2  nH	$(r_{11} = 14 \ \Omega)$
$W_{M_{c2}}$	$36~\mu{ m m}$	$C_{11}$	2.2  pF	
$L_1$	9.0  nH	$L_{22}$	6.4  nH	$(r_{22} = 7 \ \Omega)$
$C_1$	$700~\mathrm{fF}$	$C_{22}$	1.1  pF	
$C_x$	1.2  pF	$g_{m1}$	$101 \mathrm{mS}$	
$C_{gs1}$	743 fF	$R_{g1}$	$4.5 \ \Omega$	

Table 4.1: Element values used in simulations and theoretical curves.

Figure 4.6 shows the voltage gain at the two outputs, when only one output is active, i.e., when  $M_{c1}$  works in saturation and  $M_{c2}$  is cutoff, or vice-versa. The voltage gain has a peak at approximately the desired frequency, determined by the resonance of  $L_{ii}$  and  $C_{ii}$ . There is no significant influence from the inactive output. Figure 4.7 presents the same curves of figure 4.6 together with the gain at the inactive outputs. If a cascode transistor is cutoff, the voltage gain at the corresponding output is below -70 dB for all the frequency range, which is low enough to consider that output inactive.

Figure 4.8 shows the curves of figure 4.6 compared with those obtained when both outputs are active: the gain at each output drops approximately 6 dB in comparison with the case where only one output is active (the gain reduces to a half). Since both cascode transistors are equal, the current from  $M_1$  is divided equally by the two output branches.

Figure 4.9 represents the voltage gain at output 2, when  $V_{BIAS1}$  of  $M_{c1}$  is varied.



Figure 4.6: Voltage gain at "active" outputs 1 and 2 when one cascode transistor is in saturation and the other is cutoff.



Figure 4.7: Comparison of voltage gain at active and inactive outputs.

This shows that it is possible to vary significantly the gain.

The input matching, measured by parameter  $S_{11}$ , is represented in figure 4.10 in three different operating modes (both outputs active, or only one). In all three cases  $S_{11}$  is close to -10 dB at the two frequencies of interest, without significant differences between them. This means that the input impedance is not significantly



Figure 4.8: Voltage gain at outputs 1 and 2, with both outputs active and with only one output active.



Figure 4.9: Voltage gain at output 2 when  $V_{BIAS2}$  varies and  $V_{BIAS1} = V_{DD}$ .

affected by the different operating modes.

In figure 4.11, the LNA noise figure obtained by simulation is compared with the theoretical value. The values used to determine the NF curves are listed in table 4.1. The noise figure at 900 MHz and 1.8 GHz is, respectively, 1.5 dB and 2.5 dB, which is well below the 3 dB reference value, giving a margin for other noise sources



Figure 4.10:  $S_{11}$  plots in three different situations, showing the low influence of the different operating modes.

to be considered. The small difference between the theoretical and simulated curves is due to the high transconductance of  $M_1$  that attenuates significantly the influence of the noise sources ahead of  $M_1$ , like the noise sources originated at the cascode transistors. There is no significant difference between the different modes of operation. This is due to the much higher transconductance of  $M_1$  in comparison with the transconductance of the cascode transistors.



Figure 4.11: Simulated noise figure of the complete dual-band LNA circuit.

## 4.4 Conclusions

In this chapter, a new multi-band LNA is presented. This circuit has two (or more) resonant output networks tuned to different frequencies which are switched on and off by cascode transistors. The outputs are activated when the cascode transistors are in saturation. The input impedance is wideband to embrace all working bands and is not significantly affected by the number of active outputs. This is possible because the cascode transistors provide a good output-input isolation and because the current through the input transistor ( $M_1$  in figure 4.3) is kept constant.

When implementing a dual-band CMOS LNA, the voltage gain at one output has a decrease of approximately 3 dB when both output networks are active with respect to the case where only one output is active. It is possible to obtain a noise figure below 3 dB between the central frequencies of the two bands, and the different modes of operation do not affect significantly the noise figure, meaning that this dual-band LNA keeps the good noise performance of the cascode LNA with inductive degeneration which inspired this dual-band LNA.

# Chapter 5

# Concurrent Dual-Band LNA using Magnetically Coupled Inductors

5.1	Introduction	89
5.2	Concurrent Multi-Band LNAs	90
5.3	Concurrent Dual-Band CMOS LNA using Magnetically Coupled Inductors	92
5.4	Effect of the Inductors Parasitic Resistance	96
5.5	Simulation Results	98
5.6	Effect of Reversing the Orientation of the Magnetic Coupling	104
5.7	Conclusions	107

#### 5.1 Introduction

In this chapter, concurrent multi-band LNAs are studied. Concurrent multi-band LNAs work simultaneously at different frequency bands (non-concurrent multi-band LNAs operate at different frequency bands, but one at a time) [6].

A new concurrent dual-band LNA, based on the multi-band LNA studied in last chapter, is proposed in this chapter. Both circuits have the same wideband input impedance matching network; however, this new topology has the inductors of the output networks magnetically coupled. By magnetically coupling the inductors, this LNA will have two bands simultaneously at the output. By balancing the current between both output networks, it is also possible to vary the voltage gain at both bands.

In section 5.2 a short overview of concurrent multi-band LNAs is presented. In section 5.3 a new concurrent dual-band LNA is presented and, its frequency response is analyzed considering ideal elements. The frequency response is reanalyzed, in section 5.4, accounting for the parasitic series resistance of some inductors. In section 5.5 the LNA dimensioning is described and some simulation results are presented. In section 5.6 an alternative LNA circuit is proposed. It has a different voltage gain variation, in comparison with the first circuit presented. Finally, some conclusions are drawn in section 5.7.

## 5.2 Concurrent Multi-Band LNAs

Figure 5.1 presents two forms of obtaining multi-band receiver architectures, from the LNA point of view. In figure 5.1 a) the LNAs are narrowband, have different inputs and are connected through switches to the subsequent blocks [20]. In this architecture two narrowband LNAs are needed; thus, this is not an optimal solution, due to the duplication of blocks (LNAs, and(or) antennas), which leads to an increase in power and die area. Figure 5.1 b) consists of a single-input, single-output, multi-band LNA that has a frequency response with different pass-bands [6]. One advantage of this last solution is the use of a single antenna and only one LNA circuit which may have lower area and power than two narrowband LNAs.



Figure 5.1: Two multi-band receiver architectures.

As referred in last chapter, most multi-band LNAs (including concurrent LNAs) are based on the cascode topology with inductive degeneration of figure 4.2.

The LNA of figure 5.1 b) can be concurrent or non-concurrent, depending on the use of both bands simultaneously or not, respectively. One technique to obtain a nonconcurrent multi-band LNA using the cascode topology, consists of varying the LNA tuning frequency by changing the value of passive elements using transistor switches, as shown in figures 5.2 a) and b) [112, 113]. The (transistor) switches sizing has to trade between a small resistance and a large capacitance. The transistor equivalent resistance has to be lower than the series parasitic resistance of the inductance in parallel, to ensure the DC current will flow through the transistor when it is active. However to obtain a small resistance, the transistor has to be large, which increases the transistor parasitic capacitances, affecting the resonant circuit tuning frequencies [114, 115].

One technique to design a concurrent multi-band LNA, consists of tuning the input and output networks to different frequencies by using multi-resonant circuits, as happens in LNAs of figure 5.3 [6, 116, 117]. However, one problem of this solution



Figure 5.2: a) Variable capacitor. b) Variable inductor.

is a high LNA die area, mainly due to the inductors, that must have enough space between them to ensure a negligible magnetic coupling. Other works concerning the LNA bandwidth extension, mainly that of the cascode LNA can be found in literature [6, 108, 112, 113, 115, 118, 119].



Figure 5.3: Dual-band LNAs using multi-resonant circuits. (a) LNA based on cascode topology with inductive degeneration. (b) LNA based on the common-gate stage ( $C_A$  and  $C_B$  realize a series-shunt feedback). Biasing of  $M_1$  not presented.

In next section a new concurrent dual-band LNA is presented and its frequency response analyzed.

## 5.3 Concurrent Dual-Band CMOS LNA using Magnetically Coupled Inductors

The proposed concurrent dual-band CMOS LNA is presented in figure 5.4. It is similar to that of figure 4.3; however, magnetic coupling between the output inductors changes significantly the overall characteristics.



Figure 5.4: Concurrent dual-band LNA based on inductor magnetic coupling.

#### 5.3.1 Working Principle

The LNA of figure 5.4 has a wideband input stage built with transistor  $M_1$ ,  $L_S$ ,  $L_1$ , and  $C_1$ , which has already been studied in chapter 4. The two output resonant networks, with  $L_{ii}$  and  $C_{ii}$ , are magnetically coupled, with a magnetic coupling coefficient k. The magnetic coupling leads to the presence of two bands at the output, making this LNA a concurrent dual-band LNA. The relative gain of the two bands can be varied simply by changing the cascode transistor gate bias voltage.

The two magnetically coupled inductors together with the capacitors form a double tuned filter, with two different resonant frequencies, that was often used in past in tuned amplifiers [37, 120, 121]. This double tuned filter in the output of LNA of figure 5.4 gives it its dual-band feature; however, using it alone, it is not possible to have relative gain variation. This is achieved by the additional cascode transistor  $M_{c2}$ . By varying the gate voltage of  $M_{c2}$ , while keeping the gate voltage of  $M_{c1}$ constant, it is possible to change the current distribution between the two output branches. Another advantage of having magnetically coupled inductors, is that when they are integrated, they use less area than uncoupled inductors, since the inductors can be overlapped.

#### 5.3.2 Frequency Response

In figure 5.5 the schematic of the double tuned circuit with coupled inductors is represented. Considering both cascode transistors in saturation, their currents,  $i_1$ and  $i_2$  are proportional,  $i_2 = \alpha i_1$ , according to the ratios of their W/L and of their  $V_{GS}$  bias voltages.



Figure 5.5: Double tuned circuit with coupled inductors.

The equations for the circuit in figure 5.5 are:

$$V_1 = sL_{11}I_{l1} + sMI_{l2} (5.1a)$$

$$V_2 = sMI_{l1} + sL_{22}I_{l2} \tag{5.1b}$$

$$I_{l1} = I_1 - V_1 s C_{11} \tag{5.1c}$$

$$I_{l2} = I_2 - V_2 s C_{22} \tag{5.1d}$$

$$I_2 = \alpha I_1 \tag{5.1e}$$

$$M = k\sqrt{L_{11}L_{22}}$$
(5.1f)

The load impedance  $Z_{load} = V_1/I_1$ , as represented in figure 5.5, is:

$$Z_{load} = \frac{C_{22}(L_{11}L_{22} - M^2)s^3 + (L_{11} + \alpha M)s}{C_{11}C_{22}(L_{11}L_{22} - M^2)s^4 + (C_{11}L_{11} + C_{22}L_{22})s^2 + 1}$$
(5.2)

The voltage gain  $v_o/v_i$  of the LNA of figure 5.4 is proportional to  $Z_{load}$ , being

maximum at the frequency of the poles of  $Z_{load}$  and minimum at the zeros frequency. The load impedance (5.2) has two pole pairs located at:

$$\omega_{1,2}^2 = \frac{C_{11}L_{11} + C_{22}L_{22} \pm \sqrt{\zeta}}{2C_{11}C_{22}(L_{11}L_{22} - M^2)}$$
(5.3)

where

$$\zeta = (C_{11}L_{11} + C_{22}L_{22})^2 + 4C_{11}C_{22}(M^2 - L_{11}L_{22})$$
(5.4)

The load impedance has also one zero at the origin and a pair of imaginary zeros  $j\omega_z$ , with

$$\omega_z^2 = \frac{L_{11} + \alpha M}{C_{22}(L_{11}L_{22} - M^2)} = \frac{L_{11} + \alpha M}{C_{22}L_{11}L_{22}(1 - k^2)}$$
(5.5)

The pole pairs are independent from the current ratio  $\alpha$ , while the zero pair depends on it. This means that the current ratio can be used to adjust the zero pair position relatively to the pole pairs.

Observing (5.2), there is a case of particular interest that is when  $k^2 \ll 1$ . In this case, the load impedance simplifies to:

$$Z_{load} \approx \frac{C_{22}L_{22}(L_{11})s^3 + (L_{11} + \alpha M)s}{C_{11}C_{22}(L_{11}L_{22})s^4 + (C_{11}L_{11} + C_{22}L_{22})s^2 + 1}$$
(5.6)

and the two pole pairs are located at

$$\begin{cases} \omega_1^2 \approx \frac{1}{L_{11}C_{11}} \\ \\ \omega_2^2 \approx \frac{1}{L_{22}C_{22}} \end{cases}$$

$$(5.7)$$

while the zero pair is placed approximately at:

$$\omega_z^2 \approx \frac{L_{11} + \alpha M}{C_{22}L_{11}L_{22}} \approx \frac{1}{C_{22}L_{22}} + \frac{\alpha M}{C_{22}L_{11}L_{22}} \approx \omega_2^2 + \frac{\alpha M}{C_{22}L_{11}L_{22}}$$
(5.8)

Figure 5.6 represents a possible plot of the load impedance module as a function of frequency. To ensure that the zero is placed between the two poles, the pole at  $\omega_2$  has to be placed at the lower frequency, because, from (5.8), the zero is placed at a higher frequency than  $\omega_2$ .

The current ratio  $\alpha$  can be used to control the complex zero position. This ratio is changed by changing the cascode transistors size or by varying their gate bias voltages.



Figure 5.6: Possible plot of the absolute value of the load impedance  $Z_{load}$ .

#### 5.4 Effect of the Inductors Parasitic Resistance

In the last section, the frequency response was determined assuming that the inductors have zero resistance. In this section, the inductors have a parasitic resistance, meaning that  $sL_{ii}$  is replaced by  $sL_{ii} + r_{ii}$ , where  $r_{ii}$  is the series resistance. Making this substitution in (5.6),

$$Z_{load} = \frac{sC_{22}(sL_{22} + r_{22})(sL_{11} + r_{11}) + (sL_{11} + r_{11} + \alpha sM)}{(sC_{11}(sL_{11} + r_{11}) + 1)(sC_{22}(sL_{22} + r_{22}) + 1)}$$
(5.9)

To simplify the analysis of (5.9), it is assumed that the lower frequency zero is still located at the origin, i.e. the first zero is at a much lower frequency than the first pole  $\omega_2$ .

$$\omega_z = \frac{r_{11}}{L_{11} + \alpha M} \ll \omega_2 \tag{5.10}$$

 $Z_{load}$  becomes

$$Z_{load} = \frac{v_1}{i_1} = s \frac{C_{22}L_{11}L_{22}s^2 + C_{22}(L_{11}r_{22} + L_{22}r_{11})s + L_{11} + \alpha M}{(C_{11}L_{11}s^2 + r_{11}C_{11}s + 1)(C_{22}L_{22}s^2 + r_{22}C_{22}s + 1)}$$
(5.11)

The quality factor of the poles can be obtained by comparison of the two factors of the denominator of (5.11) with  $s^2 + \frac{\omega_p}{Q_p}s + \omega_p^2$ :

$$Q_1 = \frac{1}{r_{11}} \sqrt{\frac{L_{11}}{C_{11}}} \tag{5.12}$$

$$Q_2 = \frac{1}{r_{22}} \sqrt{\frac{L_{22}}{C_{22}}} \tag{5.13}$$

The pole pair frequencies are

$$\omega_1^2 \approx \frac{1}{L_{11}C_{11}} \tag{5.14}$$

$$\omega_2^2 \approx \frac{1}{L_{22}C_{22}} \tag{5.15}$$

The same procedure is used to determine the zeros' quality factor and frequency:

$$Q_z = \frac{\sqrt{L_{11}L_{22}}\sqrt{L_{11} + \alpha M}}{L_{11}r_{22} + L_{22}r_{11}}$$
(5.16)

$$\omega_z^2 = \frac{L_{11} + \alpha M}{C_{22} L_{22} L_{11}} \tag{5.17}$$

The frequencies of the poles and zeros are equal to those in last subsection, where the inductor series resistance is neglected. This means that the LNA gain is affected, but the poles location is not altered. In figure 5.7 the effect caused by the inductor series resistances is represented.



Figure 5.7: Possible plot of the load impedance  $Z_{load}$ , considering the inductor series resistances.

## 5.5 Simulation Results

To evaluate the feasibility of this dual-band LNA, the circuit of figure 5.4 is designed using the AMS 0.35  $\mu$ m CMOS technology and a 3 V supply. The frequencies of 900 MHz and 1.8 GHz are again chosen. The cascode transistors are designed with the same dimensions and the gate of  $M_{c1}$  is connected to  $V_{DD}$ . All transistors are designed with minimum length to improve the transition frequency, and the parasitic series resistance of the output network inductors is taken into account.

#### 5.5.1 Circuit Dimensioning

The magnetic coupling coefficient is choosen as k = 0.35, to make  $k^2$  much lower than 1, but still with significant magnetic coupling.

Inductances  $L_{11} = 6.4$  nH and  $L_{22} = 10.0$  nH are implemented, respectively, in metal layers 3 and 4 (top metal). They use different metal layers to be overlapped, to obtain the desired value of k, which is adjusted by shifting horizontally one inductor with respect to the other [68]. The magnetic coupling is not determined by Spectre models, and it is necessary to use an electromagnetic simulator: ASITIC is used in this example [67].

As  $k^2 \ll 1$ , equations (5.7) can be used to obtain  $C_{11} = 1.2$  pF and  $C_{22} = 3.1$  pF.

The cascode transistors are equal,  $V_{BIAS1} = V_{DD}$ , and  $V_{BIAS2}$  can be varied 0 and  $V_{DD}$  ( $\alpha$  varies between 0 and 1). Using (5.5), the zero pair frequency will be between 1.05 GHz for  $\alpha = 0$  and 1.26 GHz for  $\alpha = 1$ . The geometrical mean between the two resonant frequencies,  $\sqrt{\omega_1 \omega_2}$ , is approximately 1.27 GHz; thus, to place the zero frequency between the pole frequencies,  $\alpha$  should be 1; i.e., both cascode transistors should be equal and have the same gate bias voltage. After some iterations, the required values listed in table 5.1 were obtained.

Figure 5.8 shows the voltage gain, in which the two maxima are approximately at the desired frequencies. The minimum voltage gain is located at approximately 1.3 GHz, which corresponds to the expected value. Figure 5.9 presents the input matching impedance, in terms of parameter  $S_{11}$ . The circuit is correctly matched at the desired frequencies ( $S_{11}$  is close to -10 dB). Finally, figure 5.10 represents the noise figure simulation compared with the theoretical result (4.12) obtained in chapter 4, using the values of table 5.1.

Parameter	Value	Parameter	Value	
$W_{M1}$	$800 \ \mu m$	$L_S$	1.2  nH	
$W_{M_{c1}}$	$36~\mu{ m m}$	$L_{11}$	6.4  nH	$(r_{11} = 7 \ \Omega)$
$W_{M_{c2}} \ (\alpha \in [0,1])$	$36~\mu{ m m}$	$C_{11}$	$1.3 \mathrm{\ pF}$	
$L_1$	9.0  nH	$L_{22}$	10.0  nH	$(r_{22} = 12 \ \Omega)$
$C_1$	$700~\mathrm{fF}$	$C_{22}$	2.6  pF	
$V_{BIAS1}$	$V_{DD}$	$V_{BIAS2}$	$[0,V_{DD}]$	
$C_x$	1.2  pF	k	0.35	

Table 5.1: Element values of the concurrent dual-band CMOS LNA using coupled inductors.



Figure 5.8: Voltage gain.



Figure 5.9:  $S_{11}$  simulation of the concurrent dual-band LNA.



Figure 5.10: Noise figure: theoretical values compared with the simulation results.

#### 5.5.2 Voltage Gain Variation

As determined in (5.7) and (5.8), and considering  $k^2 \ll 1$ , the pole frequencies are independent of  $\alpha$ :

$$\begin{cases} \omega_1 \approx \sqrt{\frac{1}{L_{11}C_{11}}} \\ \omega_2 \approx \sqrt{\frac{1}{L_{22}C_{22}}} \end{cases}$$
(5.18)

and the zero frequency depends on  $\alpha$ :

$$\omega_z^2 = \frac{L_{11} + \alpha M}{C_{22}(L_{11}L_{22} - M^2)} \tag{5.19}$$

By varying  $\alpha$  it is possible to change the zero pair position and place it over one of the pole pairs in order to cancel it.  $\alpha$  is changed by varying  $V_{BIAS2}$ , while keeping  $V_{BIAS1} = V_{DD}$ , or by keeping the bias voltages constant and varying the transistors sizes. Varying  $V_{BIAS2}$  while keeping  $V_{BIAS1}$  constant ensures that the cascode transistor connected directly to the output is always in saturation. If both cascode transistors are equal,  $\alpha$  can only vary between 0 and 1; however, if the W/L ratio between the cascode transistors is increased, it is possible to obtain a wider range of variation for  $\alpha$ , when varying the bias voltage  $V_{BIAS2}$ . From now on, when  $\alpha$ varies, it is always assumed that the cascode transistor sizes and  $V_{BIAS1}$  are fixed and only  $V_{BIAS2}$  changes.

The zero pair frequency has a minimum for  $\alpha = 0$ , which from (5.5) is:

$$\omega_z^2 = \frac{L_{11}}{C_{22}(L_{11}L_{22} - M^2)} = \frac{1}{C_{22}L_{22}(1 - k^2)}$$
(5.20)

Thus, the zero pair, cannot cancel the pole pair due to  $L_{22}$  and  $C_{22}$ . Note that (5.20) is not valid for k = 0 because the output network inductors are independent. As  $\alpha$  increases, the zero approaches the higher frequency pole, and there is compensation,  $\omega_z = \omega_1$ , if:

$$\alpha = \frac{1}{M} \left( \frac{C_{22}L_{22}(1-k^2)}{C_{11}} - L_{11} \right)$$
(5.21)

In figure 5.11 the zero frequency variation with  $\alpha$  is represented, using (5.5) and the values listed in table 5.1.



Figure 5.11: Zero frequency versus  $\alpha$ .

If the two cascode transistors are equal,  $\alpha$  ranges between 0 and 1 and the zero frequency is between 1.1 GHz and 1.3 GHz approximately. This is a small variation, illustrated in figure 5.12. The gain variation at both maxima frequencies reaches almost 7 dB.

When  $M_{c2}$  is six times larger than  $M_{c1}$ ,  $\alpha$  ranges from 0 to 6. Observing figure 5.11, the zero frequency is expected to range between 1.1 GHz ( $\alpha = 0$ ) and 2.0 GHz ( $\alpha = 6$ ). It is now possible to cancel the pole pair placed at the higher frequency. There is a large variation of the gain at 1.8 GHz, as shown in figure 5.13, where the second maximum varies around 22 dB, while the first maximum only varies 10 dB, approximately. It is possible to observe the zero frequency moving from 1.1 GHz to 1.8 GHz approximately (the last zero pair frequency is not clear due to the pole cancelation).

By using the output network of figure 5.4 it is not possible to cancel the lower frequency pole, but this is possible by reversing the direction of the magnetic coupling as shown in the next section.



Figure 5.12: Voltage gain with equal cascode transistors:  $\alpha$  varies from 0 to 1  $(V_{BIAS1} = V_{DD} \text{ and } 0 \le V_{BIAS2} \le V_{DD}).$ 



Figure 5.13: Voltage gain with unequal cascode transistors :  $\alpha$  varies from 0 to 6  $(V_{BIAS1} = V_{DD} \text{ and } 0 \le V_{BIAS2} \le V_{DD}).$ 

## 5.6 Effect of Reversing the Orientation of the Magnetic Coupling

One limitation of the dual-band LNA presented in figure 5.4 is that only gain at the higher frequency band can be significantly reduced, since the zero pair cannot overlap the lower frequency pole pair. Canceling the lower frequency pole pair is possible simply by reversing the magnetic inductors coupling orientation, as shown in figure 5.14.



Figure 5.14: Concurrent dual-band LNA topology with opposite magnetic coupling.

Equations (5.1c) to (5.1f) still apply to the circuit of figure 5.14. Only equations (5.1a) and (5.1b) are different:

$$V_1 = sL_{11}I_{l1} - sMI_{l2} (5.22a)$$

$$V_2 = -sMI_{l1} + sL_{22}I_{l2} \tag{5.22b}$$

The load impedance  $Z_{load} = V_1/I_1$  becomes

$$Z_{load} = \frac{C_{22}(L_{11}L_{22} - M^2)s^3 + (L_{11} - \alpha M)s}{C_{11}C_{22}(L_{11}L_{22} - M^2)s^4 + (C_{11}L_{11} + C_{22}L_{22})s^2 + 1}$$
(5.23)

The only difference between (5.23) and (5.2) is in the sign of the  $\alpha M$  term, in the numerator. The poles are the same as before; however, the zero pair is different:

$$\omega_z^2 = \frac{L_{11} - \alpha M}{C_{22}(L_{11}L_{22} - M^2)} = \frac{L_{11} - \alpha M}{C_{22}L_{11}L_{22}(1 - k^2)}$$
(5.24)

Using the same values as in the last example, it is shown in figure 5.15 - curve 1 that the zero pair frequency decreases with  $\alpha$  (instead of increasing as in figure 5.11). The decreasing slope is higher than the increasing slope; meaning that the same variation in current produces a larger zero frequency variation in this case.

In figure 5.16 the gain variation is represented, and it is visible that the zero pair frequency is always too close to the first pole pair frequency. Consequently the maximum voltage gain at 900 MHz is low even for the lowest value of  $V_{BIAS1}$ .



Figure 5.15: Zero position versus  $\alpha$ . Curve 1: k = 0.35. Curve 2: k = 0.50.

Observing (5.24) it is possible to increase the zero pair frequency, when  $\alpha = 0$  by increasing k, or by decreasing  $C_{22}$  or  $L_{22}$ . Since  $L_{22}$  is difficult to vary and the value of  $C_{22}$  depends on  $L_{22}$ , it is simpler to change k. By using k = 0.50, the curve 2 of figure 5.15 is obtained. The zero frequency increases almost 100 MHz when  $\alpha = 0$ . The trade-off of curve 2 is an increase in the slope of the the zero frequency as a function of  $\alpha$ . In figure 5.17 the new voltage gain curves variation as function of  $\alpha$ are represented. The zero frequency, when  $\alpha = 0$ , is now closer to the geometrical mean of the pole frequencies, which is approximately 1.27 GHz. In this situation the voltage gain maxima have approximately the same value of 10 dB. The voltage gain variation of the first pole pair frequency is now around 12 dB. Due to the variation of k the lower frequency pole reduced slightly its frequency.



Figure 5.16: Voltage gain frequency response versus  $\alpha$ .  $\alpha$  in the range 0 to 1  $(V_{BIAS2} \in [1, 1.9] \text{ V})$  and k = 0.35.



Figure 5.17: Voltage gain frequency response versus  $\alpha$ .  $\alpha$  in the range 0 to 1  $(V_{BIAS2} \in [1, 1.2] \text{ V})$  and k = 0.50.

### 5.7 Conclusions

In this chapter, a new concurrent dual-band LNA with a single input and single output is presented. It is based on a cascode stage using inductive degeneration, with a wideband input impedance matching and two resonant output branches which are magnetically coupled. This circuit allows the current balancing between the two output branches.

Equations for the frequency of the maxima and minimum of the frequency response have been determined. It was shown that the maxima of the frequency response are independent of the current ratio between output branches, but the minimum depends on it. This fact allows the gain variation at each maximum by varying the current ratio.

The current ratio is determined by the DC voltage at the gate of the cascode transistors and by the relative size between those transistors. Thus, by changing the ratio between the cascode transistors width, it is possible to control the range of variation of the current ratio, and with that, obtain a different frequency response. Under certain conditions, it is possible to cancel the voltage gain maximum located at the higher frequency; however, the maximum located at the lower frequency cannot be canceled, which is a limitation of this circuit.

To overcome the problem of canceling the maximum located at the lower frequency, it is proposed a new circuit, which differs only in the magnetic coupling orientation, when compared with the first circuit analyzed. It is shown, that with this circuit it is possible to cancel the lower frequency maximum; however, the ratio of frequency variation of the voltage gain minimum is higher than the first circuit.

# Chapter 6

# Double Loop Feedback LNA

6.1	Introduction	111
6.2	DLF LNA with Ideal Blocks	112
6.3	Feedback Network Analysis	114
6.4	Amplifying Block Analysis	122
6.5	Simulation Results	129
6.6	Noise Analysis	136
6.7	Conclusions	141

### 6.1 Introduction

In chapter 3, it was shown that double loop feedback (DLF) can be applied in the design of low noise amplifiers (LNAs) to have a finite input impedance, suitable gain and good noise performance. Four topologies have been considered, two of which sampling the output voltage and the remaining sampling the output current - figures 3.32, 3.36, 3.37 and 3.38. In this chapter the design of a double loop feedback low noise amplifier (DLF LNA) is developed. The design is divided into two major parts:

- 1. the feedback network design;
- 2. the amplifying block design;

In section 6.2 the DLF LNA is analyzed considering all blocks ideal. In section 6.3 the amplifying block is assumed to be ideal, and the influence of the main feedback network non-idealities on the DLF LNA performance is analyzed. After analyzing the feedback network, the possibility of designing an amplifying block, that meets the performance limits determined in section 6.3, is evaluated in section 6.4. In section 6.5 the feedback network and the amplifying block are replaced by real elements, and the DLF LNA is simulated in order to evaluate the reliability of the LNA. In section 6.6 the noise performance of the DLF LNA is evaluated. Finally, in section 6.7, some conclusions are drawn.

#### 6.2 DLF LNA with Ideal Blocks

From the four DLF LNA topologies determined, two of them were partially analyzed at the date of this work (DLF LNAs of figures 3.36 and 3.37). From the remaining topologies, it was chosen the DLF LNA sampling the output voltage because LNAs have more often outputs in voltage suitable to be connected to mixers. The DLF LNA to be analyzed is presented in figure 6.1.



Figure 6.1: DLF LNA type 1, sampling the output voltage.

In this circuit, the transformer samples the output voltage  $v_o$  (winding 2), divides it by a factor n, and compares it with the input voltage  $v_i$  (winding 1). The resistor  $R_f$  samples the output voltage  $v_o$ , converts it into a current, which is compared with the input current  $i_i$ . The current through  $R_f$  is injected at the node between the voltage source and the transformer winding 1; thus, no current flows through this winding. The double loop feedback allows this amplifier to have a finite input impedance, which can be matched to the source impedance.

The input impedance  $Z_{in}$  and the voltage gain  $A_v$ , assuming an ideal transformer and an ideal amplifying block, are:

$$Z_{in} = \frac{v_i}{i_i} = \frac{R_f}{n+1} \tag{6.1}$$

$$A_v = \frac{v_o}{v_s} = -\frac{n}{2} \tag{6.2}$$

The input impedance matching condition  $Z_{in} = R_S = R_f/(n+1)$  will be assumed throughout this chapter. The input referred noise power spectral density, determined in chapter 3, due exclusively to the feedback resistance  $R_f$  is

$$N_i = 4k_B T R_f \left(\frac{R_S}{R_f}\right)^2 = \frac{4k_B T R_S}{n+1} \tag{6.3}$$

which leads to a minimum achievable noise factor of

$$F_{\min} = 1 + \frac{1}{n+1} \tag{6.4}$$

The three main LNA parameters (input impedance (6.1), voltage gain (6.2), and noise factor (6.4)) are all constant and frequency independent. This means that this circuit can be applied to the design of wideband LNAs. Equations (6.2) and (6.4) show that a large n benefits the gain and the noise performance. This is an important design guideline for this LNA.

The transformer model is now complemented with some non-idealities in order to evaluate the performance degradation caused by them.

#### 6.3 Feedback Network Analysis

Assuming an ideal amplifying block, the DLF LNA performance depends only on the feedback network, (6.1) to (6.4). This means that the feedback network should be designed first, considering an ideal amplifying block. Regarding the feedback network, the major concern is the transformer non-idealities.

In figure 6.2 the transformer model to be used in the further analysis is presented. This model corresponds to the ideal transformer (used in last section), to which inductances are added. Using two inductors magnetically coupled (figure 2.26) difficults the analysis. The 2-port **H** matrix of the transformer model of figure 6.2 is:

$$\mathbf{H} = \begin{pmatrix} sL_{w1} & \frac{1}{n} \\ & \\ -\frac{1}{n} & \frac{1}{sL_{w2}} \end{pmatrix}$$
(6.5)



Figure 6.2: Transformer model.

To make the equivalence between the transformer model of figure 6.2 and the transformer model corresponding to two inductors magnetically coupled and discussed in sub-section 2.5.2.2, their **H** matrixes have to be made equal. The equivalence between both matrixes results in next equalities:

- $L_{w1} = L_{11}(1-k^2),$
- $L_{w2} = L_{22}$ ,

The complete transformer model to be used in the DLF LNA analysis is presented in figure 6.3. The non-idealities considered are: the series resistances  $R_{wi}$  and the inter-winding capacitances  $C_{wi}$ . The transformer non-idealities will establish the lower and upper limit frequencies for the amplifier pass-band.


Figure 6.3: Transformer model including winding resistances  $R_{wi}$  and inter-winding capacitances  $C_{wi}$ 

## **6.3.1** Effect of Inductances $L_{wi}$ and Resistances $R_{wi}$

Figure 6.4 represents the DLF LNA using a non-ideal transformer modeled by an ideal transformer complemented with inductances  $L_{wi}$  and resistances  $R_{wi}$ .



Figure 6.4: DLF LNA with transformer model including inductances  $L_{wi}$  and resistances  $R_{wi}$ .

The input impedance  $Z_{in}(s)$  and the voltage gain  $A_v(s)$  are:

$$Z_{in}(s) = \frac{V_i}{I_i}(s) = \frac{R_f s L_{w2}}{(n+1)sL_{w2} + nR_{w2}} = \frac{R_f L_{w2}}{nR_{w2}} \frac{s}{1 + \frac{s}{\omega_{p1,Zin}}}$$
(6.6)

$$A_{v}(s) = \frac{V_{o}}{V_{s}}(s) = -\frac{n(n+1)(sL_{w2} + R_{w2})}{2(n+1)sL_{w2} + nR_{w2}} = -(n+1)\frac{\left(1 + \frac{s}{\omega_{z1,Av}}\right)}{\left(1 + \frac{s}{\omega_{p1,Av}}\right)}$$
(6.7)

where  $\omega_{p1,Zin}$ ,  $\omega_{z1,Av}$  and  $\omega_{p1,Av}$  are

$$\omega_{p1,Zin} = \frac{nR_{w2}}{(n+1)L_{w2}} \tag{6.8}$$

$$\omega_{p1,Av} = \frac{1}{2} \frac{n}{n+1} \frac{R_{w2}}{L_{w2}} \tag{6.9}$$

$$\omega_{z1,Av} = \frac{R_{w2}}{L_{w2}} \tag{6.10}$$

The first major difference between the input impedance and voltage gain equations determined for the ideal case and equations (6.6) and (6.7) is the frequency dependence of these two network functions.  $L_{w1}$  and  $R_{w1}$  do not appear in these equations, because the input impedance of the amplifying block is infinite and no current flows through winding 1. This is an advantage that results from the injection of the feedback current on the node between the voltage source and the transformer winding 1.  $L_{w2}$  and  $R_{w2}$  produce a pole and a zero in both functions, and the asymptotic Bode plots are shown in figures 6.5 and 6.6.



Figure 6.5: Asymptotic Bode plot of the input impedance  $|Z_{in}(f)|$  of the DLF LNA considering the transformer inductances  $L_{wi}$  and resistances  $r_{wi}$ .

Among the poles and zeros of both functions (input impedance and voltage gain),



Figure 6.6: Asymptotic Bode plot of the DLF LNA voltage gain  $|A_v|$  considering the transformer inductances  $L_{wi}$  and resistances  $r_{wi}$ .

the zero  $\omega_{z1,Av}$  of the voltage gain is placed at the highest frequency. For frequencies above  $f_{z1,Av}$ , the DLF LNA voltage gain and input impedance equal the constant values determined in (6.1) and (6.2); thus  $f_{z1,Av}$  determines the lower frequency of the DLF LNA pass-band.

$$f_{\min} = \frac{1}{2\pi} \frac{R_{w2}}{L_{w2}} \tag{6.11}$$

Concerning noise, the parasitic resistances  $r_{wi}$  are a new source of noise, as represented in figure 6.7.



Figure 6.7: DLF LNA noise sources.

This extra noise is accounted for the following equation of the input noise power

spectral density:

$$N_i = 4k_B T [R_f + R_{w2} + R_{w1}] \left(\frac{1}{n+1}\right)^2 + 4k_B T R_{w1}$$
(6.12)

(6.12) is demonstrated in appendix D. Considering that  $R_f \gg R_{wi}$  and  $i \in \{1, 2\}$ , these resistances can be neglected in the first term.  $R_{w2}$  is not relevant to the overall noise; however, the noise contribution of  $R_{w1}$  is fully present at the input (there is no attenuation by the topology). The minimum noise factor achievable by this topology is now:

$$F = 1 + \frac{N_i}{N_S} \approx 1 + \frac{1}{n+1} + \frac{R_{w1}}{R_S}$$
(6.13)

which shows the importance of keeping  $R_{w1}$  with a low value.

#### 6.3.2 Effect of Inter-Winding Capacitances $C_{wi}$

The inter-winding parasitic capacitances  $C_{w1}$  and  $C_{w2}$  model the capacitance existing between the two transformer windings and are represented in figure 6.8. Assuming that  $n \gg 1$ ,  $C_{w2} = C_{w1}$ , and  $R_f = R_S(n+1)$ , the position of poles and zeros of the input impedance  $|Z_{in}(s)|$  and voltage gain  $A_v(s)$  is defined by (6.14) and (6.15). The complete determination of poles and zeros of (6.14) and (6.15) is described in appendix B.

Input Impedance: 
$$\begin{cases} \omega_{z1,Zin} = 0 \\ \omega_{p1,Zin} = -\frac{n}{n+1} \frac{R_{w2}}{L_{w2}} \\ \omega_{p2,Zin} = -\frac{1}{R_f C_{w1}} \\ \omega_{p3,Zin}^2 \approx \frac{1}{nL_{w1}C_{w1}}, \qquad Q_{p2,Zin} \approx \frac{1}{R_{w1}} \frac{1}{\sqrt{n}} \sqrt{\frac{L_{w1}}{C_{w1}}} \\ \omega_{z2,Zin}^2 \approx \frac{1}{L_{w1}C_{w1}}, \qquad Q_{z2,Zin} \approx \frac{1}{R_{w1}} \sqrt{\frac{L_{w1}}{C_{w1}}} \end{cases}$$
(6.14)





Figure 6.8: DLF LNA with transformer model including inductances  $L_{wi}$ , resistances  $R_{wi}$  and, inter-winding capacitances  $C_{wi}$ .

Figures 6.9 and 6.10 show possible plots of the input impedance  $|Z_{in}(s)|$  and the voltage gain  $(v_o/v_s)(s)$  of the DLF LNA.

The input impedance has a zero at the origin and a low frequency pole, both caused by  $L_{w2}$  and  $R_{w2}$ . This pole defines the lower limit of the frequency range in which the input impedance has the desired value for impedance matching. The upper limit of this range is determined by another pole caused by the source resistance  $R_S$  and parasitic capacitances  $C_{w1}$  and  $C_{w2}$ . Above this frequency, one pair of complex poles and one pair of complex zeros are observed. The influence of the



Figure 6.9: Possible plot of the DLF LNA input impedance considering the complete transformer model including inductances, resistances and capacitances.



Figure 6.10: Possible plot of the DLF LNA voltage gain considering the complete transformer model including inductances, resistances and capacitances.

transformer winding 1 is now present, due to the leakage of current through it and via the parasitic capacitances; thus, these capacitances are an important design variable and should be minimized in the transformer implementation. The transfer function  $A_v(s)$  has one pole and one zero at low frequencies, both caused by the winding 2 inductance  $L_{w2}$  and its series resistance  $R_{w2}$ . The amplifier bandwidth is upper limited by a complex pole pair, with a frequency determined by  $L_{w1}$ , the parasitic capacitances between windings, and n. This shows again the influence of these capacitances and the importance of reducing their value. An additional pole, caused by the source resistance  $R_S$  and the capacitances  $C_{w1}$  and  $C_{w2}$  is visible, but it is out of the frequency range of interest. The bandwidth for which the input impedance is approximately  $R_f/(n+1)$  is determined by poles  $\omega_{p1,Zin}$  and  $\omega_{p2,Zin}$ ; the bandwidth where the voltage gain has approximately the expected value of n/2 is determined by the first zero, of frequency  $\omega_{z1,Av}$ , and the complex poles of frequency  $\omega_{p2,Av}$ . The LNA useful bandwidth is determined by the intersection of these two ranges.

Until now, the DLF LNA analysis has been made considering an ideal amplifying block (nullor). The analysis in the following section considers the amplifying block non-idealities and their influence on the DLF LNA performance. The objective is to establish the specifications for the amplifying block, so that it can be considered as a nullor.

# 6.4 Amplifying Block Analysis

In this section the consequences of having a non-ideal amplifying block are analyzed. The emphasis of this analysis is on the effect of the amplifying block non-idealities on the main LNA parameters.

## 6.4.1 Input Impedance and Voltage Gain Evaluation using a Non-Ideal Amplifying Block

The amplifying block is represented by the model in figure 6.11.



Figure 6.11: General amplifier model.

The 2-port  $\mathbf{H}$  parameter matrix of the general amplifier model of figure 6.11 is:

$$\mathbf{H}_{Amp} = \begin{pmatrix} R_x + \frac{1}{Y_i + sC_{io}} & \frac{sC_{io}}{Y_i + sC_{io}} \\ \frac{G_m - sC_{io}}{Y_i + sC_{io}} & Y_o + sC_{io} + \frac{(G_m - sC_{io})sC_{io}}{Y_i + sC_{io}} \end{pmatrix}$$
(6.16)

where

$$Y_o = \frac{1}{R_o} + sC_o \tag{6.17}$$

and

$$Y_i = \frac{1}{R_i} + sC_i \tag{6.18}$$

From (6.16) it is possible to determine all the other matrix descriptions, using transformation rules [43]. In practical implementations,  $R_x$  has a low value compared with  $\left|\frac{1}{Y_i+sC_{io}}\right|$  and, thus, it can be neglected in circuit analysis; however, it should be considered in noise analysis, because the noise generated by it is not negligible in comparison with the other noise sources. The amplifying block without  $R_x$  is shown in figure 6.12 and its **H** matrix is

$$\mathbf{H}_{A} = \begin{pmatrix} \frac{1}{Y_{i} + sC_{io}} & \frac{sC_{io}}{Y_{i} + sC_{io}} \\ \frac{G_{m} - sC_{io}}{Y_{i} + sC_{io}} & Y_{o} + sC_{io} + \frac{(G_{m} - sC_{io})sC_{io}}{Y_{i} + sC_{io}} \end{pmatrix}$$
(6.19)

Figure 6.12: Simplified amplifier model for circuit analysis.

The most suitable matrix description for the series-shunt feedback configuration is the **H** matrix, while for the shunt-shunt configuration is the **Y** matrix [43]. The **H** matrices of the amplifier and transformer are determined first, because the transformer feedback loop is inside the resistor feedback loop. The **H** matrices are added and convert into an **Y** matrix, that will be added to the feedback resistor **Y** matrix.

The transformer is represented in figure 6.13 and its H matrix is

$$\mathbf{H}_{T} = \begin{pmatrix} sL_{11}(1-k^{2}) & -\frac{1}{n} \\ \\ \\ \frac{1}{n} & \frac{1}{sL_{22}} \end{pmatrix}$$
(6.20)

Addition of (6.19) and (6.20) results in:

$$\mathbf{H}_{A,T} = \mathbf{H}_{A} + \mathbf{H}_{T}$$

$$\begin{pmatrix} \frac{1}{Y_{i} + sC_{io}} + sL_{11}(1 - k^{2}) & \frac{sC_{io}}{Y_{i} + sC_{io}} - \frac{1}{n} \\ \frac{G_{m} - sC_{io}}{Y_{i} + sC_{io}} + \frac{1}{n} & Y_{o} + sC_{io} + \frac{(G_{m} - sC_{io})sC_{io}}{Y_{i} + sC_{io}} + \frac{1}{sL_{22}} \end{pmatrix}$$
(6.21)



Figure 6.13: Transformer used in the DLF LNA of figure 6.1.

In practical cases,

$$\left|\frac{1}{Y_i + sC_{io}}\right| \gg \left|sL_{11}(1 - k^2)\right|$$
(6.22)

The matrix **H** of (6.21) is now converted to the corresponding **Y** matrix [43], in order to be added to the **Y** matrix of the feedback block formed by  $R_f$ :

$$\mathbf{Y}_{A,T} = \begin{pmatrix} \frac{1}{h_{11}} & -\frac{h_{12}}{h_{11}} \\ \frac{h_{21}}{h_{11}} & \frac{\det \mathbf{H}}{h_{11}} \end{pmatrix} =$$

$$= \begin{pmatrix} sC_{io} + Y_i & \frac{1-n}{n}sC_{io} + \frac{Y_i}{n} \\ G_m + \frac{1-n}{n}sC_{io} + \frac{Y_i}{n} & \frac{G_m}{n} + Y_o + \frac{1}{sL_{22}} + \frac{n^2 - 1}{n^2}sC_{io} + \frac{Y_i}{n^2} \end{pmatrix}$$
(6.23)

The feedback block formed by  $R_f$  is represented in figure 6.14 and its **Y** matrix is

$$\mathbf{Y}_{R_f} = \begin{pmatrix} \frac{1}{R_f} & -\frac{1}{R_f} \\ & & \\ -\frac{1}{R_f} & \frac{1}{R_f} \end{pmatrix}$$
(6.24)



Figure 6.14: Feedback block formed by  $R_f$  used in the DLF LNA of figure 6.1.

Adding  $\mathbf{Y}_{Amp,T}$  of (6.23) and  $\mathbf{Y}_{R_f}$  of (6.24), and assuming that  $n^2 \gg 1$ , results in:

$$\mathbf{Y}_{\text{LNA}} = \mathbf{Y}_{A,T} + \mathbf{Y}_{R_f} = \begin{cases} sC_{io} + Y_i + \frac{1}{R_f} & \xi sC_{io} + \frac{Y_i}{n} - \frac{1}{R_f} \\ \\ G_m + \xi sC_{io} + \frac{Y_i}{n} - \frac{1}{R_f} & \frac{G_m}{n} + Y_o + \frac{1}{sL_{22}} + sC_{io} + \frac{1}{R_f} \end{pmatrix} \end{cases}$$
(6.25)

where  $\xi = \frac{1-n}{n}$ .

This matrix is used to determine the input admittance:

$$Y_{in} = \frac{I_i}{V_i} = y_{11} - \frac{y_{12}y_{21}}{y_{22} + Y_l}$$
(6.26)

and voltage gain of the DLF LNA:

$$A_v = \frac{V_o}{V_i} = -\frac{y_{21}}{y_{22} + Y_l} \tag{6.27}$$

where  $Y_l$  is the load admittance. As the DLF LNA has a low output impedance it is assumed that  $Y_l \ll y_{22}$ .  $Y_{in}$  and  $A_v$  are:

$$Y_{in}(s) = nsC_{io} + Y_i + \frac{1}{R_f} +$$

$$-\frac{sL_{22}(Y_iR_f + (1-n)R_fsC_{io} - n)(Y_iR_f + nG_mR_f + (1-n)R_fsC_{io} - n)}{nR_f(nR_f + sL_{22}(n + G_mR_f + nsC_{io}R_f + nY_oR_f))}$$
(6.28)

$$A_{v}(s) = -\frac{sL_{22}(Y_{i}R_{f} + nG_{m}R_{f} + (1-n)R_{f}sC_{io} - n)}{nR_{f} + sL_{22}(n + G_{m}R_{f} + nsC_{io}R_{f} + nY_{o}R_{f})}$$
(6.29)

If the amplifying block transconductance  $G_m$  is very high:

$$\lim_{G_m \to \infty} Y_{in} = \frac{1+n}{R_f} + nsC_{io} \tag{6.30}$$

$$\lim_{G_m \to \infty} \frac{V_o}{V_i} = -n \tag{6.31}$$

Equation (6.30) shows that it is not possible to obtain a purely resistive input admittance, due to  $C_{io}$ , even if the amplifying block gain is infinite; Thus,  $C_{io}$  should be reduced, or canceled. From (6.31) it is noted that when the amplifer transconductance is very high the voltage gain does not depend on  $C_{io}$ .

If  $C_{io}$  can be neglected in (6.30),

$$\frac{n+1}{R_f} \gg |nsC_{io}| \tag{6.32}$$

Then,

$$Y_{in} \approx Y_i + \frac{1}{R_f} - \frac{sL_{22}(Y_iR_f - n)(Y_iR_f + nG_mR_f - n)}{nR_f(nR_f + sL_{22}(n + G_mR_f + nY_oR_f))}$$
(6.33)

$$A_v \approx -\frac{sL_{22}(Y_iR_f + nG_mR_f - n)}{nR_f + sL_{22}(n + G_mR_f + nY_oR_f)}$$
(6.34)

In the following, it is assumed that  $C_{io}$  is negligible, and, the conditions for which the gain of the amplifying block is high enough, for it to be assumed to be ideal, will be determined.

#### 6.4.2 Conditions for an Ideal Amplifying Block

In equation (6.33) the terms with  $G_m$  will dominate if  $G_m$  is higher than a certain value. When  $G_m$  is high enough, the third term tends to  $n/R_f - Y_i$ , leading to  $Y_{in} = (n+1)/R_f$ . To approach this limit three conditions have to be satisfied:

- condition 1:  $|nG_mR_f| \gg |Y_iR_f n| \Leftrightarrow G_m \gg \left|\frac{Y_i}{n} \frac{1}{R_f}\right|$
- condition 2:  $|G_m R_f| \gg |nR_f Y_o + n| \Leftrightarrow G_m \gg \left| n \left( Y_o + \frac{1}{R_f} \right) \right|$

If condition 2 is satisfied then a third condition has to be verified to guarantee an approximation to (6.1):

• condition 3: 
$$|G_m R_f s L_{22}| \gg |nR_f| \Leftrightarrow |G_m| \gg \left|\frac{n}{sL_{22}}\right|$$

Using (2.55), condition 3 can be rewritten as:

• condition 3:  $G_m \gg \left| \frac{1}{k} \frac{1}{s\sqrt{L_{11}L_{22}}} \right|$ 

Condition 3 gives a guideline for the transformer design: for a given n, large inductances are preferable, since the condition is satisfied for a lower  $G_m^{-1}$ .

If  $Y_i$  and  $Y_o$  are negligible in conditions 1 and 2, they can be reduced to:

• condition 1':  $G_m \gg \left| -\frac{1}{R_f} \right|$ 

• condition 2': 
$$G_m \gg \left| \frac{n}{R_f} \right|$$

In conditions 1 to 3, three inequations are determined. The second term of these three conditions is represented in figure 6.15 together with a dotted line representing the maximum of these three terms. Condition 3 dominates at low frequencies, while condition 2 dominates at high frequencies (the graphic assumes that  $|nY_o| > |Y_i/n|$ ). The point **A** of figure 6.15 corresponds to the point where both conditions intercept. If the intercept point **A** happens at a frequency lower than the point from which the influence of  $|nY_o|$  starts, then **A** corresponds to an admittance of  $n/R_f$ . This value is approximately the source impedance  $R_S$  in an impedance matching condition,  $Z_{in} = R_f/(n+1) = R_S$ . It is concluded that, in an input impedance matching situation, the amplifying block transconductance  $G_m$  must be at least one decade above the source admittance  $1/R_S$  to consider the amplifying block as ideal. The minimum value of  $G_m$  dominates for the minimum bandwidth (between  $\omega_1$  and  $\omega_2$ ). If an higher value of  $G_m$  is used, then the bandwidth, for which the amplifying block can be considered ideal, increases (band between  $\omega'_1$  and  $\omega'_2$ ).

<sup>&</sup>lt;sup>1</sup>In other words, having two transformers with same n it is preferable to use the one that has larger inductances because it lowers the transconductance needed; and thus, (eventually) saving power -  $G_m$  is typically directly related to power. This should be traded with other transformer non-idealities and their influence on the overall DLF LNA performance.



Figure 6.15: Conditions 1 to 3 along frequency and representation of the minimum  ${\cal G}_m$  determination.

With respect to the voltage gain and using (6.34), the transconductance boundaries, where amplifying block can be considered ideal, lead also to conditions 1 to 3.

In next section, a complete DLF LNA is sized and simulated. The transformer is sized using technology parameters and the amplifying block is replaced by a circuit using transistors. Simulations will evaluate the theory developed in this section and will attest if the DLF LNA is realizable.

## 6.5 Simulation Results

In this section, practical considerations concerning the design of a DLF LNA are developed. The DLF LNA design is divided into two successive tasks:

- 1. feedback network design;
- 2. amplifying block design.

In the example here, the AMS 0.35  $\mu$ m BiCMOS SiGe technology is choosen because it was available (under the project RF Chameleon). Bipolar transistors were preferred since they have higher transconductance than CMOS transistors for the same bias current.

### 6.5.1 Feedback Network Design

Assuming an ideal amplifying block and an ideal transformer the DLF LNA input impedance is given by (6.1):

$$Z_{in} = \frac{R_f}{n+1} \tag{6.35}$$

To have matched input impedance,  $Z_{in}$  must equal the source impedance  $R_S$ , which leads to

$$R_f = (n+1)R_S (6.36)$$

The voltage gain,  $A_v = -n/2$ , given by (6.2), is proportional to  $n = \frac{1}{k}\sqrt{\frac{L_{22}}{L_{11}}}$ , which in an integrated circuit is limited. A large *n* requires either large  $L_{22}$  or small  $L_{11}$ , but a large inductance leads to a large series resistance and a large area, and a small inductance is not accurate (reliably feasible in integrated circuits). An alternative is to have a low magnetic coupling coefficient k; however, if it is made too low, inductors become independent. The integrated transformer dimensioning will be presented in chapter 7. Its parameters are listed in table 6.1, and will be used in the LNA design.

Using the values of table 6.1, and considering the standard 50  $\Omega$  value for the source impedance,  $R_f = 457.5 \ \Omega$  is obtained from (6.36). The poles and zeros of the input impedance and voltage gain determined by (6.14) and (6.15), are listed in table 6.2. With these values, it is possible to estimate the band in which the DLF LNA has a constant impedance at high gain. This band is from 140 MHz to 5.8 GHz.

Parameter	Value
$L_{11}$	$0.98 \ \mathrm{nH}$
$L_{w1} = L_{11}(1 - k^2)$	$0.64 \ \mathrm{nH}$
$L_{22} = L_{w2}$	22.65  nH
$r_{w1}$	$9.8 \ \Omega$
$r_{w2}$	$20.5 \ \Omega$
$C_{w2}$	$60~\mathrm{fF}$
k	0.59
<i>n</i>	8.15

Table 6.1: Transformer parameters.

Table 6.2: Frequencies of the poles and zeros determined in (6.14) and (6.15).

Input	Impedance	Volt	age Gain
$f_{z1,zin}$	0 Hz	$f_{p1,av}$	64 MHz
$f_{p1,zin}$	$120 \mathrm{~MHz}$	$f_{z1,av}$	$140 \mathrm{~MHz}$
$f_{p2,zin}$	$5.8~\mathrm{GHz}$	$f_{p2,av}$	$9.0~\mathrm{GHz}$
$f_{p3,zin}$	$9.0~\mathrm{GHz}$	$f_{p3,av}$	$53.1~\mathrm{GHz}$
$f_{z2,zin}$	$25.7~\mathrm{GHz}$		

## 6.5.2 Amplifying Block Design

The amplifying block can be designed in order to be approximately ideal using the guidelines obtained in section 6.4. The design flow can be divided into:

- 1. topology definition;
- 2. transistor sizing and biasing.

In the topology definition, the number of stages of the amplifying block and the configuration of each stage have to be choosen. A single stage topology is choosen for the following reasons:

- for high frequency performance, circuit complexity should be kept low;
- a single stage amplifier does not require frequency compensation;
- more than one stage increases the complexity of the biasing circuits.

The simplest choice is the common emitter stage; however, it is impossible to cancel the feed-forward effect of  $C_{\mu}$  with it [122].

The cascode stage will be used to reduce the feed-forward effect of  $C_{\mu}$ . The cascode stage is represented in figure 6.16 and its approximate incremental circuit is represented in figure 6.17 [42, 89].



Figure 6.16: Cascode stage.

This topology reduces the Miller effect due to  $C_{\mu 1}$ , improves the reverse isolation of the amplifier, and increases the amplifying block output impedance in comparison with the common emitter stage [85–89]. Resistances  $R_{b1}$  and  $R_{b2}$  are neglected. It is also considered that the input-output capacitance (between  $b_1$  and  $c_2$ ) is negligible.



Figure 6.17: Simplified incremental model of the cascode stage.

 $Y_i$  and  $Y_o$ , are

$$Y_i = \frac{1}{R_{\pi 1}} + s(C_{\pi 1} + C_{\mu 1}) \tag{6.37}$$

and

$$Y_o = \frac{1}{r_{o1}g_{m2}r_{o2}} + sC_{\mu 2} \tag{6.38}$$

The transconductance  $g_m$  of a bipolar transistor is mainly proportional to its collector current  $I_C$ :

$$g_m = \frac{I_C}{V_T} \tag{6.39}$$

where  $V_T$  is the thermal voltage<sup>2</sup>. Thus,  $g_{m1}$  determines the power consumption of the amplifying block.

Conditions 1 to 3, in section 6.4 define the  $g_{m1}$  boundaries for which the amplifying block is approximately ideal. The minimum value of  $g_{m1}$  is determined by the three conditions, while the maximum value is limited by the power consumption. The  $g_{m1}$ minimum value is determined analytically by the minimum value that satisfies the three inequations of (6.40).

$$\begin{cases} g_{m1} \gg \left| \frac{Y_i}{n} - \frac{1}{R_f} \right| \\ g_{m1} \gg \left| nY_o + \frac{n}{R_f} \right| \\ g_{m1} \gg \left| \frac{n}{sL_{22}} \right| \end{cases}$$
(6.40)

 $Y_i$  and  $Y_o$  are given respectively by (6.37) and (6.38).

It was concluded in subsection 6.4.2 that  $g_{m1}$  should be much higher than the source admittance  $1/R_S$ . Assuming that a value for  $g_{m1}$  greater than one order above the source admittance is enough, means that the minimum  $g_{m1}$  that satisfies inequations (6.40) is approximately 0.2 S, obtainable with a current of approximately 5 mA for the bipolar transistor. In figure 6.18,  $g_{m1} = 0.2$  mS is represented together with the second term of the three inequations of (6.40). Values used in inequations were determined by simulation, using a biasing collector current  $I_{C1} = 5$  mA, and are listed in table 6.3. The transistor model used in simulations is the Gummel-Poon,  $Q_1$  and  $Q_2$  have an emitter length of 24  $\mu$ m and 16  $\mu$ m respectively, and both transistors have an emitter width of 0.4  $\mu$ m [123]. From figure 6.18, it is visible that condition 3 is the most restrictive for low frequencies while condition 2 is the one that dominates at higher frequencies.

Figures 6.19 and 6.20 represent the real and the imaginary part of the input impedance. By comparing the theoretical curves (dotted lines) and the simulation curves (solid lines), it is visible that for a certain frequency range the real part is close to 50  $\Omega$ , while the imaginary has a lower value. This corresponds approximately to the zone where  $g_{m1}$  validates the three inequations of (6.40), as visible in figure 6.18. The different curves of figures 6.19 and 6.20 correspond to different

<sup>&</sup>lt;sup>2</sup>The thermal voltage  $V_T$  is equal to  $\frac{k_B T}{q}$  and is typically determined at a temperature T of 300 K giving approximately 25 mV.



Figure 6.18: Amplifying block transconductance and conditions 1 to 3. Transconductance determined for a current of 5 mA and condition curves determined with values of table 6.3.

Model Element	Value	Model Element	Value
$r_{o2}$	71.43 $\Omega$	$R_{\pi 1}$	1021 $\Omega$
$g_{m2}$	$0.167 \mathrm{mS}$	$C_{\pi 1}$	$346~\mathrm{fF}$
$r_{o1}$	94.25 $\Omega$	$g_{m1}$	$0.174~\mathrm{S}$
$R_{b1}$	$20 \ \Omega$	$C_{\mu 2}$	$60~\mathrm{fF}$

Table 6.3: Bipolar cascode incremental parameters.

values of  $g_{m1}$  between 0.09 S and 0.33 S, that correspond to a collector current between 2.5 mA and 10.0 mA. It is visible that higher values of transconductance  $g_{m1}$  correspond to an input impedance closer to real 50  $\Omega$ . It is also visible that the theoretical model follows the simulation results, which means a good approximation between the theoretical and simulated curves. This means that the estimated minimum value of  $g_{m1} \approx 0.2$  S is enough to consider the amplifying block as ideal. In figure 6.21 the  $S_{11}$  parameter is represented, which is well below -10 dB for the above referred frequency range. When  $g_{m1}$  increases,  $S_{11}$  decreases, meaning that the input impedance matching improves. Figure 6.22 represents the voltage gain and it can be seen that the theoretical model and the simulation plots have a difference of approximately 1 dB, for an absolute value above 16 dB, for the zone where  $g_{m1}$  dominates. The simulations do not take into account the transformer inter-winding capacitances  $C_{wi}$ , because they have not been considered for the theoretical curves, and they only serve to validate the theory. In next chapter, the DLF LNA is simulated considering the complete models for the amplifying block and feedback network.



Figure 6.19: Real part of  $Z_{in}$ .



Figure 6.20: Imaginary part of  $Z_{in}$ .



Figure 6.21:  $S_{11}$ .



Figure 6.22: voltage gain.

## 6.6 Noise analysis

The objective of the LNA noise analysis is to determine the total equivalent noise power spectral density referred to the amplifier input [41], and to compare it with the noise power spectral density at the output of the preceding block. Usually the preceding block is an antenna with  $R_S = 50 \ \Omega$  output impedance, which has a noise power spectral density

$$N_S = 4k_B T R_S \tag{6.41}$$

where  $k_B$  is the Boltzmann constant and T is the absolute temperature.

In figure 6.23 the following noise sources are represented:

- $v_{n,A}$ : equivalent noise voltage source due to the amplifying block and referred to its input;
- $i_{n,A}$ : equivalent noise current source due to the amplifying block and referred to its input;
- $v_{n,R_{w1}}$ : noise voltage source of the resistance  $R_{w1}$ ;
- $v_{n,R_f}$ : noise voltage source of the feedback resistance  $R_f$ .



Figure 6.23: DLF LNA noise sources.

In order to simplify the analysis, but keep it reasonably accurate, the following approximations are considered:

1. the noise sources are reported to the amplifying block input, and the amplifying block is a nullor;

- 2. the noise analysis is performed at a frequency above the pole frequency due to  $L_{w2}$  and  $R_{w2}$  and below the frequency at which inter-winding capacitances  $C_{wi}$  starts to influence significantly the LNA parameters. This means that  $C_{wi}$ ,  $L_{w2}$  and  $R_{w2}$  are neglected;
- 3. all noise sources are uncorrelated.

The amplifying block noise power is mainly due to the input transistor  $Q_1$ , it is assumed that  $Q_2$  has little influence on the total input noise power [3]. Three independent noise sources are originated in the input transistor. These noise sources are represented in figure 6.24 and are:

- thermal noise voltage due to the base resistance  $R_b$  of transistor  $Q_1$ :  $v_{n,b}$ ;
- base shot noise current  $i_{n,b}$ ;
- collector shot noise current  $i_{n,c}$ .



Figure 6.24: BJT noise sources for RF design.

The influence of resistor  $R_b$  has been neglected in the input impedance and gain analysis, however concerning noise, this approximation is no longer valid and is now considered. Their power spectral noise densities are respectively [57]:

$$\begin{cases}
N_{Rb}(f) = 4k_B T R_b \\
N_{I_B}(f) = 2q I_B \\
N_{I_C}(f) = 2q I_C
\end{cases}$$
(6.42)

where  $R_b$  is the base resistance of transistor  $Q_1$ ,  $I_B$  and  $I_C$  are respectively the base and collector DC currents and q is the electron charge. It is possible to determine the equivalent current and voltage noise at the transistor input in accordance with figure 6.25. The noise sources  $v_{n,A}$  and  $i_{n,A}$  are given by:

$$\begin{cases} v_{n,A} = v_{n,Rb} + \frac{1}{g_m} (1 + Z_\pi R_b) i_{n,c} \\ i_{n,A} = i_{n,b} + \frac{Y_\pi}{g_m} i_{n,c} \end{cases}$$
(6.43)



Figure 6.25: Equivalent noise sources at the transistor input.

 $Y_{\pi}$  is the admittance due to  $C_{\pi}$ .



Figure 6.26: Equivalent noise sources at the LNA input.

The equivalent input noise voltage source  $v_{n,i}$  and current noise source  $i_{n,i}$ , represented in figure 6.26, are determined using the source transformations presented in appendix C and are demonstrated in appendix D. The input noise voltage is:

$$v_{n,i}' = v_{n,i} + i_{n,i}R_S = \frac{1}{R_f}(R_f + R_S)(v_{n,A} + v_{n,R_{w1}}) + \left(\frac{Z_{w1}}{R_f}(R_f + R_S) + R_S\right)i_{n,A} + \frac{R_S}{R_f}v_{n,R_f}$$
(6.44)

 $Z_{w1} = Z_{Lw1} + R_{w1}, Z_{Lw1}$  is the impedance due to  $L_{w1}$ . Replacing (6.43) in (6.44) and considering  $R_f = R_S(n+1)$ :

$$v_{n,i}' \approx \frac{n+2}{n+1} \left( v_{n,Rb} + v_{n,Rw1} + \frac{1}{g_{m1}} (1+Y_{\pi}R_b)i_{n,c} \right) + \left( \frac{n+2}{n+1} Z_{Lw1} + R_S \right) \left( i_{n,b} + \frac{Z_{\pi}}{g_{m1}} i_{n,c} \right) + \frac{1}{n+1} v_{n,R_f}$$
(6.45)

Rearranging (6.45) to put into evidence all noise sources:

$$v_{n,i}' \approx \frac{n+2}{n+1} (v_{n,Rb} + v_{n,Rw1}) + \left[ \frac{n+2}{n+1} \frac{1}{g_{m1}} + \frac{n+2}{n+1} \frac{Z_{Lw1} Z_{\pi}}{g_{m1}} + \frac{Y_{\pi}}{g_{m1}} \left( \frac{n+2}{n+1} (R_{w1} + R_{b1}) + R_S \right) \right] i_{n,c} + \left( \frac{n+2}{n+1} R_{w1} + R_S + \frac{n+2}{n+1} Z_{Lw1} \right) i_{n,b} + \frac{1}{n+1} v_{n,R_f}$$
(6.46)

Using (6.46) and the Wiener-Khintchine theorem (2.29) the equation of the equivalent input spectral noise density of this DLF LNA is obtained.

$$N_{i}(f) = \left|\frac{n+2}{n+1}\right|^{2} \left(N_{Rb}(f) + N_{Rw1}(f)\right) + \left|\frac{n+2}{n+1}\left(\frac{1}{g_{m1}} - \frac{\omega^{2}L_{w1}C_{\pi}}{g_{m1}}\right) + \frac{j\omega C_{\pi}}{g_{m1}}\left(\frac{n+2}{n+1}(R_{w1}+R_{b}) + R_{S}\right)\right|^{2} N_{I_{C}}(f) + \left|\frac{n+2}{n+1}R_{w1} + R_{S} + \frac{n+2}{n+1}j\omega L_{w1}\right|^{2} N_{I_{B}}(f) + \left|\frac{1}{n+1}\right|^{2} N_{Rf}(f)$$
(6.47)

Considering  $n \gg 1$ , (6.47) can be simplified to:

$$N_{i}(f) = N_{Rb1}(f) + N_{Rw1}(f) + \left| \frac{1}{g_{m1}} - \frac{\omega^{2}L_{w1}C_{\pi}}{g_{m1}} + \frac{j\omega C_{\pi}}{g_{m1}} \left( R_{w1} + R_{b1} + R_{S} \right) \right|^{2} N_{I_{C}}(f) + \left| R_{w1} + R_{S} + j\omega L_{w1} \right|^{2} N_{I_{B}}(f) + \left| \frac{1}{n+1} \right|^{2} N_{Rf}(f)$$

$$(6.48)$$

The noise factor is

$$F(f) = 1 + \frac{N_i(f)}{N_S(f)}$$
(6.49)

It is possible to determine a noise figure value independent on frequency considering only the terms of (6.48) that are independent on frequency:

$$F = 1 + \frac{R_{b1} + R_{w1}}{R_S} + \varphi \frac{I_C}{R_S g_{m1}^2} + \varphi \frac{(R_{w1} + R_S)^2}{R_S} I_B + \frac{1}{n+1}$$
(6.50)

where  $\varphi = (2q)/(4k_BT) \approx 19.34 \text{ V}^{-1}$ . In (6.50) the importance of minimizing  $R_{b1}$ and  $R_{w1}$  is visible. Assuming that  $g_{m1} \gg 10/R_S$  and  $R_S \gg R_{w1}$ , and by taking into account that  $I_C = g_m/V_T$  and  $I_B = I_C/\beta$ , (6.50) becomes:

$$F = 1 + \frac{R_{b1} + R_{w1}}{R_S} + \varphi \left(\frac{1}{10} + \frac{10}{\beta}\right) V_T + \frac{1}{n+1} \approx \\ \approx 1 + \frac{R_{b1} + R_{w1}}{R_S} + 0.5 \left(\frac{1}{10} + \frac{10}{\beta}\right) + \frac{1}{n+1}$$
(6.51)

This result leads to several important guidelines for the DLF LNA design concerning noise:

- $R_{w1}$  and  $R_b$  should be minimized;
- the amplifying block transconductance  $g_{m1}$  should be maximized;
- the transformer voltage ratio n should be maximized.

These guidelines are in accordance with the guidelines concerning the input impedance and voltage gain. Figure 6.27, obtained in the same conditions of figures 6.19 to 6.22, shows the noise figure obtained by (6.47) and by simulation. At low frequencies, as the transconductance increases, the noise figure tends to a constant value.



Figure 6.27: Noise figure.

# 6.7 Conclusions

When considering ideal blocks (ideal transformer and a nullor), the DLF LNA studied in this chapter has frequency independent and constant input impedance, voltage gain and noise figure. This indicates that this LNA is potentially wideband.

When a non-ideal transformer is considered, but the amplifying block is considered ideal, the DLF LNA has frequency dependent parameters. With proper design it is possible to obtain a frequency band where there is input impedance matching, with reasonable gain and low noise figure.

The amplifying block is designed so that it can be considered as ideal. It is shown that by using a single stage amplifying block (cascode stage preferably), it is possible to achieve the desired LNA performance.

Equations were determined for the input impedance, voltage gain and equivalent input spectral noise density of the DLF LNA. The amplifying block transconductance has a minimum value that allows the amplifying block to be considered as an ideal amplifier. The upper limit of the transconductance is set by the power consumption. These two limits indicate whether it is possible to implement the DLF LNA in a given technology with a reasonable power consumption.

The simulation of a complete LNA, considering non-ideal transformer, and an amplifying block designed using transistors, confirms that it is possible to split the LNA design into two distinct tasks: design of the feedback network and design of the amplifying block.

# Chapter 7

# LNA Prototype Design and Experimental Results

7.1	Introduction	145
7.2	DLF LNA Design	146
7.3	DLF LNA Layout	155
7.4	Test Board	163
7.5	Experimental Results	166
7.6	Conclusions	177

# 7.1 Introduction

This chapter is dedicated to the design of a prototype of the double loop feedback low noise amplifier (DLF LNA) studied in chapter 6 and to the presentation of experimental evaluation of a test chip. The DLF LNA was designed using the AMS SiGe BiCMOS 0.35  $\mu$ m technology.

In section 7.2, the DLF LNA design is described from the specifications to the complete circuit layout. In section 7.3, the transformer layout is described, and some considerations about the complete LNA circuit layout are discussed. In section 7.4 the test board design and the chip bonding are described, and in section 7.5 experimental results are presented, and some improvement suggestions are proposed. Finally, in section 7.6, some conclusions are drawn.

## 7.2 DLF LNA Design

The DLF LNA circuit to be prototyped is represented in figure 7.1.



Figure 7.1: DLF LNA to be designed.

The LNA is to be tested using equipment with standard 50  $\Omega$  port impedances; thus, the LNA input and output must be connected to 50  $\Omega$ . The LNA is designed to have input impedance 50  $\Omega$ , but the output impedance is different from 50  $\Omega$ ; thus, it is necessary to use a buffer with 50  $\Omega$  output. In figure 7.2 the different blocks that have to be sized are presented.



Figure 7.2: Block diagram of the prototype (A is the amplifying block and  $\beta$  is the feedback network).

The design is divided into three parts:

- 1. feedback network;
- 2. amplifying block and its biasing;
- 3. buffer and its biasing.

The feedback network is dimensioned firstly considering an ideal amplifying block. Then, the amplifying block (and its biasing circuit) is designed in order to be a good approximation to an ideal amplifying block. Since the DLF LNA has an output impedance different from the standard 50  $\Omega$ , a buffer is required with 50  $\Omega$ output impedance. The circuit is designed with the guidelines listed in table 7.1.

Technology	AMS 0.35 $\mu m$
Voltage Supply	3 V
$S_{11}$	$\lesssim -10 \text{ dB}$
Voltage Gain	around 10 dB
Noise Figure	around 3 dB
Frequency Bandwidth	around 1 GHz

Table 7.1: LNA guidelines.

#### 7.2.1 The Feedback Network Design

The DLF LNA performance, considering ideal blocks, depends only on the feedback network, and the input impedance, voltage gain, and noise factor are:

$$Z_{in} = \frac{R_f}{n+1}$$

$$A_v = \frac{v_o}{v_i} = -n$$

$$F = 1 + \frac{1}{n+1}$$
(7.1)

*n* should be maximized in order to improve the voltage gain, and then  $R_f$  is determined using the input matching condition  $Z_{in} = R_S = 50 \ \Omega$ .

From

$$n = \frac{1}{k} \sqrt{\frac{L_{22}}{L_{11}}} \tag{7.2}$$

to maximize n,  $L_{22}$  should be maximized, while  $L_{11}$  and k should be minimized. The value of n is limited, since a large  $L_{22}$  leads to a large series resistance and a large area, and a small  $L_{11}$  is not feasible reliably in integrated circuits.

Considering the complete transformer windings model, including its most common

parasitics (figure 2.19), the design guidelines for the transformer are:  $r_{22}$  should be minimized to reduce the lower bound of the frequency band and  $r_{11}$  should be minimized to reduce the noise figure.  $C_{w1}$  and  $C_{w2}$  should be low, to increase the upper bound of the frequency band. The transformer layout is described in subsection 7.3.1 and its model element values are listed in table 7.2.

Parameter	Value
$L_{11}$	0.98 nH
$L_{w1} = L_{11}(1 - k^2)$	$0.64 \ \mathrm{nH}$
$L_{w2} = L_{22}$	22.65  nH
$r_{w1}$	$9.8 \ \Omega$
$r_{w2}$	$20.5 \ \Omega$
$C_{w1}$	$60~\mathrm{fF}$
k	0.59
n	8.15

Table 7.2: Transformer parameters.

Using the values listed in table 7.2 and equation (7.1), the feedback resistance  $R_f$  should be 457.5  $\Omega$ .

## 7.2.2 Amplifying Block Design

The amplifying block design follows the guidelines established in chapter 6. Some major decisions concerning the amplifying block design are:

- transistor type;
- number of stages;
- stage topology.

The SiGe BiCMOS AMS 0.35  $\mu \rm m$  technology provides both MOS and bipolar transistors. Due to the need of a high transconductance, bipolar transistors are chosen.

Using more than one stage leads to some problems concerning biasing, coupling between stages (coupling capacitors are not ideal), and stability. Thus, the amplifying block is designed using a single stage. In chapter 6 it is concluded that a cascode configuration should be used. In figure 7.3 the amplifying block implemented with transistors is presented together with the feedback network. Winding 2 is connected to the supply voltage to bias the transistors. The LNA is designed in order to minimize components size, mainly the coupling capacitors size that increase significantly the die area<sup>1</sup>.



Figure 7.3: DLF LNA (biasing of  $Q_1$  not shown).

In chapter 6 is was found that the amplifying block transconductance  $G_m$  should be much higher than the input admittance:

$$G_m \gg \frac{1}{R_S} \tag{7.3}$$

Choosing  $G_m = 10R_S^{-1}$ , with  $R_S = 50 \ \Omega$ , the minimum value of  $G_m$  is 0.2 S. In the cascode stage  $G_m$  is equal to the transconductance  $g_m$  of transistor  $Q_1$ . In a bipolar,

$$g_m = \frac{I_C}{V_T} \tag{7.4}$$

Considering  $V_T$  approximately equal to 25 mV, the minimum DC current to achieve a transconductance of 0.2 S is approximately 5 mA. This value is reasonable in comparison with those used in other LNA architectures.

In figure 7.4 the amplifying block biasing circuit is represented. Transistor  $Q_{B1}$  has the same size of transistor  $Q_1$ .  $R_{B2}$  and  $C_B$  form a low-pass filter that prevents the input high frequency signal  $v_i$  from being injected into the biasing circuit.  $Q_{B2}$ ,

<sup>&</sup>lt;sup>1</sup>The area is dominated by inductors, but good coupling capacitors have also large area.

 $R_{B1}$  and  $R_{B2}$  ensure that  $I_{Bias}$  is approximately equal to  $I_{C,Q1}$  [89, 124]. Transistor  $Q_{B3}$  is used to ensure that the cascode transistor  $Q_2$  never leaves the saturation region.



Figure 7.4: Amplifying block biasing.

## 7.2.3 Buffer Design

The DLF LNA has an output impedance that is not equal to 50  $\Omega$ . To connect the LNA output to the test equipment a buffer is used, that provides 50  $\Omega$  output impedance and has a high input impedance. An emitter-follower is used. It has a high input impedance and the output impedance is approximately  $g_m^{-1}$ , which can be made equal to 50  $\Omega$ . In figure 7.5 the LNA buffer is represented together with the biasing circuit. The biasing circuit is formed by  $Q_{B4}$ ,  $Q_{B5}$ ,  $R_{B3}$ ,  $R_{B4}$ ,  $C_{B1}$ , and is similar to the LNA biasing circuit discussed in the last subsection.



Figure 7.5: Buffer with biasing circuit.
#### 7.2.4 Simulation Results

In figure 7.6 the complete DLF LNA circuit is represented, including the biasing circuit. Note that by using the external coupling capacitor there is no DC current through winding 1. There is no DC current through  $R_f$ , due to  $C_c$ .



Figure 7.6: Complete DLF LNA circuit.

In figures 7.7 to 7.9 are represented, respectively,  $S_{11}$ , the voltage gain, and the noise figure, for different values of the bias current. The values used in the simulations are listed in table 7.3.

Typically, the input impedance matching is considered good when  $S_{11}$  is below -10 dB. Using this criterion, for all bias values the bandwidth of almost one decade is obtained. When the bias current increases, the first frequency at which  $S_{11}$  is -10 dB decreases, which is in accordance with the results obtained in chapter 6; however, there is not a clear rule for the higher frequency where  $S_{11}$  equals -10 dB.

In figure 7.8 the voltage gain is represented for the same bias current values. At low frequencies,  $g_{m1}$  is not dominant over  $|n/(\omega L_{22})|$  (condition 3 of figure 6.15); thus the gain is low. However, when  $g_{m1}$  increases, the voltage gain tends to a constant value. The bandwidth where the voltage gain is almost constant increases as  $g_{m1}$  increases, meaning that  $g_{m1}$  satisfies all the three conditions of figure 6.15 when the



Figure 7.7:  $S_{11}$  with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.

bias current increases.

Figure 7.9 displays the noise figure for the same bias conditions. The noise figure has a flat zone in a wide band; however the value decreases when the transconductance increases, reaching a value of approximately 2.4 dB for the two higher current values. This means that for high values of bias current the amplifying block behaves as an ideal block (high gain and low sensitivity to bias current variations).

Figures 7.10 to 7.12 represent  $S_{11}$ , voltage gain, and noise figure for the same bias conditions of figures 7.7 to 7.9, but considering the transformer resistances and inter-winding capacitances.

In figure 7.10, the input impedance matching is slightly degraded in comparison with figure 7.7, and this happens for all current values. Another difference is the decrease of the bandwidth. This is due to the resonance of  $L_{11}$  with the inter-winding capacitances, as demonstrated in chapter 6.

In figure 7.11, the voltage gain behavior is similar as in figure 7.8 at low frequencies; however, at high frequencies, there is again a resonance, due to  $L_{11}$  and the inter-winding capacitance.



Figure 7.8: Voltage gain with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.



Figure 7.9: Noise figure with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.



Figure 7.10:  $S_{11}$  considering resistances and inter-winding capacitances, with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.



Figure 7.11: Voltage gain considering resistances and inter-winding capacitances, with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.

The noise figure in figure 7.12 has the same behavior as in figure 7.9; however, all curves are approximately 0.6 dB higher within the wanted band. This difference is due to the series resistance of winding 1, and is approximately  $10 \log(1 + r_{11}/R_S)$ , obtained from equation (6.13).



Figure 7.12: Noise figure considering resistances and inter-winding capacitances, with DC current values: 2.5, 5.0, 7.5, and 10.0 mA.

## 7.3 DLF LNA Layout

In this section, the transformer layout is described, and general considerations concerning the DLF LNA layout are presented.

#### 7.3.1 Transformer layout

Concerning the transformer implementation, since the substrate is highly resistive, there is no need for a patterned ground shield which is used in low resistivity substrates. However, when the transformer was designed, this was unknown to the author, and the patterned ground shield represented in figure 7.13 was implemented (the design strategy used is explained in chapter 2).

#### CHAPTER 7. LNA PROTOTYPE DESIGN AND EXPERIMENTAL RESULTS



Figure 7.13: Patterned Ground Shield.

Concerning the design of the windings, several aspects should be taken into account. Area should be a concern, but the need for a large inductance  $L_{22}$  with a reduced series resistance leads to a large winding, due to the large line width required to minimize the series resistance. In [62] a table is presented with an extensive list of inductor values and corresponding dimensions; to obtain a large inductance, around 20 nH, an outer diameter above 200  $\mu$ m and a large number of turns, around 10 are required. Also, in order to minimize the inductor series resistance, the top metal layer 4 is used, due to its larger thickness. The technology has the option of having a thicker layer for the top metal: layers 1, 2 and 3 have a typical thickness of 0.640  $\mu$ m, while the top layer 4 has an average thickness of 0.925  $\mu$ m. This option reduces the series resistance due to the lower metal resistivity and simultaneously reduces the substrate capacitance due to the higher distance to ground of this layer.

From the possible transformer topologies, the stacked transformer was chosen because it has a high magnetic coupling, it is suitable for asymmetric inductors, and its inter-winding capacitance can be minimized. Inductance  $L_{11}$  should be minimum. Metal layers 2 and 3 are used, and the inductance is placed below the other winding. This inductor has one turn. The two metal layers are short-circuited to reduce the winding series resistance [59, 60, 125]. The transformer layout is presented in figure 7.14. Winding 1 is between two turns of winding 2. This was done to reduce the inter-winding capacitance, that would be much higher if the turns of both winding were superimposed.



Figure 7.14: Transformer layout.

Simulations using the program ASITIC produced the output in figure 7.15 and in table 7.3.

Parameter		Winding 1	Winding 2
External diameter	$d_{out}$	$238~\mu\mathrm{m}$	$250~\mu\mathrm{m}$
Spacing between turns	s	N/A	$4.5 \ \mu \mathrm{m}$
Turn width	w	$3.0 \ \mu \mathrm{m}$	$5.0 \ \mu \mathrm{m}$
Winding inductance	$L_{ii}$	$0.83 \ \mathrm{nH}$	22.65  nH
Series resistance	$r_{ii}$	$9.8 \ \Omega$	$20.5 \ \Omega$
Inductor magnetic coupling	k	0.	63
Inter-winding capacitance	$C_{12}$	$134 \mathrm{~fF}$	
	$C_{wi}$	$C_{12}/2 = 67 \; \mathrm{fF}$	
Voltage ratio	n	8.	29

Table 7.3: Main transformer parameters.

\_ 🗆 🗵

```
X ~/Asitic/Asitic
```

```
ASITIC> pi P 2

Pi Model at f=2 GHz: Q = 5.645 , 5.611 , 8.187

L = 22.65 nH R = 20.51

Cs1 = 49.13 fF Rs1 = 1.465 k

Cs2 = 49.77 fF Rs2 = 1.183 k Est. Resonance = 4.771 GHz

ASITIC> pi S 2

Pi Model at f=2 GHz: Q = 1.05 , 1.05 , 1.052

L = 829.5 pH R = 9.898

Cs1 = 9.031 fF Rs1 = 6.689 k

Cs2 = 8.836 fF Rs2 = 6.443 k Est. Resonance = 58.15 GHz

ASITIC> k P S

Coupling coefficient of P and S: k = 0.63079 and M = 2.216 nH.

ASITIC> ■
```

Figure 7.15: ASITIC results for the transformer.

### 7.3.2 General Layout Considerations

The final step of the LNA prototype design consists of the circuit layout. The layout of RF circuits is critical, and should respect some important rules that improve its reliability. Some of these rules are listed below:

- The ground lines should be distributed along the circuit, and connections to ground should be the shortest possible. The distributed ground improves the current flow. Several ground pads should be used to improve current distribution through the circuit.
- All lines should be designed to support a current higher than expected.
- The use of long connecting lines should be avoided, and current paths inside the layout should be designed to minimize parasitic inductances and unwanted parasitic magnetic fields.
- A safety distance between transformer and other devices should be used to reduce magnetic coupling between devices.
- A ground shield must be used below the transformer to prevent eddy currents in the substrate, if the substrate has a low resistivity.
- All devices should have a guard ring to improve the substrate biasing and reduce noise.

The DLF LNA layout is designed to allow both on wafer and board testing. Next, some specific layout aspects of this particular LNA are listed:

- The input and output pads are designed with minimum area and using only metal layer 3 and the thick top metal layer 4, to minimize parasitic capacitances. The estimated value of the input pads capacitance to ground is 150 fF,
- to allow on wafer testing, the input and output pads should have a minimum distance to other pads of at least 200  $\mu$ m; thus, the layout is not optimized in terms of area.

The final DLF LNA layout is presented in figure 7.16, and has an area of 740  $\mu$ m × 890  $\mu$ m = 0.659 mm<sup>2</sup>.



Figure 7.16: Complete DLF LNA and buffer layout.

### 7.3.3 Post-Layout Simulations

In figures 7.17 to 7.20 the S-parameters of the DLF LNA plus buffer are represented. The DLF LNA is biased with 10 mA and the buffer with 0.5 mA. Two simulation results are plotted: original circuit simulation and post-layout simulation.

It is visible that the post-layout simulation of  $S_{11}$  is higher than the original simulation in the higher frequency region, due to layout parasitic capacitances. The remaining S-parameters are approximately unaltered by the layout parasitics.

In figure 7.21 the noise figure is represented. The simulated noise figure presented in figure 7.21 is slightly higher than in figure 7.9 where only the DLF LNA is considered. This means that the noise generated by the buffer is not negligible. Comparing the pre with the post-layout results, there is an increase in the noise figure and the frequency behavior is also altered, due to parasitic capacitances and resistances resulting from the layout.



Figure 7.17:  $S_{11}$  of the complete circuit (LNA and buffer): pre and post-layout simulations.



Figure 7.18:  $S_{12}$  of the complete circuit (LNA and buffer): pre and post-layout simulations.



Figure 7.19:  $S_{21}$  of the complete circuit (LNA and buffer): pre and post-layout simulations.



Figure 7.20:  $S_{22}$  of the complete circuit (DLF LNA and buffer): pre and post-layout simulations.



Figure 7.21: Noise figure of the complete circuit (DLF LNA and buffer): pre and post-layout simulations.

## 7.4 Test Board Design and Wire Bonding

To perform the DLF LNA test, a printed circuit board of FR-4<sup>2</sup> (PCB) laminate with a thickness of 0.8 mm was used with two metal (copper) layers. The input and output lines should have 50  $\Omega$  characteristic impedance, which leads to a width of 1.27 mm for these lines [127]. In figure 7.22 the board layout is represented in natural size.



Figure 7.22: Board 1 layout - natural size.

In figure 7.23 the circuit with its access points (pads) is represented, and in figure 7.24 the DLF LNA circuit is represented connected to the test setup and voltage supply. Variable resistances are used to obtain the reference currents for biasing.  $R_{DLF}$  controls the current in the DLF LNA and  $R_{BUFFER}$  controls the current in the buffer. All bonding wire inductances are represented.



Figure 7.23: Test circuit with pads.

 $<sup>^2{\</sup>rm The}$  "FR" means Flame Retardant (to UL94V-0), and Type "4" indicates woven glass reinforced epoxy resin [126].



Figure 7.24: Test setup.

The two variable resistors in the schematic of figure 7.24 are adjusted to the following values:

- $R_{\rm DLF} = 70 \ \Omega$
- $R_{\rm BUFFER} = 3500 \ \Omega$

which produce biasing currents of approximately 10 mA in the DLF LNA and 0.5 mA in the buffer.

The connection of the circuit die to the board is done by bonding wires, which have series inductance and resistance with values given in [128]. A gold bond wire was used with approximately 18  $\mu$ m diameter (approximately 0.8 mil). Figure 7.25 shows the photo of the test board. In table 7.4 the list of laboratory equipment used is presented.

Table 7.4: List of laboratory equipment.

Material:		
WILTRON Vector Network Analyzer (NA), model 360B		
HP Noise Figure Meter, model 8970B		
HP Noise Source, model 346A		
10  dB attenuator		
DC blocking coupler		



Figure 7.25: Test board with die.

In [129] the calibration process of the Vector Network Analyzer and the test procedure used are presented.

## 7.5 Experimental Results

In this section, the main experimental results are presented and discussed. First, the transformer experimental results are presented, and then, the LNA experimental results are given.

### 7.5.1 Transformer

An isolated transformer was included in the test chip and was tested individually by AMS. In figures 7.26 to 7.29 the (de-embedded) experimental and simulated Sparameters of the transformer are compared. In the simulated results, the model of figure 2.19 was used with the values listed in figure 7.15. There is a good agreement between the (de-embedded) experimental and the simulation results, meaning that the transformer parameters obtained using the electromagnetic simulator ASITIC are trustworthy. The patterned ground shield does not affect significantly the transformer performance.



Figure 7.26:  $S_{11}$  of the transformer - experimental and simulated results.



Figure 7.27:  $S_{12}$  of the transformer - experimental and simulated results.



Figure 7.28:  $S_{21}$  of the transformer - experimental and simulated results.



Figure 7.29:  $S_{22}$  of the transformer - experimental and simulated results.

### 7.5.2 LNA

Before testing the circuit, the maximum input power allowed at the LNA input was determined. The buffer has a current of 500  $\mu$ A for an output impedance of 50  $\Omega$ . This allows a maximum voltage swing of 500  $\mu$ A × 50  $\Omega$  = 25 mV. Considering a voltage gain of around 10, the maximum input peak-to-peak voltage is approximately 2.5 mV (or -39 dBm). The network analyzer power was then adjusted to -40 dBm. The DLF LNA is biased with a DC current of 10 mA.

Using the bias currents determined above, the circuit presents resonances in the  $S_{21}$  parameter. If the buffer bias current is varied to a higher value (approximately 40 mA), it is verified that the resonance disappears. One consequence of this increase was the degradation of the  $S_{22}$  parameter. Note that only the bias currents can be varied to debug the circuit.

To evaluate these results, a simulation of the test setup using the estimated of the bonding inductances was performed. The estimated bonding inductances are listed in table 7.5, and the bias currents and voltages measured at the test setup are listed in table 7.6. The experimental and simulated results considering the bonding inductances are represented in figures 7.30 to 7.33.

Inductance	Value	Inductance	Value
$L_{GND}$	1 nH	$L_{VDD1}$	1 nH
$L_{VDD}$	1  nH	$L_{DLF}$	3  nH
$L_{BUF}$	$3 \mathrm{nH}$	$L_{out}$	$3 \mathrm{nH}$
$L_{in}$	$3 \mathrm{nH}$		

Table 7.5: Estimate of the bond wire inductances.

Table 7.6: D	C voltages and	currents.
--------------	----------------	-----------

Currents			
Current	Value	Current	Value
$I_{BUF}$	38.60 mA	$I_{DLF}$	5.02  mA
Voltages			
Voltage	Value	Voltage	Value
$V_{BUF}$	$2.02 \mathrm{~V}$	$V_{DLF}$	$1.58 { m V}$
$V_{in}$	$0.85 \mathrm{V}$	V <sub>out</sub>	1.85 V

The measured  $S_{12}$  and  $S_{21}$  are in a good agreement with the simulation results. The measured and simulated results for  $S_{11}$  and  $S_{22}$  vary roughly in the same way with frequency, but there are significant differences of value. This is a consequence of the high sensitivity of these two S-parameters when the LNA input and output circuit impedances approach the 50  $\Omega$  characteristic impedance, since

$$S_{11} = 10 \log \left| \frac{Z_i - Z_0}{Z_i + Z_0} \right| \tag{7.5}$$

and

$$S_{22} = 10 \log \left| \frac{Z_L - Z_0}{Z_L + Z_0} \right| \tag{7.6}$$

where  $Z_i$  is the LNA input impedance,  $Z_L$  is the buffer output impedance and  $Z_0$  is the test equipment impedance. If  $Z_i$  (or  $Z_L$ ) and  $Z_0$  are almost equal,  $S_{11}$  (or  $S_{22}$ ) tend to  $-\infty$ .

In figure 7.34 the measured noise figure is represented and compared with the simulated results of the test setup considering the bonding inductances. Again, the strong influence of the bonding inductances on the noise figure is apparent.



Figure 7.30: Measured  $S_{11}$  and simulation considering bonding inductances.



Figure 7.31: Measured  $\mathcal{S}_{12}$  and simulation considering bonding inductances.



Figure 7.32: Measured  $S_{21}$  and simulation considering bonding inductances.



Figure 7.33: Measured  $S_{22}$  and simulation considering bonding inductances.



Figure 7.34: Measured noise figure and simulation considering bonding inductances.

#### 7.5.3 Discussion and improvements for a second prototype

Due to the difference between the expected DLF LNA performance and that measured, it is desirable to design a second prototype of the circuit.

The test results show a strong influence of the bonding inductances, mainly of the voltage supply (verified by simulation). To reduce the bonding inductance several pads can be used for the voltage supply; in addition, a large capacitance can be placed in parallel with the voltage supply.

In figure 7.35 the layout of the second prototype is represented. Each terminal of the voltage supply has 10 pads, and a large capacitance (1.4 nF) is used to further minimize the bond wires effect.

In figures 7.36 to 7.40 the simulated S-parameters and the noise figure are represented, considering two different situations. In one situation (solid lines), the bonding inductances  $L_{GND}$  and  $L_{VDD}$  have 1 nH, which is the inductance value considered in the test setup used for the first prototype. In the other situation (dashed lines), the bonding inductances  $L_{GND}$  and  $L_{VDD}$  have 0.2 nH, which corresponds to 10 bonding wires in parallel, each one having an inductance of 2 nH (this is a pessimistic situation, because it assumes that each bonding has a length of

#### 7.5. EXPERIMENTAL RESULTS



Figure 7.35: Layout of the second circuit prototype.

2 mm). It is visible that the effects of the bonding inductances are reduced in both curves of figures, the reduction being more significant when inductances  $L_{GND}$  and  $L_{VDD}$  are lower. If the bonding wire inductances are neglected, the S-parameters and noise figure are very similar to those in figures 7.17 to 7.21, obtained by simulation of the first prototype without the bonding inductances.



Figure 7.36:  $S_{11}$  considering  $L_{GND} = L_{VDD} = 1$  nH (solid line) and  $L_{GND} = L_{VDD} = 0.2$  nH (dashed line).



Figure 7.37:  $S_{12}$  considering  $L_{GND} = L_{VDD} = 1$  nH (solid line) and  $L_{GND} = L_{VDD} = 0.2$  nH (dashed line).



Figure 7.38:  $S_{21}$  considering  $L_{GND} = L_{VDD} = 1$  nH (solid line) and  $L_{GND} = L_{VDD} = 0.2$  nH (dashed line).



Figure 7.39:  $S_{22}$  considering  $L_{GND} = L_{VDD} = 1$  nH (solid line) and  $L_{GND} = L_{VDD} = 0.2$  nH (dashed line).



Figure 7.40: Noise figure considering  $L_{GND} = L_{VDD} = 1$  nH (solid line) and  $L_{GND} = L_{VDD} = 0.2$  nH (dashed line).

# 7.6 Conclusions

In this chapter, a prototype was designed and tested of the DLF LNA studied in chapter 6. The dimensioning of the different elements of the LNA and biasing circuits was explained.

The use of a single stage amplifying block is shown to be a good solution for the implementation of the DLF LNA. It was found that low output swing of the buffer limits the LNA input power.

The agreement between the simulated and measured results for the transformer confirms that the electromagnetic simulator ASITIC was a good choice for the design of the transformer.

The LNA test results are not in close agreement with the simulation results without bonding inductances, but the agreement is good when bonding inductances are considered in the simulations.

A second prototype was designed, in which the LNA circuit is identical to that in the first prototype, but the number of pads is increased with parallel wire bonding, to minimize the bonding inductances. A large capacitance was also placed in parallel with the voltage supply. The simulations show that this is an effective solution to minimize the negative effects of the bonding inductances.

Experimental results for the second prototype cannot be included in the thesis, since there was no time available for fabrication and testing. CHAPTER 7. LNA PROTOTYPE DESIGN AND EXPERIMENTAL RESULTS

# Chapter 8

# **Conclusions and Future Work**

- 8.1 Conclusions 181
- 8.2 **Future Work** 182

## 8.1 Conclusions

At the beginning of wireless communication, most receivers were narrowband. However, the demand for more flexible receivers has led to the development of multi-band and wideband receivers. In this thesis, multi-band and wideband low noise amplifiers (LNAs) were studied.

The first multi-band LNA studied is based on a cascode stage with inductive degeneration and has wideband input impedance matching that embraces the bands to be amplified. At the output there are different resonant branches with their cascode transistors, each one tuned to a different band. The output branches can be active simultaneously, or some can be switched off by controlling the gate voltage of the cascode transistors. A noise analysis of this LNA shows that it keeps the noise performance of the basic cascode LNA. It is also shown that it is possible to vary the voltage gain in one band without affecting significantly the other band.

The proposed multi-band LNA has a circuit similar to the previous one, but has two resonant output branches that are magnetically coupled. The output is taken at one of the branches being present two bands simultaneously; this is known as a concurrent multi-band LNA. By varying the gate bias voltage of one of the cascode transistors, while keeping the other constant, it is possible to vary the relative voltage gain in the two bands. The range of variation depends on the width ratio of the cascode transistors. It is shown that the noise study of the first multi-band LNA is also applicable to this concurrent LNA.

The suitability of double loop feedback (DLF) to the realization of wideband LNAs was investigated in chapter 3. It was concluded that to obtain input impedance matching independent of the output load, both feedback networks must sample the same output variable (voltage or current) and compare different variables (voltage and current) at the input. Using one transformer and one resistance to realize the two feedback loops, four different LNA topologies are obtained: two of them sampling the output current and two sampling the output voltage. All four topologies were studied in terms of input impedance, gain, and noise figure.

One of the DLF LNAs with output voltage sampling at the output was studied in more detail. The LNA performance was evaluated considering a non-ideal transformer and an ideal amplifying block (nullor). It was concluded that the transformer parasitic elements limit the LNA band of operation, but it is still possible to realize a wideband LNA. In a second part of the study, the transformer model is limited to two magnetically coupled inductors to simplify the analysis, and the amplifying block model includes its most important non-idealities. It is shown that a cascode stage should be used to minimize the effect of the drain-base capacitance of the input transistor. Equations for the minimum acceptable transconductance of this stage were developed. It was shown, using electrical simulations, that it is possible to realize a wideband DLF LNA with a reasonable power consumption.

A prototype of the above DLF LNA was designed. The transformer was designed using an electromagnetic simulator (ASITIC). Testing of the integrated transformer proved that the electromagnetic simulator is reliable. The experimental results of the DLF LNA were different from those obtained by simulation ignoring the bonding inductances; however, a good agreement was obtained with simulation results taking into account the bonding inductances, meaning that the circuit under test is very sensible to them.

In parallel with the work reported in this thesis, a method for the impedance evaluation of circular inductors was developed, but this was not included here, since it is somewhat outside the scope of the thesis.

# 8.2 Future Work

In the continuation of the research performed, the following further work could be done.

The concurrent multi-band LNA proposed in this thesis should be prototyped to evaluate its functionality. Two prototypes should be realized, with opposite magnetic coupling orientation. The design should follow the rules described in chapter 7 to overcome testing problems.

The DLF LNA prototype should be revised, preferably with the same technology, taking into account the bonding inductances.

The DLF LNA type 4, described in chapter 3.1, should be studied considering the transformer non-idealities to evaluate its performance limits. This study has already been done for the remaining topologies [24, 27]. The circuit presented in figure 4.3 can also be used to obtain dual mixing by making  $v_{BIAS1}$  and  $v_{BIAS2}$  the voltages of two local oscillators with different frequencies  $f_{L1}$  and  $f_{L2}$ , as represented in figure 8.1.



Figure 8.1: Proposed dual-band LNA - Mixer.

In this mode of operation the circuit of figure 8.1 can be viewed as a 2-quadrant variable transconductance multiplier, composed of a differential pair  $(M_{c1}, M_{c2})$  and a controlled current source  $M_1$ ; transistors  $M_{c1}$  and  $M_{c2}$  must remain in the active zone of operation.

If the input signal  $v_i$  has two frequency bands centered at  $f_{i1}$  and  $f_{i2}$ , the currents of  $M_{c1}$  and  $M_{c2}$  will contain 8 modulation products centered at frequencies  $f_{L1} \pm f_{i1}$ ,  $f_{L1} \pm f_{i2}$ ,  $f_{L2} \pm f_{i1}$ ,  $f_{L2} \pm f_{i2}$ . Two of these can be selected to appear at the outputs ( $v_{o1}$  and  $v_{o2}$ ) by tuning the output branches to the desired frequencies. This provides a great design flexibility, since the two frequencies are chosen from a set of eight. This circuit seems to have a potential that should be further explored.

# Appendix A

# **2-Port Matrices**

In table A.1, several 2-port matrix descriptions are presented. The voltages and currents are in accordance with figure A.1 [43].

Reference	Matrix description	Name
Y	$\begin{bmatrix} i_1 \\ i_2 \end{bmatrix} = \begin{pmatrix} y_{11} & y_{12} \\ y_{21} & y_{22} \end{pmatrix} \begin{bmatrix} v_1 \\ v_2 \end{bmatrix}$	Admittance Matrix
Z	$\begin{bmatrix} v_1 \\ v_2 \end{bmatrix} = \begin{pmatrix} z_{11} & z_{12} \\ z_{21} & z_{22} \end{pmatrix} \begin{bmatrix} i_1 \\ i_2 \end{bmatrix}$	Impedance Matrix
Н	$\begin{bmatrix} v_1 \\ i_2 \end{bmatrix} = \begin{pmatrix} h_{11} & h_{12} \\ h_{21} & h_{22} \end{pmatrix} \begin{bmatrix} i_1 \\ v_2 \end{bmatrix}$	Hybrid Matrix
H'	$\begin{bmatrix} i_1 \\ v_2 \end{bmatrix} = \begin{pmatrix} h'_{11} & h'_{12} \\ h'_{21} & h'_{22} \end{pmatrix} \begin{bmatrix} v_1 \\ i_2 \end{bmatrix}$	Dual Hybrid Matrix
T	$\begin{bmatrix} v_1 \\ i_1 \end{bmatrix} = \begin{pmatrix} t_{11} & t_{12} \\ t_{21} & t_{22} \end{pmatrix} \begin{bmatrix} v_2 \\ -i_2 \end{bmatrix}$	Chain Matrix

Table A.1: 2-port matrix descriptions.



Figure A.1: 2-port block.
### Appendix B

# Poles and Zeros of the DLF LNA considering Transformer Parasitic Capacitances and Resistances

Consider figure B.1 of the of the double loop feedback low noise amplifier (DLF LNA) that considers the transformer including the series resistances  $R_{wi}$  and interwinding capacitances  $C_{wi}$ .



Figure B.1: DLF LNA with transformer model including winding inductances  $L_{wi}$ , series resistances  $R_{wi}$  and inter-winding capacitances  $C_{wi}$ .

### **B.1** Input Impedance Analysis

The first step to determine  $Z_{in} = v_i/i_i$  consists on determining  $v_o/v_n$ :

$$v_o = v_n + \left(\frac{v_n}{sL_{w2}} + \frac{sC_{w1}}{n}v_o\right)R_{w2} \Leftrightarrow v_o = \frac{n(sL_{w2} + R_{w2})}{sL_{w2}(n + sC_{w1}R_{w2})}v_n$$
(B.1)

(B.1) is used to determine  $v_o/v_i$ ,

$$v_{i} = -\frac{v_{n}}{n} - (sL_{w1} + R_{w1})sC_{w1}v_{o} \Leftrightarrow$$

$$\Leftrightarrow v_{o} = -\frac{n^{2}(sL_{w2} + R_{w2})}{sL_{w2}(n + sC_{w1}R_{w2}) + n^{2}(sL_{w2} + R_{w2})(sL_{w1} + R_{w1})sC_{w1}}v_{i}$$
(B.2)

Finally,  $i_i$  is determined

$$i_i = sC_{w2}v_i + \frac{v_i - v_o}{R_f} - sC_{w1}v_o$$
(B.3)

The input impedance  $Z_{in}$  is determined using (B.2) and (B.3):

$$Z_{in} = \frac{v_i}{i_i} =$$

$$\frac{R_f[sL_{w2}(1 + n^{-1}sC_{w1}R_{w2}) + (sL_{w2} + R_{w2})(sL_{w1} + R_{w1})sC_{w1}]}{(1 + R_f sC_{w2})[sL_{w2}(1 + n^{-1}sC_{w1}R_{w2}) + n(sL_{w2} + R_{w2})(sL_{w1} + R_{w1})sC_{w1}] + n(1 + R_f sC_{w1})(sL_{w2} + R_{w2})}$$
(B.4)

To determine the poles and zeros of  $Z_{in}$ , it is assumed that different frequency ranges exist. At lower frequencies, it is assumed that the influence of  $C_{w1}$  and  $C_{w2}$  is not significant; i.e.  $C_{w1} = C_{w2} = 0$ , becoming  $Z_{in}$ 

$$Z_{in}(s) = \frac{v_i}{i_i}(s) = \frac{R_f s L_{w2}}{(n+1)sL_{w2} + nR_{w2}} = \frac{R_f L_{w2}}{nR_{w2}} \frac{s}{1 + \frac{s}{\omega_{p1,Zin}}}$$
(B.5)

(B.5) has one zero at the origin and one pole at

$$\omega_{p1,Zin} = \frac{nR_{w2}}{(n+1)L_{w2}} \tag{B.6}$$

At high frequencies the simplification assumed is  $sL_{w2} \gg R_{w2}$ 

$$Z_{in} = \frac{v_i}{i_i} = (B.7)$$

$$\frac{R_f [1 + n^{-1} s C_{w1} R_{w2} + (s L_{w1} + R_{w1}) s C_{w1}]}{(1 + R_f s C_{w2}) [1 + n^{-1} s C_{w1} R_{w2} + n(s L_{w1} + R_{w1}) s C_{w1}] + n(1 + R_f s C_{w1})}$$

 $\mathcal{C}_{w1}$  and  $\mathcal{C}_{w2}$  model the distributed capacitance between transformer windings 1 and

2. At the transformer model they are considered lumped and connected between the winding terminals. The total capacitance is usually distributed equally through  $C_{w1}$  and  $C_{w2}$  meaning that these capacitances can be assumed as having the same value. Considering  $C_{w2} = C_{w1}$ , (B.7) simplifies to

$$Z_{in} = \frac{v_i}{i_i} =$$

$$\frac{R_f}{n+1} \frac{[1+n^{-1}sC_{w1}R_{w2} + (sL_{w1} + R_{w1})sC_{w1}]}{(1+R_fsC_{w1})[1+n^{-1}sC_{w1}R_{w2} + n(sL_{w1} + R_{w1})sC_{w1}]}$$
(B.8)

(B.8), one single pole is directly determined and is equal to:

$$p_{2,Zin} = -\frac{1}{R_f C_{w1}} \tag{B.9}$$

The remaining poles are determined from the equality

$$1 + n^{-1}sC_{w1}R_{w2} + n(sL_{w1} + R_{w1})sC_{w1} = 0 \Leftrightarrow$$
(B.10)

$$\Leftrightarrow s^2 n^2 L_{w1} C_{w1} + s C_{w1} (n^2 R_{w1} + R_{w2}) + n = 0$$
(B.11)

Now, assuming that  $n^2 R_{w1} \gg R_{w2}$  and that a second order polynomial can be written in the following form,

$$s^2 + \frac{\omega_0}{Q}s + \omega_0^2 = 0 \tag{B.12}$$

where  $\omega_0$  is the pole frequency and Q is the quality factor. With this in mind,  $\omega_{p3,Zin}$  and  $Q_{p3,Zin}$  are determined:

$$\omega_{p3,Zin}^2 \approx \frac{1}{nL_{w1}C_{w1}} \tag{B.13}$$

$$Q_{p3,Zin} \approx \frac{1}{R_{w1}} \frac{1}{\sqrt{n}} \sqrt{\frac{L_{w1}}{C_{w1}}}$$
 (B.14)

Concerning the zero of (B.8) it is determined by next equality:

$$1 + n^{-1}sC_{w1}R_{w2} + (sL_{w1} + R_{w1})sC_{w1} = 0 \Leftrightarrow$$
(B.15)

$$\Leftrightarrow s^2 n L_{w1} C_{w1} + s C_{w1} (n R_{w1} + R_{w2}) + n = 0$$
(B.16)

Considering (B.12),  $\omega_{z2,Zin}$  and  $Q_{z2,Zin}$  are determined:

$$\omega_{z2,Zin}^2 \approx \frac{1}{L_{w1}C_{w1}} \tag{B.17}$$

$$Q_{z2,Zin} \approx \frac{n}{nR_{w1} + R_{w2}} \sqrt{\frac{L_{w1}}{C_{w1}}}$$
 (B.18)

#### **B.2** Voltage Gain Analysis

To determine the voltage gain  $v_o/v_s$ , it is considered the next equality:

$$i_i = \frac{v_S - v_i}{R_S} \tag{B.19}$$

(B.19) and (B.3) are used to obtain next equality:

$$\frac{v_S - v_i}{R_S} = sC_{w2}v_i + \frac{v_i - v_o}{R_f} - sC_{w1}v_o \Leftrightarrow$$
$$\Leftrightarrow \left(\frac{1}{R_f} + sC_{w1}\right)v_o - \left(\frac{1}{R_S} + \frac{1}{R_f} + sC_{w2}\right)v_i = -\frac{1}{R_S}v_S \tag{B.20}$$

 $A_v = v_o/v_S$  is determined by using (B.2) in (B.20). The representation of  $A_v$  does not give any inside over the circuit besides being a too large expression. Thus,  $A_v$ will be determine considering several frequency ranges. At lower frequencies, it is assumed that the interstage capacitances do not have effect on the voltage gain; i.e.,  $C_{w1} = C_{w2} = 0$ . It is also assumed that  $R_f = R_S(n+1)$  (input impedance matching condition). Under these conditions,  $A_v$  becomes:

$$A_{v}(s) = \frac{v_{o}}{v_{S}}(s) = -\frac{n(n+1)(sL_{w2} + R_{w2})}{2(n+1)sL_{w2} + nR_{w2}}$$
(B.21)

(B.21) has one single zero and one single pole located at

$$\omega_{p1,Av} = -\frac{1}{2} \frac{n}{n+1} \frac{R_{w2}}{L_{w2}} \tag{B.22}$$

$$\omega_{z1,Av} = -\frac{R_{w2}}{L_{w2}} \tag{B.23}$$

After  $p_{1,Av}$  and  $z_{1,Av}$ ,  $sL_{w2}$  becomes dominant over  $R_{w2}$ . Assuming again the input

2n

matching condition  $R_f = R_S(n+1), A_v$  becomes

$$A_{v} = \frac{v_{o}}{v_{s}} =$$

$$(B.24)$$

$$\frac{n^{2}(n+1)}{(2n^{2}C_{w1}L_{w1} + n^{3}C_{w1}L_{w1})s^{2} + (n^{2}C_{w1}R_{S} + n^{3}C_{w1}R_{S} + 2C_{w1}R_{w2} + nC_{w1}R_{w2})s + 2n^{2} + n^{2}C_{w1}R_{w2}}$$

(B.24) does not have zeros. To determine the pole consider next approximations:

• 
$$n^3 C_{w1} L_{w1} \gg 2n^2 C_{w1} L_{w1};$$

•  $n^3 C_{w1} R_S \gg n^2 C_{w1} R_S + 2C_{w1} R_{w2} + nC_{w1} R_{w2};$ 

• 
$$n^2 \gg n$$

= -

The above approximations allow (B.24) to be simplified further to

$$A_v = \frac{v_o}{v_s} = -\frac{(n+1)}{nC_{w1}L_{w1}s^2 + nC_{w1}R_Ss + 2}$$
(B.25)

Considering (B.12), (B.28) has a pair of complex poles at

$$\omega_{p2,Av}^2 \approx \frac{1}{nL_{w1}C_{w1}} \tag{B.26}$$

and having a quality factor of

$$Q_{p2,Av} \approx \frac{1}{R_S} \sqrt{\frac{2}{n}} \sqrt{\frac{L_{w1}}{C_{w1}}}$$
(B.27)

At frequencies above  $\omega_{p2,Av}$  next approximations can be made over equality (B.20):

- $sL_{w1} \gg R_{w1};$
- $sL_{w2} \gg R_{w2};$
- $sC_{w1}R_{w2} \gg n;$
- $sC_{w1} \gg 1/R_f;$
- $sC_{w2} \gg 1/R_f;$

 $A_v$  becomes

$$A_v = \frac{v_o}{v_s} = -\frac{n}{nC_{w1}^2 L_{w1} R_S s^3 + ((n+1)R_S + nL_{w1}s)C_{w1}s + 1}$$
(B.28)

Considering  $nL_{w1}s \gg (n+1)R_s$  and 1 negligible in comparison with the other denominator terms, (B.28) simplifies further to

$$A_v = \frac{v_o}{v_s} = -\frac{1}{C_{w1}^2 L_{w1} R_S s^3 + L_{w1} C_{w1} s^2}$$
(B.29)

(B.29) has a pole at

$$p_{3,Av} \approx -\frac{1}{R_S C_{w1}} \tag{B.30}$$

### Appendix C

### Source Transformations

In this appendix, some source transformation rules suitable for noise analysis are presented [23, 41, 47].

The Blakesley transformation or noise voltage source shifting is represented in figure C.1. The polarity of the new voltage sources must be respected as represented, to ensure the network equations are not changed.



Figure C.1: Voltage source shifting (Blakesley transformation).

A current source can be split into two current sources having the same value, and being connected to one of the terminals of the original current source - figure C.2. Their polarity must be as indicated.

The voltage source  $v_n$  in series with the resistance R has a Norton equivalent that is a current source  $i_n = v_n/R$  in parallel with resistance R as represented in figure C.3, and vice-versa.

In a 2-port, it is possible to transfer a source connected to one of the ports to the other port. Consider the 2-port of figure C.4.a) characterized by a chain matrix  $\mathbf{T}$ 



Figure C.2: Splitting of a current source.



Figure C.3: Norton-Thevenin sources equivalence.

and having a voltage source and a current source at output. The resulting sources at the input have the values represented in figure C.4.b). If the 2-port is a nullor, all terms of the chain matrix are zero, and this transfer is not applied - the input sources are zero.



Figure C.4: 2-port transformation of noise sources.

# Appendix D DLF LNA Input Noise Sources

In this appendix, the equivalent noise sources referred to the input of the DLF LNA noise sources are determined. They use the source transformations presented in appendix C.

#### D.1 Demonstration of (6.12)

In figure D.1, the different noise sources considered in the determination of (6.12) are represented. In this circuit, the amplifying block is considered an ideal voltage amplifying block with infinite voltage gain and input impedance and a zero output impedance. The amplifying block is noiseless.



Figure D.1: DLF LNA noise sources.

In figure D.2 the noise voltage source  $v_{n,Rw2}$  due to  $R_{w2}$  is transformed towards the

#### DLF LNA input.



Figure D.2: Transformation of the noise voltage source  $v_{n,Rw2}$ .

In step 1, the voltage source is moved towards the DLF LNA output and split into three sources, one in series with the output, one in series with the amplifying block output and one in series with  $R_f$ . The first two are canceled because they are in parallel with the zero output impedance of the ideal amplifying block, while the remaining is converted to a current source  $v_{n,Rw2}/R_f$  at  $R_f$ , by the Norton equivalence - step 2. This current source is split into one current source in parallel with the source resistance  $R_S$  and one current source in parallel with the zero output impedance of the amplifying block. The equivalent voltage noise source  $v_{n,i,Rw2}$  due to  $R_{w2}$  is then

$$v_{n,i,Rw2} = \frac{R_S}{R_f} v_{n,Rw2} \tag{D.1}$$

In figure D.3 the voltage noise source  $v_{n,Rf}$  due to  $R_f$  is moved towards input. It is firstly transformed to a current source by the Norton equivalence and then split into two current sources, one connected in parallel with the source resistance  $R_S$ and one connected to the amplifying block output. The second is canceled due to the ideality of the amplifying block (the output impedance is zero).



Figure D.3: Transformation of the noise voltage source  $v_{n,Rf}$ .

The equivalent voltage noise source  $v_{n,i,Rf}$  due to  $R_f$  is

$$v_{n,i,Rf} = \frac{R_S}{R_f} v_{n,Rf} \tag{D.2}$$

In figure D.4 the voltage noise source  $v_{n,Rw1}$  due to  $R_w1$  is moved towards input. The initial noise voltage source is split into two voltage sources, one connected in series with the source resistance  $R_S$  and one in series with the  $R_f$ . The second is transformed to the input like that in the last example.

The equivalent voltage noise source  $v_{n,i,Rw1}$  due to  $R_w1$  is

$$v_{n,i,Rw1} = \left(1 + \frac{R_S}{R_f}\right) v_{n,Rw1} \tag{D.3}$$

The total input noise voltage is the sum of (D.1) to (D.3), resulting:

$$v_{n,i} = v_{n,i,Rw1} + v_{n,i,Rw2} + v_{n,i,Rf} =$$

$$= \frac{R_S}{R_f} (v_{n,Rw1} + v_{n,Rw2} + v_{n,Rf}) + v_{n,Rw1}$$
(D.4)

Assuming  $R_f = (n + 1)R_S$ , and using the Wiener-Khintchine theorem (2.29), the



Figure D.4: Transformation of the noise voltage source  $v_{n,Rw1}$ .

DLF LNA equivalent input spectral noise density becomes:

$$N_i = 4k_B T [R_f + R_{w2} + R_{w1}] \left(\frac{1}{n+1}\right)^2 + 4k_B T R_{w1}$$
(D.5)

### D.2 Demonstration of (6.44)

In figure D.5, the different noise sources considered in the determination of (6.12) are represented. In this circuit, the amplifying block is considered an ideal voltage amplifying block with infinite voltage gain and input impedance and a zero output impedance. The amplifying block has two noise sources at its input:  $v_{n,A}$  and  $i_{n,A}$ .



Figure D.5: DLF LNA noise sources.

The equivalent noise voltage and current sources due to  $v_{n,A}$  are similar to that

determined to  $v_{n,Rw1}$ :

$$v_{n,i,A}' = v_{n,A} \tag{D.6}$$

$$i'_{n,i,A} = \frac{1}{R_f} v_{n,A}$$
 (D.7)

The equivalent noise current source due to  $v_{n,Rf}$  is equal to that determined in section D.1 and is

$$i_{n,i,Rf} = \frac{1}{R_f} v_{n,Rf} \tag{D.8}$$

Finally, in figure D.6 are represented the source transformations to determine the equivalent noise sources referred to the input and due to  $i_{n,A}$ .



Figure D.6: Transformation of the noise voltage source  $v_{n,Rw1}$ .

The current source  $i_{n,A}$  is firstly split into two current sources, one in parallel with winding 1 and the other connected between the winding and  $L_{w1}$ . The current source in parallel with winding 1 is translated to the output via the transformer current gain; and there, the current source is killed by the zero output impedance of the amplifying block. The second current source is again split into one current source in parallel with the series of  $L_{w1}$  and  $R_{w1}$  and one current source in parallel with the source resistance  $R_S$ . The current source ... is converted to a voltage source  $i_{n,A}Z_{w1}$  and from now on, its treatment is similar to that of  $v_{n,Rw1}$ . The equivalent noise voltage and current sources due to  $i_{n,A}$  are:

$$v_{n,i,A}'' = Z_{w1}i_{n,A}$$
 (D.9)

$$i_{n,i,A}^{\prime\prime} = \left(1 + \frac{Z_{w1}}{R_f}\right) i_{n,Amp} \tag{D.10}$$

 $Z_{w1}$  is the impedance due to the series of  $L_{w1}$  and  $R_{w1}$ . The equivalent input noise voltage is:

$$v'_{n,i} = v_{n,i} + i_{n,i}R_S = \frac{1}{R_f}(R_f + R_S)(v_{n,A} + v_{n,R_{w1}}) + \left(\frac{Z_{w1}}{R_f}(R_f + R_S) + R_S\right)i_{n,A} + \frac{R_S}{R_f}v_{n,R_f}$$
(D.11)

where  $v_{n,i}$  and  $i_{n,i}$  are

$$v_{n,i} = v'_{n,i,A} + v''_{n,i,A} \tag{D.12}$$

$$v_{n,i} = i'_{n,i,A} + i''_{n,i,A} + i_{n,i,R_f}$$
(D.13)

### Appendix E

## Technology File for ASITIC Program

The technology file used in the ASITIC program to design the inductors used along the work is listed below. Figure E serves to a better understanding of the technology file.

#### Technology File:

```
<chip>
       chipx = 512; dimensions of the chip in x direction in microns
       chipy = 512
                     ; dimensions of the chip in y direction
       fftx = 1024 ; x-fft size (must be a power of 2)
       ffty = 1024
                       ; y-fft size
       TechFile = tech_sige.tek ; the name of this file
       TechPath = /home/mam/ ; the pathname of the data files
       freq = .1
<layer> 0
             ; Bulk Substrate
       rho = 19
                               ; Resistivity:
                                               ohm-cm
       t = 700
                               ; Thickness: microns
       eps = 11.9
                               ; Permitivity: relative
<layer> 1
             ; Oxide Layer
       rho = 1e10
                               ; ohm-cm
       t = 10
                                ; microns
```

```
eps = 4
                             ; relative
<metal> 0 ; metal layer 1
       layer = 1
       rsh = 70
       t = 0.665
       d = 0.927
       name = m1
       color = red
<metal> 1 ; metal 2
       layer = 1
       rsh = 70
       t = 0.640
       d = 2.592
       name = m2
       color = blue
<metal> 2 ; metal 3
       layer = 1
       rsh = 70
       t = 0.640
       d = 4.232
       name = m3
       color = green
<metal> 3 ; metal 4
       layer = 1
       rsh = 10
       t = 2.800
```

d = 5.872
name = m4
color = yellow



Figure E.1: Figure containing the different layers of a technology process.

### Appendix F

# Impedance Evaluation of Integrated Circular Spiral Inductors

The work presented in this appendix was realized in collaboration with professors Vitor Maló Machado and Luís Bica de Oliveira, during the course of *Simulation Tools for Electromagnetics*, and resulted in a publication [30]. It is presented in appendix because it is more related with element analysis, while the thesis is more dedicated to circuit analysis; however, it gives a good inside on problems related with inductance determination.

### F.1 Introduction

An inductor is a "circuit element which is a conductor wrapped into a coil to create a magnetic field" [130]. "The voltage across an inductor is directly proportional to the rate of change of the current through it divided by the rate of change of time (difference current/difference time  $= \frac{di}{dt}$ ). The proportionality constant which makes this true is L, the inductance of the inductor component. It is denoted by L and its units are the Henry (H). Therefore, the voltage v across an inductor is given by

$$v = L \frac{di}{dt} \tag{F.1}$$

"[130].

The aim of this work is the determination of the inductance of a spiral integrated inductor. To obtain inductance value L of equation (F.1) it is necessary to describe

the electromagnetic fields, define the material conditions, the boundaries, etc... The electromagnetic field is described by *Maxwell Equations*. Their local form are showed below:

$$\nabla \times H = J + \frac{\partial D}{\partial t} \tag{F.2}$$

$$\nabla \times E = -\frac{\partial B}{\partial t} \tag{F.3}$$

$$\nabla \cdot D = \rho \tag{F.4}$$

$$\nabla \cdot B = 0 \tag{F.5}$$

*B* is the magnetic field, *E* is the electric field, *H* is the magnetic field strength, *D* is the electric displacement and  $\rho$  is the charge density. Equation (F.2) with the omission of the electric displacement density from the second term,  $\frac{dD}{dt}$  represents the local form of the *Ampere* equation: a current density *J* that generates a magnetic field *H*, i.e.  $\nabla \times H = J$ . This formulation is only valid in quasi-static situations, i.e. when the size of the physical elements is much smaller than the minimum wave length of interest. The inclusion of the second term was one great *Maxwell* contribution which allowed the study of electromagnetic waves. Equation (F.3) is the local form of the *Faraday* induction law. A temporal variation of the magnetic field generates an electric field. Equation (F.4) represents the local form of the total charge enclosed within the surface. Equation (F.5) is the local form of *Gauss* law for magnetism and means that the net magnetic flux out of any closed surface is zero. This amounts to a statement about the sources of magnetic field: it is not possible to store magnetic charges.

The electric field E can be deduced from two other fields, the vector potential A and the electric potential  $\phi$  by means of the next relation.

$$E = -\frac{\partial A}{\partial t} - \nabla\phi \tag{F.6}$$

Inside a linear conductor, the current density J can be related with the electric field  ${\cal E}$  by

$$J = \sigma E \tag{F.7}$$

 $\sigma$  represents the conductor conductivity. Applying equation (F.6) into equation (F.9) it is possible to relate the two fields introduced with J as

$$J = -\sigma \frac{\partial A}{\partial t} - \sigma \nabla \phi \tag{F.8}$$

### F.2 Description of the problem

In this work an algorithm to determine the inductance of an integrated circular spiral inductor is developed. A top view of the inductor is presented in figure F.1. The inductance will be determined by the moment method. This method consists basically in dividing each ring in several finite elements and then determining the magnetic flux caused by each element over the others. The final result is a square matrix with the contribution of one element over all the others. In figure F.2 is shown a vertical projection of the inductor. As a first approach, the inductance will be determined by approximating the spiral inductor to metal concentric rings as in figure F.3.



Figure F.1: Spiral square inductor.

In each turn of the inductor, the vector current density is

$$\vec{J} = J\vec{u}_{\varphi} \tag{F.9}$$



Figure F.2: Vertical plane projection of the concentric inductor.



Figure F.3: Concentric ring inductor.

and the vector potential is

$$\vec{A} = A\vec{u}_{\varphi} \tag{F.10}$$

With a constant frequency stimulation, vector J is equal to

$$J = -j\omega\sigma A + J_s \tag{F.11}$$

 $\omega$  is equal to  $2\pi f$ , where f is the frequency.  $\sigma$  is the conductivity of the metal and Js is a current inside the conductor due to the source.

From equations (F.8) and (F.11), using the fact that the current has zero divergence inside the conductor and that  $\nabla A = 0$ , the electrical potential satisfies Laplace's equation inside conductor:  $\nabla^2 \phi = 0$ . With this approximation, it is neglected the inductor capacitance. Therefore, this work is only suitable for determining inductance L and resistance R. To determine the magnetic flux  $\Psi$  related caused by all the elements it is necessary to use a specific Green function to solve the flux integral. The Green function G for concentric wires is

$$Green = \mu G = \frac{1}{2\pi} \int_0^\pi \frac{\rho' \cos\varphi}{\sqrt{\rho'^2 + \rho^2 - 2\rho'\rho\cos\varphi + (Z - Z_0)^2}} d\varphi$$
(F.12)

The  $(Z - Z_0)^2$  component of the Green function above might be neglected in order to simplify the calculation.

The potential vector A is related with the current density J by

$$A = \int_{S} J\mu G dS \tag{F.13}$$

The magnetic flux  $\Psi$  is related with the potential vector A by equation

$$\Psi = 2\pi\rho A \tag{F.14}$$

Gathering equations (F.13) and (F.14) the magnetic flux  $\Psi$  relates with the current density J by

$$\Psi = \int_{S} J\mu(2\pi\rho G)dS \tag{F.15}$$

Relating equations (F.11), (F.14) and (F.15) it is obtained an equation where the only unknown variables are the magnetic flux  $\Psi$  and  $J_s$ 

$$\Psi + j\omega\mu\sigma \int_{S} \Psi \frac{2\pi\rho G}{2\pi\rho} dS - \int_{S} J_s 2\pi\rho\mu G dS = 0$$
(F.16)

The expression (F.16) is a complete description of our configuration, where all inductor rings are included.

$$\sum_{n} -j\omega\mu\sigma \int_{\Delta S_n} \frac{\Psi}{2\pi\rho} \frac{1}{\mu} dS + \sum_{n} \int_{\Delta S_n} J_s dS = I$$
(F.17)

The next step is to divide the global equation (F.16) in smaller equations. In order to do that there are defined N equations that represent the N inductor rings, equation (F.17). Each inductor ring is also divided into smaller elements. These elements determine the accuracy of the calculation: few elements leads to less accuracy, while many elements increase the calculation time. In all those rings it is considered a constant current with value I. Equation (F.17) can not be computed because  $\Psi$  is continuous and unknown. In order to determine it,  $\Psi$  is considered constant inside each element, there they assume the value  $\Psi_n$ . This approximation is another source of errors.

$$\sum_{n} -j\left(\frac{\sqrt{2}}{\delta}\right)^{2} \frac{\Psi_{n}}{2\pi\mu} \int_{\Delta S_{n}} \frac{1}{\rho} dS + \sum_{n} \frac{K_{s}}{2\pi} \int_{\Delta S_{n}} \frac{1}{\rho} dS = I$$
(F.18)

Equation (F.18) allows the determination of the flux in a matrix form. To avoid, or reduce, numerical errors during the resolution of Gauss methods, the different sub-matrixes are normalized. Next equations are used for normalization.

$$\rho_{norm} = \frac{\rho}{\alpha}, \alpha = 10^{-6} \tag{F.19}$$

$$\Psi_{norm} = \frac{\Psi}{2\Pi\alpha\mu_0 I} \tag{F.20}$$

$$K_{snorm} = \frac{\alpha K_s}{2\Pi i} \tag{F.21}$$

$$p_{norm} = p\alpha, p = \sqrt{\omega\mu\sigma} \tag{F.22}$$

The matrix form is shown in equation (F.23). Variables are  $\Psi_n$  of each element and  $K_s$ , both in the normalized form.

$$\begin{bmatrix} L & H \\ \hline Q & S \end{bmatrix} \begin{pmatrix} \Psi_{norm} \\ \hline K_{snorm} \end{pmatrix} = \begin{pmatrix} 0 \\ \hline 1 \end{pmatrix}$$
(F.23)

where the sub-matrix elements are:

$$L_{i,j} = jp_{norm}^2 \int_{\Delta S_{jn}} g(\rho = \rho_i, \rho' = \rho_j) dS, i \neq j$$
(F.24)

$$L_{i,i} = 1 + j p_{norm}^2 \int_{\Delta S_{in}} g(\rho = \rho_i, \rho') dS$$
(F.25)

$$H_{i,k} = -\sum_{m_k} \int_{\Delta S_{m_k n}} g(\rho = \rho_i, \rho') dS$$
(F.26)

$$Q_{k,j} = -jp_{norm}^2 \int_{\Delta S_{jn}} \frac{1}{\rho_{norm}} dS, ifj \in ring_k$$
(F.27)

$$Q_{k,j} = 0, ifj \notin ring_k \tag{F.28}$$

$$S_{k,k} = \int_{S_{kn}} \frac{1}{\rho_{norm}} dS \tag{F.29}$$

After solving equation (F.23), inductance and resistance are determined respectively with next expressions:

$$L = \frac{1}{2} \sum_{n} \left( \frac{1}{2} Re(K_s^* \Psi_n) \frac{1}{2\Pi} \int_{S_n} \frac{1}{\rho_n} dS \right)$$
(F.30)

$$R = \frac{1}{2} \sum_{n} \left( \frac{1}{2\Pi} \frac{1}{2\sigma} \left( -j\omega\sigma\Psi_n + K_s \right)^2 \int_{S_n} \frac{1}{\rho_n} dS \right)$$
(F.31)

#### F.3 Results

To evaluate the algorithm exposed above some known results are used. From [131], results listed in table F.1 have been used as reference. Results obtained with the presented algorithm are shown on figures F.4 to F.9. Inductance graphics, figures F.4, F.6 and F.8 show a constant value for low frequencies and then a sharp decrease. Quality factor graphics, figures F.5, F.7 and F.9 have a peak at medium frequencies. These behaviors are in accordance with the results of the work used as reference, however they have numerical errors.

Table F.1: Geometrical parameters (N - number of turns, D - diameter, w - width, s - section), inductance at maximum quality factor, L, and maximum quality factor, Q, of three inductors in medium-resistivity factor.

Inductor	N	$D[\mu m]$	$w[\mu m]$	$s[\mu m]$	$L[\mathrm{nH}]$	Q
А	4	240	13.7	10.27	1.93	6.26
В	4	280	11.38	3	5.17	5.90
$\mathbf{C}$	6	320	11.25	3	10.69	4.83



Figure F.4: Inductance A vs frequency.



Figure F.5: Quality factor of inductor A.



Figure F.6: Inductance B vs frequency.



Figure F.7: Quality factor of inductor B.



Figure F.8: Inductance C vs frequency.



Figure F.9: Quality factor of inductor C.

### Bibliography

- P.-In. Mak, S.-P. U and R. P. Martins, "Transceiver Architecture Selection: Review, State-of-the-Art Survey and Case Study", *IEEE Circuits and Systems Magazine*, vol. 7, no. 2, pp. 6-25, March 2007.
- [2] T. H. Lee, The Design of CMOS Radio-Frequency Integrated Circuits, Cambridge University Press, 2004.
- [3] B. Razavi, *RF Microelectronics*, Prentice Hall Ptr, 1998.
- [4] B. Matinpour, J. Laskar and S. Chakraborty, Modern Receiver Front-Ends -Systems, Circuits and Integration, John Wiley and Sons, Inc., 2004.
- [5] J. Crols and M. Steyaert, CMOS Wireless Transceiver Design, Kluwer Academic Publishers, 1997.
- [6] H. Hashemi and A. Hajimiri, "Concurrent Dual-Band CMOS Low Noise Amplifiers and Receiver Architectures", *IEEE Transactions on Microwave Theory* and Tecniques, vol. 50, no. 1, pp. 288-331, January 2002.
- [7] J. van der Tang D. Leenaerts and C. Vaucher, Circuit Design for RF Transceivers, Kluwer Academic Publishers, Dordrecht, 2001.
- [8] T. Lee, *Planar Microwave Engineering A Pratical Guide to Theory, Measurements and Circuits*, Cambridge University Press, 2004.
- [9] R. Ludwig and P. Bretchko, *RF Circuit Design Theory and Applications*, Prentice Hall, 2000.
- [10] L. B. Michael M. Ghavami and R. Kohno, Ultra Wideband Signals and Systems in Communication Engineering, John Wiley and Sons, Inc., 2004.
- [11] C.-W. Kim, M.-S. Tang, P. T. Anh, H.-T. Kim and S.-G. Lee, "An Ultra-Wideband CMOS Low Noise Amplifier for 3-5 GHz UWB System", *IEEE J. Solid-State Circuits*, vol. 40, no. 2, pp. 544-547, February 2005.

- [12] J. C. Rudell, Jia-Jiunn Ou, T. B. Cho, G. Chien, F. Brianti and J. A. Weldon, "A 1.9-GHz Wide-Band IF Double Conversion CMOS Receiver for Cordless Telephone Applications", *IEEE J. Solid-State Circuits*, vol. 32, no. 12, pp. 2071-2088, December 1997.
- [13] C.-S. Wang, W.-C. Li and C.-K. Wang, "A Multi-Band Multi-Standard RF Front-End IEEE 802.16a for IEEE 802.16a and IEEE 802.11 a/b/g Applications", Proc. IEEE International Symposium on Circuits and Systems -ISCAS 2005, vol. 4, pp. 3974-3977, May 2005.
- [14] J. Ko, J. Kim, S. Cho and K. Lee, "A 19-mW 2.6-mm/sup 2/ L1/L2 Dual-Band CMOS GPS receiver", *IEEE J. Solid-State Circuits*, vol. 40, no. 7, pp. 1414-1425, July 2005.
- [15] N. Darbanian, S. Farahani, S. Kiaei, B. Bakkaloglu and M. Smith, "Tri-Mode Integrated Receiver for GPS, GSM 1800, and WCDMA", Proc. Radio Frequency Integrated Circuits Symposium, RFIC-2006, June 2006.
- [16] H. Hashemi and A. Hajimiri, "Concurrent Dual-Band CMOS Low Noise Amplifiers and Receiver Architectures", *Radio Frequency Integrated Circuits Symposium*, *RFIC-2001*, pp. 247-250, June 2001.
- [17] Z. Li, R. Quintal and K. O. Kenneth, "A Dual-Band CMOS Front-End with Two Gain Modes for Wireless LAN Applications", *IEEE J. Solid-State Circuits*, vol. 39, no. 11, pp. 2069-2073, November 2004.
- [18] "P. Heydari", "Design Considerations for Low-Power Ultra-Wideband Receivers", Proc. Sixth International Symposium on Quality of Electronic Design, pp. 668-673, March 2005.
- [19] D. M. W. Leenaerts. "Transceiver Design for Multiband OFDM UWB". J. on Wireless Communications and Networking - EURASIP, 2006.
- [20] J. Ryynanen, K. Kivekas and J. Jussila, "A Dual-Band RF Front-End for WCDMA and GSM Applications", *IEEE J. Solid-State Circuits*, vol. 36, no. 8, pp. 1198-1204, August 2001.
- [21] M.-Yi Wang, R. R.-B. Sheen, O. T. -C. Chen and R. Y. J. Tsen, "A Dual-Band RF front-end for WCDMA and GPS applications", *Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2005*, vol. 4, pp. 113-116, May 2002.

- [22] R. Bagheri, A. Mirzaei, S. Chehrazi, M. E. Heiradi, M. Lee, M. Mikheimai, W. T. and A. A. Abidi, "A 800-MHz-6-GHz Software-Defined Wireless Receiver in 90-nm CMOS", *IEEE J. Solid-State Circuits*, vol. 41, no. 12, pp. 2860-2876, December 2006.
- [23] E. H. Nordholt, "Classes and Properties of Multiloop Negative-Feedback Amplifiers", *IEEE Trans. on Circuits and Sistems*, vol. 28, no.3, pp. 203-211, March 1981.
- [24] K. van Hartingsveldt, M.H.L Kouwenhoven and C.J.M. Verhoeven, "HF Low Noise Amplifiers with Integrated Transformer Feedback", *Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2002*, vol. 2, pp. II-815
   - II-818, May 2002.
- [25] M. A. Martins, J. R. Fernandes and M. M. Silva, "Techniques for Dual-Band LNA Design using Cascode Switching and Inductor Magnetic Coupling", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2007, vol. I, pp. 1449-1452, May 2007.
- [26] M. A. Martins, J. R. Fernandes and M. M. Silva, "Dual-Band CMOS Low Noise Amplifier without Switches and with Continuously Adjustable Gain", *IEE Electronics Letters*, vol. 43, no.17, August 2007.
- [27] M. A. Martins, K. van Hartingsveldt, C. J. M. Verhoeven and J. R. Fernandes, "Wide-Band Low Noise Amplifier with Double Loop Feedback", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2005, vol. VI, pp. 5353-5356, May 2005.
- [28] M. A. Martins, K. van Hartingsveldt, C. J. M. Verhoeven and J. R. Fernandes, "Structured Design of a Wide-Band Low Noise Amplifier", Proc. XX Conference on Design of Circuits and Integrated Systems - DCIS'05, vol. I, November 2005.
- [29] M. A. Martins, J. R. Fernandes, M. M. Silva and C. J. M. Verhoeven, "On the Design Procedure for a Wideband LNA with Double Loop Feedback", Proc. XXII Conference on Design of Circuits and Integrated Systems - DCIS'07, November 2007.
- [30] M. A. Martins, L. B. Oliveira and V. M. Machado, "Impedance Evaluation of Integrated Circular Spiral Inductors", Proc. 6th International Conference on Computation in Electromagnetics - CEM'06, pp. 67-68, April 2006.

- [31] M. de A. Faro. Propagação e Radiação de Ondas Electromagnéticas Radiação. TECNICA-AEIST, 1980.
- [32] B. Razavi, et al, "A UWB CMOS Transceiver", IEEE J. Solid-State Circuits, vol. 40, no. 12, pp. 2555-2562, December 2005.
- [33] P. Heydari, "A Study of Low-Power Ultra Wideband Radio Transceiver Architectures", Proc. IEEE Wireless Communications and Networking Conference, 2005, vol. 2, pp. 758-763, March 2005.
- [34] D. Porcino and W. Hirt, "Ultra-Wideband Radio Technology: Potential and Chalanges Ahead", *IEEE Communications Magazine*, vol. 41, no. 7, pp. 66-74, July 2003.
- [35] T. K. K. Tsang and M. N. El-Gamal, "Ultra-Wideband (UWB) Communications Systems: an Overview", Proc. 3rd IEEE International Northwest Workshop on Circuits and Systems - NEWCAS 2005, pp. 381-386, June 2005.
- [36] D. K. Shaeffer. The Design and Implementation of Low-Power CMOS Radio Receivers. PhD thesis, Standford University, 1998.
- [37] J. Smith. Modern Communication Circuits. McGraw-Hill, 1986.
- [38] J. R. Long, "Next Generation Narrowband RF front-ends in silicon IC technology", Proc. 8th Great Lakes Symposium on VLSI, 1998, pp. 275-280, February 1998.
- [39] P. P. Newaskar, R. Blazquez and A. R. Chandrakasan, "A/D Precision Requirements for an Ultra-Wideband Radio Receiver", Proc. IEEE Workshop on Signal Processing Systems, 2002, pp. 270-275, October 2002.
- [40] G. R. Aiello, "Challenges for Ultra-Wideband (UWB) CMOS Integration", Proc. Sixth International Symposium on Quality of Electronic Design, 2005, pp. 668-673, March 2005.
- [41] G. L. E. Monna M. H. L. Kouwenhoven C. J. M. Verhoeven, A. van Staveren and E. Yildiz. Structured Electronic Design - Negative-Feedback Amplifiers. Kluwer Academic Publishers, , 2003.
- [42] M. de M. Silva, Circuitos com Transistores Bipolares e MOS, Gulbenkian, 2003.

- [43] M. de M. Silva, Introdução aos Circuitos Eléctricos e Electrónicos, Gulbenkian, 2001.
- [44] T. M. Apostol, Calculus One-Variable Calculus, with an Introduction to Linear Algebra, volume 1, 2nd edition, John Wiley and Sons, Inc., New York, 1967.
- [45] P. Wambacq and W. Sansen, Distortion Analysis of Analog Integrated Circuits, Kluwer Academic Publishers, Boston, 1998.
- [46] G. Palumbo and S. Pennisi, "High-Frequency Harmonic Distortion in Feedback Amplifiers: Analysis and Applications", *IEEE Trans. on Circuits and Systems - I: Fundamental Theory and Applications, TCAS-I*, vol. 50, no. 3, pp. 328-340, March 2003.
- [47] A. van der Ziel, Noise in Solid State Devices and Circuits, John Wiley and Sons, 1986.
- [48] J.-S. Goo, High Frequency Noise in CMOS Low Noise Amplifiers, PhD thesis, Standford University, 2001.
- [49] A. B. Carlson, COMMUNICATION SYSTEMS An Introduction to Signals and Noise in Electrical Communication, McGraw-Hill, 1986.
- [50] H. T. Friis, "Noise Figure of Radio Receivers", Proc. IRE, vol. 32, pp. 419-422, July 1944.
- [51] J. Randa, "Noise characterization of multiport amplifiers", IEEE Trans. on Microwave Theory and Techniques, October 2001.
- [52] W. Namgoong and J. Lerdworatawee, "Noise Figure of Digital Communication Receivers-Revisited", *IEEE Trans. on Circuits and Systems - I: Fundamental Theory and Applications, TCAS-I*, vol. 51, no. 7, pp. 1330-1335, July 2004.
- [53] A. J. Sholten, L. F. Tiemeijer, R. van Langevelde, R. J. Havens, A. T. A. Zegers van Duijnhoven and V. C. Venezia, "Noise Modeling for RF CMOS Circuit Simulation", *IEEE Trans. on Electron Devices*, vol. 50, no. 3, pp. 618-632, March 2003.
- [54] P. Lafrance, Fundamental Concepts in Communication, Prentice-Hall International, Inc., 1990.

- [55] K. K. Hung, P. K Ko, C. Hu and Y. C. Cheng, "A Unified Model for the Flicker Noise in Metal-Oxide Semiconductor Field - Effect Transistors", *IEEE Trans. on Electron Devices*, vol. 37, no.3, March 1990.
- [56] D. K. Shaeffer and T. H. Lee, "A 1.5-V, 1.5-GHz CMOS Low Noise Amplifier", *IEEE J. Solid-State Circuits*, vol. 32, no. 5, pp. 745-759, May 1997.
- [57] G. Niu, J. D. Cressler, S. Zhang, W. E. Ansley, C. S. Webster and D. L. Harame, "A Unified Approach to RF and Microwave Noise Parameter Modeling in Bipolar Transistors", *IEEE Trans. on Electron Devices*, vol. 48, no. 11, pp. 2568-2574, November 2001.
- [58] J. F. B. da Silva, *Electrotécnia Teórica Parte I*, Centro de Documentação AEIST, 1995.
- [59] J. N. Burghartz, D. C. Edelstein, M. Soyuer, H. A. Ainspan and K. A. Jenkins, "RF Circuit Design Aspects of Spiral Inductors on Silicon", *IEEE J. Solid-State Circuits*, vol. 33, no. 12, pp. 2028-2034, December 1998.
- [60] J. R. Long and M. A. Copeland, "The Modeling, Characterization, and Design of Monolithic Inductors for Silicon RF IC's", *IEEE J. Solid-State Circuits*, vol. 32, no. 3, pp. 357-369, March 1997.
- [61] H. M. Greenhouse, "Design of Planar Rectangular Microelectronic Inductors", *IEEE Trans. on Parts, Hybrids and Packaging*, vol. 10, no. 2, pp. 101-109, June 1974.
- [62] S. S. Mohan, The Design, Modeling and Optimization of On-Chip Inductors and Transformer Circuits, PhD thesis, Standford University, 1999.
- [63] Yu Cao, R. A. Groves, X. Huang, N. D. Zamdmer, J.-O. Plouchart, R. A. Wachnik, Tsu-Jae King and C. Hu, "Frequency-Independent Equivalent-Circuit Model for On-Chip Spiral Inductors", *IEEE J. Solid-State Circuits*, vol. 38, no. 3, pp. 419-426, March 2003.
- [64] C. P. Yue, C. Ryu, J. Lau, T. H Lee and S. S. Wong, "A Physical Model for Planar Spiral Inductors On Silicon", Proc. International Electron Devices Meeting - IEDM 2000, December 1996.
- [65] C. P. Yue and S. S. Wong, "On-Chip Spiral Inductors with Patterned Ground Shield for Si-Based RF IC's", *IEEE J. Solid-State Circuits*, vol. 33, no.5, pp. 743-752, May 1998.

- [66] S. S. Mohan, M. del Mar Hershenson, S. P. Boyd and T. H. Lee, "Simple Accurate Expressions for Planar Spiral Inductances", *IEEE J. Solid-State Circuits*, vol. 34, no. 10, pp. 1419-1424, October 1999.
- [67] A. M. Niknejad, "Modeling of Passive Elements with ASITIC", Proc. 2002 IEEE MTT-S International Microwave Symposium Digest, vol. 1, pp. 149-152, June 2002.
- [68] S. S. Mohan, C.P. Yue, M. del Mar Hershenson, S. S. Wong and T. H. Lee, "Modeling and Characterization of On-Chip Transformers", *Proc. Interna*tional Electron Devices Meeting - IEDM'98, pp. 531-534, December 1998.
- [69] R. A. DeCarlo and P.-M. Lin, Linear Circuit Analysis Time Domain. Phasor, and Laplace Transform Approaches, 2nd edition, Oxford University Press, 2001.
- [70] D. J. Allstot, Li Xiaoyong and S. Shekhar, "Design Considerations for CMOS Low-Noise Amplifiers", *Radio Frequency Integrated Circuits Sympo*sium, RFIC-2004, pp. 97-100, June 2004.
- [71] F. Bruccoleri, E. A. M. Klumperink and B. Nauta, "Wide-Band CMOS Low-Noise Amplifier Exploiting Thermal Noise Canceling", *IEEE J. Solid-State Circuits*, vol. 39, no. 2, pp. 275-282, February 2004.
- [72] X. Li, S. Shekhar and D. J. Allstot, "G<sub>m</sub>-Boosted Common-Gate LNA and Differential Colpitts VCO/QVCO in 0.18-μm CMOS", *IEEE J. Solid-State* Circuits, vol. 40, no. 12, pp. 2609-2619, December 2005.
- [73] W. M. C. Sansen, Analog Design Essentials, Springer, 2006.
- [74] A. N. Karanicolas, "A 2.7-V 900-MHz CMOS LNA and Mixer", IEEE J. Solid-State Circuits, vol. 31, no. 12, pp. 1939-1944, December 1996.
- [75] H. Fouad, K. Sharaf, E. El-Diwany and H. El-Hennawy, "An RF CMOS cascode LNA with current reuse and inductive source degeneration", Proc. IEEE Midwest Symposium on Circuits and Systems - MWSCAS 2001, vol. 2, pp. 824-828, August 2001.
- [76] F. Gatta, E. Sacchi, F. Svelto, P. Vilmercati and R. Castello, "A 2-dB Noise Figure 900-MHz Differential CMOS LNA", *IEEE J. Solid-State Circuits*, vol. 36, no. 10), pp. 1444-1452, October 2001.

- [77] H. Doh, Y. Jeong, S. Jung and Y. Joo, "Design of CMOS UWB Low Noise Amplifier with Cascode Feedback", Proc. IEEE Midwest Symposium on Circuits and Systems - MWSCAS 2004, vol. II, pp. 641-644, July 2004.
- [78] J. Lee and Y. Kim, "CMOS Low Noise Amplifier Design Techniques using Shunt Resistive Feedback", Proc. Asia-Pacific Microwave Conference 2005 -APMC 2005, vol. 3, December 2005.
- [79] B. M. Ballweber, R. Gupta and D. J. Allstot, "A Fully Integrated 0.5-5.5-GHz CMOS Distributed Amplifier", *IEEE Trans. on Solid-State Circuits*, vol. 35, no. 2, pp. 231-239, February 2000.
- [80] S.-C. Shin, C.-S. Lin, M.-Da Tsai, K.-Y. Lin and H. Wang, "A Low-Voltage and Variable-Gain Distributed Amplifier for 3.1-10.6 GHz UWB Systems", *IEEE Microwave and Wireless Components Letters*, vol. 16, no. 4, pp. 179-181, April 2006.
- [81] S. Arekapudi, "A Low-Power Distributed Wide-Band LNA in 0.18 μm CMOS", Proc. IEEE International Symposium on Circuits and Systems -ISCAS 2005, vol. 5, pp. 5055-5058, May 2005.
- [82] Payam Heydari, Denis Lin, Amin Shameli and Ahmad Yazdi, "Design of CMOS Distributed Circuits for Multiband UWB Wireless Receivers", Proc. Radio Frequency Integrated Circuits Symposium, RFIC-2005, pp. 695-698, June 2005.
- [83] D. J. Cassan and J. R. Long, "A 1-V Transformer-Feedback Low-Noise Amplifier for 5-GHz Wireless LNA in 0.18-μ m CMOS", *IEEE J. Solid-State Circuits*, vol. 38, no. 3, pp. 427-435, March 2003.
- [84] C. Xin and E. S.-Sinencio, "A GSM LNA using Mutual-Coupled Degeneration", *IEEE Microwave and Wireless Components Letters*, vol. 15, no. 2, pp. 68-70, February 2005.
- [85] G. Nohra, R. Raut and M. Sawan, "A 0.85V Tunable Gain 5GHz Cascode Low Noise Amplifier", Proc. IEEE Northeast Workshop on Circuits and Systems -NEWCAS 2004, pp. 353-356, June 2004.
- [86] P. Leroux, J. Janssens and M. Steyaert, "A 0.8-dB NF ESD-Protected 9-mW CMOS LNA operating at 1.23 GHz", *IEEE J. Solid-State Circuits*, vol. 37, no. 6, pp. 760-765, June 2002.
- [87] K.-J. Koh, Y.-J. Lee, Y.-S. Youn, J.-H. Chang and H.-K. Yu. A 2GHz 16dBm IIP3 Low Noise Amplifier in 0.25µm CMOS Technology. Proc. IEEE International Solid-State Circuits Conference - ISSCC 2003, 2003.
- [88] H. Fouad, K. Sharaf, E. El-Diwany and H. El-Hennawy, "An RF CMOS Modified-Cascode LNA with Inductive Source Degeneration", *Proc. Nine*teenth National Radio Science Conference, Alexandria, vol. 1, pp. 450-457, Alexandria, Egypt, March 2002.
- [89] A. S. Sedra and K. C. Smith, *Microelectronic Circuits*, Oxford University Press, 1998.
- [90] L. Belostotski and J. W. Haslett, "Noise Figure Optimization of Inductively Degenerated CMOS LNAs with Integrated Gate Inductors", *IEEE Trans. on Circuits and Systems - I: Fundamental Theory and Applications - TCAS-I*, vol. 53, no. 7, pp. 1409-1422, July 2006.
- [91] D. Linten, S. Thijs, M. I. Natarajan, P. Wambacq, W. Jeamsaksiri, J. Ramos, A. Mercha, S. Jenei, S. Donnay and S. Decoutere, "A 5-GHz Fully Integrated ESD-Protected Low-Noise Amplifier in 90-nm RF CMOS", *IEEE J. Solid-State Circuits*, vol. 40, no. 7, pp. 1434-1442, July 2005.
- [92] J.-S. Goo, H.-T. Ahn, D. J. Ladwig, Z. Yu, T. H. Lee and R. W. Dutton, "A Noise Optimization Technique for Integrated Low-Noise Amplifiers", *IEEE J. Solid-State Circuits*, vol. 37, no. 8, pp. 994-1002, August 2002.
- [93] P. Andreani and H. Sjoland, "Noise Optimization of an Inductively Degenerated CMOS Low Noise Amplifier", *IEEE Trans. on Circuits and Sistems -II: Analog and Digital Signal Processing*, *TCAS-II*, vol. 48, no. 9, pp. 835-841, September 2001.
- [94] T.-K. Nguyen and S.-G. Lee, "A Sub-mA, High-Gain CMOS Low-Noise Amplifier for 2.4 GHz Applications", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2006, May 2006.
- [95] H. S. Savci, Z. Wang, A. Sula, N. S. Dogan and E. Arvas, "A 1-V UHF Low Noise Amplifier for Ultralow Power Applications", *Proc. IEEE International* Symposium on Circuits and Systems - ISCAS 2006, May 2006.
- [96] T.-K. Nguyen, S.-K. Han and S.-G. Lee, "Ultra-Low-Power 2.4 GHz Image-Rejection Low-Noise Amplifier", *IEE Electronic Letters*, vol. 41, no. 15, July 2005.

- [97] I. Filanovsky, "On Design of Narrow-Band Low-Noise Amplifiers with Inductive Source Degeneration", Proc. IEEE Midwest Symposium on Circuits and Systems, MWSCAS 2000, vol I, pp. 64-67, August 2000.
- [98] R. Ramzan, L. Zou and J. Dabrowski, "LNA Design for on-Chip RF Test", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2006, vol. I, May 2006.
- [99] T.-K. Nguyen, C.-H. Kim, G.-Ju Ihm, M.-Su Yang and S.-G. Lee, "CMOS Low-Noise Amplifier Design Optimization Techniques", *IEEE Trans. on Microwave Theory and Tecniques*, vol. 52, no. 5, pp. 1433-1442, May 2004.
- [100] J. L. R. Marrero, "Simplified Analysis of Feedback Amplifiers", *IEEE Trans.* on Education, vol. 48, no. 1, pp. 53-59, 2005.
- [101] S. Rosenstark, "A Simplified Method of Feedback Amplifier Analysis", IEEE Trans. on Education, vol. E-17, no. 4, pp. 192-198, November 1974.
- [102] P. R. Gray and R. G. Meyer, Analysis and Design of Analog Integrated Circuits, John Wiley and Sons, Inc., 1984.
- [103] B. Nikolic and M. Slavoljub, "A General Method of Feedback Amplifier Analysis", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 1998, vol. 3, pp. 415-418, May 1998.
- [104] P. J. Hurst, "A Comparison of Two Approaches to Feedback Circuit Analysis", *IEEE Trans. on Education*, vol. 35, no. 3, pp. 253-261, August 1992.
- [105] H. T. Russell Jr., "A Loop-Breaking Method for the Analysis and Simulation of Feedback Amplifiers", IEEE Trans. on Circuits and Systems - I: Fundamental Theory and Applications, TCAS-I, vol. 49, no. 8, pp. 1045-1061, August 2002.
- [106] E. H. Nordholt, Design of High-Performance Negative-Feedback Amplifiers, Delftse Uitgevers Maatschappij, 1993.
- [107] S. Wu and B. Razavi, "A 900-MHz/1.8-GHz CMOS Receiver for Dual-Band Applications", *IEEE J. Solid-State Circuits*, vol. 12, no. 12, pp. 2178-2185, December 1998.
- [108] C. Garuda and M. Ismail, "A Multi-Band CMOS RF Front-end for 4G WiMAX and WLAN Applications", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2006, vol. 1, pp. 3049-3052, May 2006.

- [109] P. Andreani and H. Sjoland, "Noise Optimization of an Inductively Degenerated CMOS Low Noise Amplifier", *IEEE Trans. on Circuits and Sistems -II: Analog and Digital Signal Processing*, *TCAS-II*, vol. 48, no. 9, pp. 835-841, August 2001.
- [110] C. Xin and E. S.-Sinencio, "A Linearization Technique for RF Low Noise Amplifier", Proc. IEEE International Symposium on Circuits and Systems -ISCAS 2004, vol. 4, pp. 313-316, May 2004.
- [111] A. Ismail and A.A. Abidi, "A 3-10 GHz Low-Noise Amplifier with Wideband LC-Ladder Matching Network", *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2269-2277, December 2004.
- [112] S. H. M. Lavasani, B. Chaudhuri and S. Kiaei, "A Pseudo-Concurrent 0.18μm Multi-Band CMOS LNA", Proc. IEEE MTT-S International Microwave Symposium, vol. 1, pp. A181-A184, June 2003.
- [113] T. K. K. Tsang and M. N. El-Gamal, "Dual-Band Sub-1V CMOS LNA for 802.11A/B WLAN Applications", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2003, vol. 1, pp. 217-220, May 2003.
- [114] C. F. Jou, K.-H. Cheng, P.-R. Huang, M.-C. Chen, "Design of a Fully Integrated High Linearity Dual-Band CMOS LNA", Proc. IEEE International Conference on Electronics, Circuits and Systems - ICECS 2003, vol. 3, pp. 978-981, December 2003.
- [115] V. Vidojkovic, J. van der Tang, E. Hanssen, A. Leeuwenburgh and A. van Roermund, "Fully Integrated DECT/Bluetooth Multi-band LNA in 0.18µm CMOS", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2004, vol. I, pp. 565-568, May 2004.
- [116] P. Rossi, A. Liscidini, M. Brandolini and F. Svelto, "A Variable Gain RF Front-End, Based on a Voltage-Voltage Feedback LNA, for Multistandard Applications", *IEEE J. Solid-State Circuits*, vol. 40, no. 3, pp. 690-697, March 2005.
- [117] G. Cusmai, M. Brandolini, P. Rossi and F. Svelto, "A 0.18µm CMOS Selective Receiver Front-End for UWB Applications", *IEEE J. Solid-State Circuits*, vol. 41, no. 8, pp. 1764-1771, August 2006.

- [118] M.T. Reiha, J.R. Long and J.J. Pekarik, "A 1.2 V reactive-feedback 3.1-10.6 GHz ultrawideband low-noise amplifier in 0.13 um CMOS", Proc. IEEE Radio Frequency Integrated Circuits Symposium - RFIC-2006, June 2006.
- [119] L. Belostotski, J. W. Haslett and B. Veidt, "Wide-Band CMOS Low Noise Amplifier for Applications in Radio Astronomy", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2006, pp. 1347-1350, May 2006.
- [120] G.E. Valley and H. Wallman, Vacuum-Tube Amplifiers, McGraw Hill, 1948.
- [121] J.M. Petit and M.M. McWhorter, *Electronic Amplifier Circuits*, McGraw Hill, 1961.
- [122] D. Johns and K. Martin, Analog Integrated Circuit Design, John Wiley and Sons, 1997.
- [123] AMS, 0.35 µm HBT BiCMOS RF SPICE Models, AMS, 2005.
- [124] H. Chen, F. Whiteside and R. Geiger, "Current Mirror Circuit with Accurate Mirror Gain for Low β Transistors", Proc. IEEE International Symposium on Circuits and Systems - ISCAS 2001, vol. 1, pp. 536-539, May 2001.
- [125] Y. K. Koutsoyannopoulos and Y. Papananos, "Systematic Analysis and Modeling of Integrated Inductors and Transformers in RF IC Design", *IEEE Trans.* on Circuits and Sistems - II: Analog and Digital Signal Processing, TCAS-II, vol. 47, no. 8, pp. 699-713, August 2000.
- [126] http://www.airborn.com.au/method/fr4pcb.html
- [127] http://mcalc.sourceforge.net/#calc
- [128] N. Karim and A. P. Agrawal, "Plastic Packages'Electrical Performance: Reduced Bond Wire Diameter", http://www.amkor.com/services/newabstr.pdf.
- [129] L. B. Oliveira and M. A. Martins, "Rf Amplifier Characterization using a Network Analyser", *Technical Report 2*, INESC-ID, January 2005.
- [130] http://ece.wpi.edu/ frontier/glossary.html
- [131] B. Rejaei, "Mixed-Potential Volume Integral-Equation Approach for Circular Spiral Inductors", *IEEE Trans. on Microwave Theory and Techniques*, vol. 52, no. 8, pp 1820-1829, August 2004.

## Index

1 dB Compression Point, 21 2-Port, 15 ADC, 14 Admittance matrix, 185 Amplifying Block, 122 Analog-to-Digital Converter, 14 Antenna, 3 Available Power Gain, 19 Bandwidth Fractional, 11 Signal, 21 System, 21 Baseband Signal, 12, 14 Blakesley transformation, 193 Boltzmann Constant  $k_B$ , 27 Carrier Number Fluctuation Theory, 27Cascaded Systems, 25 Chain matrix, 185 Channel Bandwidth, 9 Channel Bandwidth, 9 Communication System, 9 Compression Point 1 dB, 21 Concurrent Dual-Band LNA using Magnetically Coupled Inductors, 92 Concurrent Multi-Band LNA, 89

DLF, 59 DLF LNA having series-series feedback inside shunt-series feedback, 68 having shunt-series feedback inside series-series feedback, 69 having shunt-shunt feedback inside series-shunt feedback, 64, 67 lower frequency boundary, 117 Double Loop Feedback, 59 Amplifying Block Design, 148 Feedback Network Design, 147 LNA, 111 LNA Buffer Design, 150 LNA Design, 146 LNA Input Impedance, 112, 115, 118, 122 LNA Layout, 155 LNA Voltage Gain, 112, 115, 118, 122Test Board, 163 Topologies, 63 Transformer Layout, 155 Dual hybrid matrix, 185 Dynamic Range, 21 Eddy Current, 35 Excess Noise Factor, 23 Feedback, 56 Amplifying block A, 56 Feedback factor, 56

Feedback network  $\beta$ , 56 Gain, 56 Loop gain, 56 Low Noise Amplifier, 58 Positive feedback, 56 Series-Series Topology, 57 Series-Shunt Topology, 57 Shunt-Series Topology, 57 Shunt-Shunt Topology, 57 Feedback Network, 114 Flicker Noise, 27 Carrier Number Fluctuation Theory, 27 Mobility Fluctuation Theory, 27 Fourier Transform, 22 Fractional Bandwidth, 11 Friis Law, 26 Gain, 19 Hybrid matrix, 185 IF Signal, 14 IF-Receiver, 12 IIP<sub>3</sub>, 20, 26 Impedance Matching, 13, 15 Impedance matrix, 185 Inductor, 31 Integrated, 31 Resistance, 115 Integrated Inductor, 31 Intermediate Frequency (IF), 12 Linearity, 20 LNA, 3, 13  $1/g_m$  Termination, 46 Cascode LNA with Inductive Degeneration, 51

Common-Gate, 46

Common-Source LNA with Inductive Degeneration, 47 Concurrent Dual-Band LNA using Magnetic Coupled Inductors, 92Concurrent Multi-Band, 89 DLF, 56 Double Loop Feedback, 56 Dual-Band LNA with Reverse Orientation of the Magnetic Coupling, 104Input Impedance, 53, 112, 115, 118, 122 LNA using Cascode Transistor, 52 multi-band LNA, 76 Multi-Band LNA using Cascode Transistors for Band Selection, 77 Noise Figure, 53 Non-Concurrent Multi-Band, 89, 90 using Input Resistor, 45 Voltage Gain, 112, 115, 118, 122 with Current Reuse, 48 with Resistive Feedback, 49 with Transformer Feedback, 51 with Tuned Inductor Feedback, 52 Local Oscillator (LO), 3, 12, 14Low Noise Amplifier, 3, 13, 44 Gain, 44 Input Impedance, 44 Linearity, 44 Noise, 44 Lumped Element Circuit, 15 Magnetic Coupling Coefficient, 38 Mixer, 3, 14 Mobility Fluctuation Theory, 27

Multi-band LNA Transistors using Cascode for Band Selection, 77 negative feedback, 56 Noise, 22 1/f (see Flicker Noise), 27 Bipolar, 29 CMOS Transistors, 28 External, 22 Flicker, 27 Fundamental, 22 Power, 9 Shot, 27 Sources, 26 Thermal, 26 Noise Factor, 23, 26 Spot, 23, 24 Noise Figure, 23 Noise Power, 9 Non-Concurrent Multi-Band LNA, 89, 90 nullor, 56, 58 Operating Power Gain, 19 Patterned Ground Shield, 35 Port, 15 Power Available, 23 Power Gain, 19 Available, 19 Operating, 19 Transducer, 19 Power Spectral Density, 22 Radio Frequency, 3 Receiver Heterodyne, 12

Homodyne, 12 IF, 12 Multi-Band, 11 Narrowband, 11 Wideband, 11 Zero-IF, 12 Reflection Coefficient, 15 RF, 3 S-Parameters, 17 Scattering-Parameters, 17 Shannon Equation, 9 Shot Noise, 27 Signal Bandwidth, 21 Baseband, 12, 14 Distortion, 20 IF, 14 Power, 9 Signal-to-Noise Ratio, 9, 23 Spot Noise Factor, 23 System Bandwidth, 21 Taylor Coefficients, 20 Series, 20 Test Board, 163 Thermal Noise, 26 Third-Order Intercept Point, 20, 26 Transceiver, 9 Capability, 9 Transducer Power Gain, 19 Transformer, 35 Ideal, 38 Integrated, 35 Inter-Winding Capacitances, 118 Interleaved, 36 Magnetic Coupling Coefficient, 38 non-ideal, 38 Stacked, 37 Tapped, 36 Ultra-Wideband Signal, 12 UWB, 12 Voltage Controlled Oscillator, VCO, 14 Wiener-Khintchine Theorem, 22 Wireless Receiver, 3 Zero-IF Receiver, 12